NUREG/CR-3065 813-1163 Vol. 1

In Situ Test Programs Related to Design and Construction of High-Level Nuclear Waste (HLW) Deep Geologic Repositories

Main Report Final Report (Task 2) - June 1981-November 1982

Prepared by W. Roberds, F. Bauhof, L. Gonano, E. Wildanger, W. Dershowitz, F. Marinelli, K. Jones, C. Weldon, B. Nebbitt, M. Stewart, R. Gates, D. Pentz, J. Byrne

Golder Associates

Prepared for U.S. Nuclear Regulatory Commission

> 8303290114 830331 PDR NUREC CR-3065 R PD

NOTICE

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency thereof, or any of their employees, makes any warranty, expressed or implied, or assumes any legal liability of responsibility for any third party's use, or the results of such use, of any information, apparatus, product or process disclosed in this report, or represents that its use by such third party would not infringe privately owned rights.

Availability of Reference Materials Cited in NRC Publications Most documents cited in NRC publications will be available from one of the following sources: 1. The NRC Public Document Room, 1717 H Street, N.W. Washington, DC 20555 2. The NRC/GPO Sales Program, U.S. Nuclear Regulatory Commission, Washington, DC 20555 3. The National Technical Information Service, Springfield, VA 22161 Although the listing that follows represents the majority of documents cited in NRC publications, it is not intended to be exhaustive. Referenced documents available for inspection and copying for a fee from the NRC Public Document Room include NRC correspondence and internal NRC memoranda; NRC Office of Inspection and Enforcement bulletins, circulars, information notices, inspection and investigation notices; Licensee Event Reports; vendor reports and correspondence; Commission papers; and applicant and licensee documents and correspondence. The following documents in the NUREG series are available for purchase from the NRC/GPO Sales Program: formal NRC staff and contractor reports, NRC-sponsored conference proceedings, and NRC booklets and brochures. Also available are Regulatory Guides, NRC regulations in the Code of Federal Regulations, and Nuclear Regulatory Commission Issuances. Documents available from the National Technical Information Service include NUREG series reports and technical reports prepared by other federal agencies and reports prepared by the Atomic Energy Commission, forerunner agency to the Nuclear Regulatory Commission. Documents available from public and special technical libraries include all open literature items, such as books, journal and periodical articles, and transactions. Federal Register notices, federal and state legislation, and congressional reports can usually be obtained from these libraries. Documents such as theses, dissertations, foreign reports and translations, and non-NRC conference proceedings are available for purchase from the organization sponsoring the publication cited. Single copies of NRC draft reports are available free upon written request to the Division of Technical Information and Document Control, U.S. Nuclear Regulatory Commission, Washington, DC 20555 Copies of industry codes and standards used in a substantive manner in the NRC regulatory process are maintained at the NRC Library, 7920 Norfolk Avenue, Bethesda, Maryland, and are available there for reference use by the public. Codes and standards are usually copyrighted and may be purchased from the originating organization or, if they are American National Standards, from the American National Standards Institute, 1430 Broadway, New York, NY 10018. GPO Printed copy price: \$7.50

NUREG/CR-3065 813-1163 Vol. 1

In Situ Test Programs Related to Design and Construction of High-Level Nuclear Waste (HLW) Deep Geologic Repositories

Main Report Final Report (Task 2) - June 1981-November 1982

Manuscript Completed: November 1982 Date Published: March 1983

Prepared by W. Roberds, F. Bauhof, L. Gonano, E. Wildanger, W. Dershowitz, F. Marinelli, K. Jones, C. Weldon, B. Nebbitt, M. Stewart, R. Gates, D. Pentz, J. Byrne

Golder Associates 295C Northup Way Bellevue, WA 98004-1486

Prepared for Division of Waste Management Office of Nuclear Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20555 NRC FIN B6983 REPORTS IN SERIES

Report Number	Issuance Date	Contract Task	Title
NUREG/CR-2613	3/12/82	1	Identification of Characteristics which Influence Repository Design - Tuff
NUREG/CR-3065 Vols. 1 & 2	3/83	1	Identification of Characteristics Which Influence Repository Design - Domal Salt
(draft)	2/82	2	In Situ Test Programs Related to Design and Construction of High Level Nuclear Waste (HLW) Deep Geologic Repositories
NUREG/CR-2854	7/9/82	3	Evaluation of Alternative Shaft Sinking Techniques for High Level Nuclear Waste (HLW) Deep Geologic Repositories
NUREG/CR-2959	10/12/82	4	Relationship of an In Situ Test Facility to a Deep Geologic Repository for High Level Nuclear Waste
(in preparatio	n)	5	Evaluation of Engineering Aspects of Backfill Placement for High Level Nuclear Waste (HLW) Deep Geologic Repositories

ABSTRACT

This report represents the results of Task 2, "In Situ Test Programs Related to Design and Construction of High Level Nuclear Waste (HLW) Deep Geologic Repositories," of U.S. Nuclear Regulatory Commission (NRC) Contract NRC-02-81-037, "Technical Assistance for Repository Design." The purpose of the complete project is to provide the NRC with technical assistance to enable the focused, adequate review by NRC of aspects related to design and construction of an in situ test facility and final geologic repository, as presented in U.S. Department of Energy (DOE) Site Characterization Reports (SCR) and License Application (LA).

The Task 2 results include the general recommendation of available tests which should be considered in designing media/site specific in situ test programs. Tests will be conducted within an in situ test facility, consisting of an exploratory shaft and an underground test facility at the prospective repository horizon. Plans for these programs are expected to be presented in the initial SCR and the complete results presented in the LA. The media and sites considered include (1) basalt at Hanford, Washington; (2) tuff at Yucca Mountain, Nevada Test Site; (3) domal salt at specific Gulf Coast sites; (4) bedded salt at an unspecified site; (5) granite at an unspecified site.

A licensing perspective is outlined and a defensible rationale developed and utilized for the test selection process. This rationale essentially consists of:

- Establishing the information needs for construction authorization
- Assessing the relevant capabilities of available tests
- Matching the capabilities of specific tests to the perceived information needs.

The information needs at any time consist of the additional information (if any) needed in order to predict satisfactory repository system performance with the required level of confidence, and thus are a function of:

- The significance of the repository engineered components and site characteristics to system performance
- The currently available information, which may be supplemented with time
- The acceptable level of confidence in satisfactory performance for each licensing step.

Determination of the acceptable levels of confidence and the significance of repository system components is outside the scope of this report. Suitable assumptions have thus been made regarding the development of information needs for construction authorization by the time of initial SCR submittals.

Tests which are available and respond to the perceived media/site specific information needs, either by simulation or assessment of site characteristics, are identified and their capabilities assessed. Specific in situ tests are investigated and described in detail. Research and development which might be effective in improving test capabilities have been recommended.

EXECUTIVE SUMMARY

ES.1 INTRODUCTION

This report represents the results of Task 2, "In Situ Test Programs Related to Design and Construction of High Level Nuclear Waste (HLW) Deep Geologic Repositories," of U.S. Nuclear Regulatory Commission (NRC) Contract NRC-02-81-037, "Technical Assistance for Repository Design."

The purpose of the complete project is to provide NRC with technical assistance for the following reasons:

- To enable the focused, adequate review by NRC of aspects related to design and construction of an in situ test facility and final geologic repository, as presented in U.S. Department of Energy (DOE) Site Characterization Reports (SCR)
- To ascertain that the DOE site characterization will provide, as far as possible, all the information necessary to permit a review to be conducted by NRC of a License Application (LA).

This report presents a licensing perspective, describes a methodology for designing an in situ test program and recommends a tentative test program together with test details which should be considered between initial SCR submittal and LA. The in situ tests, tentatively recommended to be conducted within an in situ test facility, have been compiled with reference to the currently perceived specific information needs of five media/sites: (1) basalt at Hanford, Washington; (2) tuff at Yucca Mountain, Nevada Test Site; (3) domal salt at specific Gulf Coast sites; (4) bedded salt at an unspecified site; (5) granite at an unspecified site.

ES.2 PERSPECTIVE

Any deep geologic repository for the permanent disposal of high level waste (HLW) must be designed to achieve certain performance objectives, which can be summarized as:

- Short term construction and operation objective (through decommissioning, about 100 years) of minimizing hazards jeopardizing the safety of the public and personnel during repository construction and operation (including possibly retrieval and decommissioning activities).
- Long term waste containment and isolation objective (post-decommissioning, from about 100 to 10,000's years) of minimizing radionuclide flux (rate/unit area) to accessible environment and thus minimizing hazards jeopardizing public safety after decommissioning. This objective dictates maintaining a waste retrieval capability for a specified period after waste emplacement

and prior to decommissioning, thereby providing the opportunity for verifying a sufficiently high probability of satisfactory long-term performance and also providing a contingency plan for demonstrated non-verification.

Performance criteria define and, where possible, quantify performance objectives. These criteria can be given either deterministically (i.e., an absolute numerical limit) or probabilistically (i.e., an acceptable level of confidence that a numerical limit will not be exceeded by the particular repository performance indicator). Where the criteria are given deterministically, a determination must still be made, although implicitly rather than explicitly, of the level of confidence in not exceeding the deterministic limit.

The performance of the repository will be a function of both the engineered components of the repository and the inherent characteristics of the site in which it is located. In many cases, the response of the combined system is a result of an interaction of both engineered components and site characteristics. Thus, performance objectives, criteria and assessment can be related to the integrated system of engineered components and site characteristics. This concept forms the basis for the methology of establishing the information needs and the associated in situ testing requirements presented here.

There are two primary ways of maximizing the level of confidence that the performance criteria will be achieved and that the actual performance of the repository system will be satisfactory:

- Selecting a suitable repository site
- Appropriately designing, constructing, and operating the repository.

Several questions are apparent:

- How can a suitable site for a repository be selected so as to maximize the level of confidence in satisfactory performance?
- Once a site has been selected, how can the repository be designed and constructed so as to maximize the level of confidence in satisfactory performance?
- How can a high level of confidence in satisfactory performance be demonstrated at each step in repository development?

These three questions are critical to the licensing process, and define the development of a repository. There is a need to assess, at various discrete stages during development, the level of confidence in satisfying each designated performance criterion, based on the available data on engineered components and site characteristics and the uncertainty of this data. If that confidence level is found to be acceptably high, the appropriate authorization or license amendment necessary to initiate the next phase of repository development is granted. However, if the level of confidence in compliance is found to be unacceptably low, then one of two approaches can be taken by the applicant:

- If the level of confidence can be increased by additional cost effective characterization or design modification, then that characterization or modification may be performed and the application updated.
- (2) If the level of confidence cannot be cost effectively increased by further characterization or design modification, then repository development should be stopped at this site.

The discrete points in repository development are (see Figure ES-1):

- (1) Site characterization report (SCR) submittals, primarily DOE
- decision points for which NRC only offers opinions
- (2) License application (LA)
- (3) Updated application for license to operate (emplace waste)
- (4) License amendment to decommission.

The acceptable level of confidence in satisfactory performance for each step can be established by considering the consequences of not satisfying the performance criteria at that particular stage, as well as the risks for other alternatives in HLW disposal. These acceptable levels will increase with repository development. As presently perceived, it should be highly probable at LA that ultimate repository system performance will be satisfactory in order to provide reasonable assurance that the repository will be fully licensable, prior to extensive development and major expenditure of funds.

Acceptable levels of confidence may be established either:

- Implicitly, through progressive technical discussions between the regulator and applicant
- Explicitly, through rigorous decision-making utilizing quantitative risk assessment methodology.

Although recognizing the difficulties in performing quantitative risk assessments and subsequently defending the results in the decisionmaking process (especially within the institutional framework), Golder Associates considers it important to be attempting to utilize uncertainty and probability assessments of the important parameters affecting repository performance. These assessments should be used as a guidance tool during the site characterization phase. While these probability approaches are being established, the current approach of assessing acceptability implicitly through technical discussions needs to be continued.



It is anticipated that certain testing/monitoring methods will need to be utilized in order to demonstrate an acceptable level of confidence in satisfactory performance at each step of repository development. This perspective of the licensing process can be summarized, as follows:

- For site screening/selection, as summarized in the initial SCR submittal, site investigation will focus generally on large-scale features and consists of surface, borehole and laboratory testing. This generally rough assessment of characteristics will allow for comparison between sites and for the demonstration of an acceptable level of confidence in satisfactory performance based on a conceptual repository design.
- For a detailed repository design, as summarized in a license application (LA), site characterization will focus on features adjacent to planned shafts and the repository horizon, as well as on large-scale features. This site characterization will consist primarily of the construction and operation of an in situ test facility. In addition to the improved assessment of site characteristics provided by in situ testing and by monitoring of the in situ test facility, certain aspects of key issues may be adequately resolved by prototype simulation (and extrapolation of results) and predictive models may be partially verified. The construction and operation of this in situ test facility may thus provide information for the detailed repository design and for the demonstration of an acceptable level of confidence in satisfactory performance with that design.
- In an updated application for license to operate (emplace waste), the assessment of site characteristics will be further refined as more areas underground are exposed and as the performance of the repository is monitored. This monitoring may also provide information for modifications in design (if required), additional verification of predictive models, updating of performance predictions, and for the demonstration of an acceptable level of confidence in satisfactory performance.
- Finally, in a license amendment to decommission, the assessment of site characteristics will be further refined by monitoring repository performance. This monitoring may also provide information for additional verification of predictive models and updating of performance predictions. A final assessment of the level of confidence in satisfactory long-term performance will be made, and a determination made by the NRC as to whether this confidence level is sufficiently high. If it is determined that the confidence level is too low, modifications to the repository or retrieval of the waste will be required.

ES.3 APPROACH

A defensible rationale has been developed and utilized to tentatively select available tests to be included in the media/site specific in situ test programs. This rationale essentially consists of:

- Establishing the information needs for construction authorization at each site
- Assessing the capabilities of available tests to meet the specific information needs
- Matching the capabilities of specific tests to the perceived information needs.

The information needs existing at any time result from the unacceptable uncertainties in the prediction of repository system performance. Information needs are determined as follows:

- Identify the existing information and assess the associated level of confidence in satisfactory repository system performance
- Compare the assessed level of confidence with the acceptable level, either implicitly or explicitly
- Determine what additional information is needed to raise the level of confidence in satisfactory performance to the acceptable level, by:
 - establishing the relationship between each component of the repository system and system performance (i.e., sensitivity)
 - identifying where the existing information regarding significant components of the system is insufficient and can be readily supplemented.

The information needs at any time are thus a function of:

- The significance of repository engineered components and site characteristics to system performance
- The currently available information, which may be supplemented with time
- The acceptable level of confidence in satisfactory performance for each licensing step.

The significance of the repository system components to system performance and the acceptable levels of confidence have not been determined and are outside the scope of this report. Qualitative assessments of acceptable levels of confidence and of the significance of system components, as well as assumptions regarding the information developed by the time of the initial SCR submittal, have thus been made for the purpose of establishing information needs for construction authorization.

The selection of tests to satisfy the perceived information needs involves the prior determination of the relevant capabilities of candidate tests, and the integrated compilation of a program of tests that collectively best respond to the information needs. In the selection of these tests, only presently available tests and potential advancements to the state-of-the-art within current concepts and technologies are considered. With future development, the test capabilities may be improved.

These tests satisfy the information needs either by:

- Simulating various aspects of the repository (e.g., construction techniques) for extrapolation of results
- Assessing identified media/site specific characteristics (e.g., hydraulic conductivity) to be used in numerical modeling
- Verifying predictive numerical models.

It is expected that the in situ test program will evolve with time somewhat independently for each media/site considered as the perceived information needs and test capabilities develop. It is even possible that the information needs for construction authorization might be satisfied prior to the initial SCR submittal, precluding the need for most (if not all) in situ testing. The in situ test program must also be flexible enough to take into account new information which becomes available during its performance, as presented in SCR updates. The complete design and specifications for any particular test cannot be accomplished a priori without detailed descriptions of each individual test location, as well as identification of specific information needs. Hence, the recommendations for the conduct of specific tests are of a scoping nature only.

ES.4 STUDY ACTIVITIES

In the course of developing recommendations for in situ testing, the following activities have been undertaken (see Figure ES-2):

• The licensing perspective, including the various licensing steps, for repository development has been identified, based on the procedural rule of 10-CFR-60. At each step, the level of confidence in satisfactory repository system performance should be assessed, and a determination made of whether that level of confidence is sufficient to allow further development. The acceptable level of confidence at each step underlies the establishment of the information needs.



- Critical engineering variables, which in conjunction with the site will determine repository system performance, have been identified. This has been accomplished by first identifying the primary engineering variables of repository design and construction (e.g., shaft dimension, shape, etc.). The influence of these primary engineering variables on the level of confidence in satisfactory performance has been qualitatively assessed and those perceived to have a significant potential impact, in conjunction with a capability for change at reasonable cost, have been judged to be critical. These critical engineering variables should be investigated by in situ testing, as well as emphasized during NRC's review process.
- Key issues, relating to the performance criteria, have been previously identified in Task 1 of this project. These key issues impact the choice of values for the critical engineering variables and must be adequately resolved to demonstrate an acceptable level of confidence in satisfactory performance. These key issues, which must be addressed in both the SCR and the LA review, include:
 - <u>Constructability</u>. Can the facility be constructed in a timely and safe fashion, and so that it will not jeopardize the waste containment/isolation capability of the facility? Both the unavoidable creation of a disturbed zone of rock around underground openings and the construction of engineered barriers will have an effect on the response of the repository.
 - Thermal Response. Can the temperature field be adequately predicted as a function of time and reliably incorporated into mechanical, hydrological and geochemical models?
 - Mechanical Response. Can the stability and deformation of underground openings (including around the waste package) be adequately predicted for the periods of short-term construction/ operation and long-term waste containment/ isolation?
 - <u>Hydrologic Response</u>. Can an adequate prediction be made regarding the resaturation time of the repository (post-closure) and of the long-term groundwater flow through the repository?
 - <u>Geochemical Response</u>. Can an adequate prediction be made of the extent and effect of geochemical alteration of the engineered barriers and the rock? Can the quantity and rate of migration of specific radionuclides over the long-term be adequately predicted?

Information needs are thus related to the resolution of these key issues.

 Characteristics which have a significant impact on the resolution of the key issues have been identified (see Table ES-1). Each of the characteristics has subsequently been evaluated for each media/site

SIGNIFICANT CHARACTERISTICS

Table ES-1

	. 1	KE Y	ISS	UES	5
CHARACTERISTICS	Constructability	Thermal Response	Mechanical Response	Hydrologic Response	Geochemical Response
GEOLOGIC SETTING					
Stratigraphic/structural*		٠	٠		
Tectonic	0		٠	٠	0
In situ stress field	0	0	٠	0	0
In situ hydraulic head field	•	0	0	٠	0
In situ temperature field	•	٠	0	0	0
RESPONSE					
(MECHANICAL)					
Strength			٠	0	0
Deformation	0		٠	0	
Creep/fusing	0		•	0	0
(THERMAL)	Γ				
Thermal conductivity		•	0		
Heat capacity			0		
Linear thermal expansion		•	•		
(HYDROLOGIC)	Г				
Hydraulic conductivity					0
Effective porosity					
Specific storage				•	
(GEOCHEMICAL)	Г				
Dispersivity					
Adsorption/retardation	1				
Alteration/solubility	0	0	0		

 includes the physcial and chemical characteristics of each rock mass unit (including pore fluid composition).

characteristic significantly impacts resolution of key issues.

o characteristic impacts resolution of key issues, but to a lesser degree

BR Eng 20 10 0.1.0 No AS.2.443 ----

in terms of its influence on satisfying each of the performance criteria. This evaluation has taken into account the following general attributes of each characteristic:

- availability of cost-effective design and construction techniques which allow for a conservative assumption of the value of the characteristic
- uncertainty in the representation of the phenomenological laws of nature by the performance prediction model
- sensitivity of the performance prediction model to the value of the characteristic
- cost effectiveness and scheduling limitations of measures to reduce the uncertainty in the assessment of the value of the characteristic.

From this evaluation, a tentative determination has been made regarding the maximum acceptable level of uncertainty in the assessment of each of the characteristics. However, until the licensing perspective (Figure ES-1) has been clarified and the sensitivity of system performance to all the system components determined, these maximum acceptable levels of uncertainty for each characteristic can be considered as qualitative indicators only.

- The current assessment of the significant characteristics for each media/site has been based on previous work by Golder Associates, performed either as part of this project or in previous projects. These assessments have been presented in terms of a best estimate and an indication of the level of uncertainty in that value.
- Test methods (including surface, borehole, laboratory, and in situ tests) which are available to assess the significant characteristics have been identified. Existing repository-related in situ lesting programs have been summarized to assist in the identification of available in situ tests. The inclusion of other tests has been based on experience.

The <u>capabilities</u> of each of the available test methods regarding its determination of the significant characteristics have been assessed. This assessment has been based partly on the test's representation of those environmental conditions which affect the characteristic. The assessment has also included the factor of test scale, i.e., whether a representative volume of rock mass is tested. The importance of index testing in the assessment of characteristics and their variability throughout the zone of influence has been recognized.

 Those in situ tests which can be used to resolve the key issues by simulation have been identified. These tests, as well as the in situ test facility itself, simulate various construction/ operation aspects of the repository, so that their results can be used to directly predict expected repository performance.

- The testing which will precede in situ testing has been anticipated. The information obtained by this testing at sites where it is now incomplete will supplement the existing information, and the present assessment of significant characteristics may change prior to initial SCR submittal.
- The information needs for construction authorization, which are perceived to exist at the time of initial SCR submittal, have been identified, based on the current media/site specific assessment of significant characteristics, anticipated future site investigation, and the licensing perspective.
- Media/site-specific in situ test programs have been tentatively recommended (see Table ES-2). These programs consist of available in situ test methods which best respond to the information needs, either by simulation or by adequate assessment of the significant characteristics. Where no one appropriate method exists, a combination of independent methods which adequately responds to the perceived information needs has been identified.

An example in situ test facility which can accommodate these programs has been developed under Task 4 of this project.

- Scoping recommendations have been made regarding specific in situ tests. These tests have been investigated in detail, and a description, an evaluation, and a recommendation regarding methodology and utilization of results have been presented for each.
- Potentially effective research and development of in situ testing techniques have been identified and recommended. Currently available in situ tests may present difficulties in fulfilling perceived information needs, especially in assessing characteristics for the additional effects of radiation, temperature, and long term behavior. These recommendations are primarily concerned with advancements within available concepts and technologies and not with the development of new or hybrid tests.

To the extent possible, consideration has been given in each of the above activities to aspects which are perceived to be unique to each of the media and sites under consideration, i.e., basalt, tuff, domal salt, bedded salt, and granite.

ES.5 SUMMARY AND CONCLUSIONS

Golder Associates believes that this report presents defensible recommendations regarding those tests which should be specifically considered in a reasonable in situ test program conducted within an in situ test facility prior to construction authorization at any site. These tests

Rev.			
Dwa	No 18:3-163	Data 12 BL	Eng BR

METHOD (Appendix, Section):			Plate Test (9) (A.1)	Block Test (A.2)	Chamber Test (A.3)	Mine-By Test (A.4)	Heater Test (Large Scale) (A.5)	Heater Test (Small Scale) (A.5)	(A.6)	Multiple Borehole Permeability Test (10) (A.7)	Overcoring (A.8)	Flatjack Test (A.9)	Acoustic Emission Monitoring (A.10)	Exposure	
CH .	Ge	Stratigraphic/ Structural (la)	-											•	
ARAC	Olo	Tectonics											•		
H H	gic	In Situ Stress Field (15)									•	•			
(see	Sett	In Situ Hydraulic Head Field (1b)							•						
CS AS Table	ing	In Situ Tempera- ture Field(1b)													
ES-1)	Mechanical Characteristics (2	•	•		•						•	•		
SED)	Thermal Characteristics (2)		•			•	•							
BY TH	SY TE	,	Hydrologic Characteristics (2		•	•				•	•				•
EST)	Geochemical Characteristics (2)		•	•		•		•					•	
ASPE SIMU ASPE		Constructability				•	•						•	•	
LATED		Repository Performance	1		•	•	•						•	_	
MINI NUMB OF T PERF IN:		Exploratory Shaft	1	•	+	1	1	Ø	ø	۵	σ	8	0	0	
ORMED		Underground Test Facility	0	N	-	-	-	80	8	ω	σ	8	0	0	
RE		Basalt (3)	•	•	•	•	•	•	•	•	•	•	•	•	
CON		Fuff (4)	• 1	•	•	•	•	•	•	•	•	•	•	•	
TENINE NI		Domal Salt (5)	1	•		•	•	•			•	•	•	•	
M H M		Bedded Salt (6)	1	•		•	•	•			•	•	•	•	
IS POR		Granite (7)		•	•	•	•	•	•	•	•	•	•	•	

>

TINX

REASONABLE IN SITU TEST PROGRAMS

Table ES-2

2 of 3

IN SITU TEST METHOD (Appendix, Section):	CHARACTERISTICS ASSESSED BY TEST: (see Table ES-1)								ASPE SIMU BY T	CTS LATED EST:	MININ NUMBE OF TH PERFO IN:	AUM ER ESTS DRMED	WHICH TEST IS RECOMMENDED					
	Geo	108	ic	Sett	ing	2)	2)	2)	2)									
	Stratigraphic/ Structural (la)	Tectonics	In Situ Stress Field(1b)	In Situ Hydraulic Head Fieid (1b)	In Situ Tempera- ture Field (1b)	Mechanical Characteristics (2	Thermal Characteristics (2	Hydrologic Characteristics (Geochemical Characteristics (:	Constructability	Repository Performance	Exploratory Shaft	Underground Test Facility	Basalt (3)	Tuff (4)	Domal Salt (5)	Redded Salt (6)	
Exploratory Excavations (11)												*	*	•	•	•	•	
Coreholes (12)	•									•		b	b	•	•	•	•	
Rock Mass Sampling (13b)	•					•	•	•	•			b	b	•	•	•	•	
Groundwater Sampling (13a)									•			ъ	b	•	•	•	•	
Permeability Test (Single Borehole) (10)								•				b	b	•	•	•	•	
Hydro- fracturing (17)			•									a	b	•	•	•	•	
Laboratory Tests (14)	•					•	•	•	•	•		b	b	•	•	•	•	
Index Tests on Exposures (15)						•				•		b	b	•	•	•	•	
Geophysic/Seismic (X-hole, Exposure-borehole)	•					•				•		ь	b	•	•	•	•	
Seismic Monitoring		•									•	c	c	•	•	•	•	
Monitor Temperature in Rock Mass and Excavations (16)					•		•			•	•	c	c	•	•	•	•	
Monitor Displace- ments in Rock bass and Excavations (16)						•	•			•	•	c	c		•	•	•	
Monitor Pore Pressures in Rock Mass (16)				•				•			•	c	c	•	•	•	•	
Monitor Drainage into Excavation(16)								•		•	•	c	c	•	•	•	•	,
Monitor Alteration, Solutioning of Exposures	1							1	•		•	c	c	•	•	•	•	•
Construction Monitoring										•	•	c	c	•	•	•		
Operation Monitoring											•	c	c	•		•		

Eng 2/2 Date /2 32 Pwg. No. 49/3-1/62

REASONABLE IN SITU TEST PROGRAMS

Table ES-2

3 of 3

 њz.		 -	
n	n . 1	D 1	C
	_	 _	

38

Eng

10-82

D....

163

181

No Rev

- a = several (≤ 10)
- b = numerous (>10) C = continual
- # = Exploratory shaft and underground test facility are exploratory excavations.
- a) The stratigraphy/structure includes the physical and chemical characteristics of each rock mass unit (including poor fluid composition).
 - b) The in situ (i.e., pre-excavation or virgin) stress, hydraulic, and temperature fields can be indirectly assessed or inferred from the stratigraphy/structure and tectonics (e.g., in situ stress field can be inferred from the geomorphology and tectonics of the site).
- (2) The response characteristics refer to the rock mass, which consists of intact rock, discontinuities and pore fluid. These reponse characteristics can be assessed either:
 - Directly by testing a large scale sample which contains a significant number of discontinuities
 - Indirectly by separately assessing the response characteristics of the intact rock, discontinuities, and pore fluid, and then assembling by a model. Hence, the rock mass response characteristics can often be inferred from the stratigraphy/structure.
- (3) Basalt at Hanford, Washington (8)
- Tuff at Yucca Mountain, Nevada (8) (4)
- (5) Domal Salt at Gulf Coast Sites (8)
- Bedded Salt at unspecified site (8) (6)
- (7) Granite at unspecified site (8)
- (8) Significance of characteristics for in situ testing, as they relate to design, for each media/site was subjectively evaluated using a specific process.
- (9) Plate test is very similar to two other tests:
 - · Cable jacking test, in which the reaction is provided by an anchor in the rock mass rather than the opposite wall of the excavation.
 - · Radial jacking, in which the entire circumference of the opening is jacked using, for example, several plate jack systems.

Because of these similarities, only the plate test will be discussed, although cable jacking or radial jacking might be suitable alternatives.

- (10) Tests are constant head injection, constant head withdrawal, constant flow rate withdrawal, pulse injection, or gas injection permeability test.
- (11) Exploratory excavations include the exploratory shaft and underground test facility.
- (12) Coreholes include coring and core logging, as well as possibly borehole surveying, caliper logging, oriented coring, integral sampling, impression packer, borehole TV/camera, borehole radar, and geophysical well logging (electrical, acoustic, and nuclear).
- (13) a) Groundwater sampling implies subsequent laboratory determination of groundwater composition and age.
 - b) Rock mass sampling, either coring or large block samples, implies subsequent laboratory tests.
- (14) Laboratory tests are performed on rock mass or groundwater samples
- (15) Index tests do not assess the characteristic directly, but by empirical correlations (e.g., use of a Schmidt hammer on rock core or on an exposure is an index test whose results can be roughly correlated with the modulus of deformation, based on experience). There are too many index tests, with varying reliability, to list.
- (16) Monitoring performance implies associated analysis to assess characteristic.
- (17) Hydrofracturing must be very carefully performed in order to control the extent of fractures which are generated.

adequately respond to the information needs for construction authorization which are perceived to exist at the time of initial SCR submittal, and hence assist in resolving the key issues related to short- and long-term performance criteria. The defensibility of these recommendations relies on Golder Associates' current perception of the information needs, which in turn are based on their licensing perspective and assessment of currently available information.

Although some of the judgements made in the selection of these in situ tests are necessarily subjective and the licensing perspective may not be universally shared, the rationale is clearly outlined so that specific areas of technical disagreement can be readily identified and these disagreements (if any) resolved. It must be emphasized, however, that the media/site specific in situ test programs will be a function of both the information needs and test capabilities at the time of initial SCR submittal. It is expected that the in situ test program will evolve with time somewhat independently for each media/site considered as these information needs and test capabilities develop. In addition, the program and the design of specific tests must be flexible enough to take into account new information which becomes available during program performance, as presented in SCR updates.

Golder Associates thus recommends that the NRC should, accordingly, identify the information needs for each site and then focus on (1) the plans of the in situ test program in their review of an SCR and (2) the results of this program, and the appropriate incorporation of these results in design and performance assessment, in their review of a LA.

TABLE OF CONTENTS

ė,

VOLU	IME I -	MAIN TEXT	Page
REPO	RACT	SERIES	ii iii
EXEC	UTIVE	SUMMART	V
	ES.1 ES.2 ES.3	Introduction Perspective Approach	v v
	ES.4 ES.5	Study Activities Summary and Conclusions	xi xvi
TABL LIST LIST	E OF C	ONTENTS GURES BLES	xxi xxv xxix
PREF	ACE -	LICENSING PERSPECTIVE	1
	P.1 P.2	Introduction Overview of High Level Nuclear Waste Isolation Program	1 1
	P.3 P.4 P.5 P.6	Site Selection Repository Design Demonstrability Summary	4 5 7 13
1.0	INTRO	DUCTION	15
	1.1 1.2 1.3	Introduction Approach Study Activities	15 16 20
2.0	KEY I	SSUES FOR REPOSITORY DESIGN AND CONSTRUCTION	27
	2.1	Critical Repository Design and Construction	27
	2.2	Key Issues Which Impact the Choice of the Values of Critical Engineering Variables	27
3.0	SIGNI OF KE	FICANT CHARACTERISTICS WHICH IMPACT THE RESOLUTION Y ISSUES	35
	3.1 3.2	Site Characteristics Variables which Affect Assessment of Response Characteristics	35 38
		3.2.1 Sensitivity 3.2.2 Representative Sample	38 40
	3.3	Uncertainty in the Assessment of	43
	3.4	Present Assessment of Media/Site Specific	45

Table of Contents continued.

			Page
		3.4.1 Basis for Present Assessment 3.4.2 Present Media/Site Specific Assessments	45 45
		3.4.2.1 Basalt 3.4.2.2 Tuff 3.4.2.3 Domal Salt 3.4.2.4 Bedded Salt 3.4.2.5 Granite	45 48 49 49
	3.5	Significance of Each Characteristic	49
		3.5.1 Process of Evaluation 3.5.2 Media/Site Specific Evaluations	49 54
		3.5.2.1 Basalt 3.5.2.2 Tuff 3.5.2.3 Domal Salt 3.5.2.4 Bedded Salt 3.5.2.5 Granite	54 54 58 58
4.0	AVAIL	ABLE TEST METHODS	
	4.1	Existing Repository-Related In Situ Test Programs	61
		4.1.1 Significance of Existing Programs4.1.2 Existing Media Independent Programs	61 61
	4.2	Test Methods Available to Assess Significant	65
	4.3	Comparison of Characteristic Assessment by Specific	65
	4.4	Test Methods In Situ Test Methods Available which Simulate the Repository	71
5.0	REASO	NABLE IN SITU TEST PROGRAMS	85
	5.1	Program Considerations	85
		5.1.1 Testing Which is Expected to Precede	85
		In Situ Test Program 5.1.2 In Situ Test Program Objectives 5.1.3 Tentative Media Independent In Situ Test	85 90

Program

Table of Contents continued.

			Page
	5.2	Tentative Media/Site Specific In Situ Test Programs Example In Situ Test Facility for Basalt at Hanford, Washington	94 98
6.0	RECOMM	ENDED IN SITU TESTING RESEARCH	101
	6.1 6.2	Introduction Potential Improvements to Existing In Situ Tests	101 101
		 6.2.1 Plate Test 6.2.2 Block Test 6.2.3 Chamber Test 6.2.4 Mine-by Test 6.2.5 Heater Test 6.2.6 Tracer Test 6.2.7 Multiple Borehole Permeability Test 6.2.8 Overcoring 6.2.9 Flatjack Test 6.2.10 Acoustic Emission Monitoring 6.2.11 Exposure Mapping 	101 102 103 104 104 105 105 106 106 107
	6.3 6.4	Development of New or Hybrid Tests Integration of In Situ Test Program in Repository Development	107 108
7.0	SUMMAR	AND CONCLUSIONS	109
GLOS	SARY		113
	G.1 G.2	General Terms Common Rock Mass Response Characteristics	113 118
		G.2.1 Mechanical G.2.2 Thermal G.2.3 Thermomechanical G.2.4 Hydrologic G.2.5 Geochemical	118 124 126 126 129
	G.3	Common Instrumentation Used for In Situ Testing	131
		 G.3.1 Strain and Displacement G.3.2 Stress G.3.3 Temperature G.3.4 Groundwater Pressure G.3.5 Data Acquisition 	131 152 161 163 169

Table of Contents continued.

Page

G.4	Common Activi	Site Preparation and Characterization ties for In Situ Testing	173
	G.4.1 G.4.2 G.4.3	Excavation Drilling, Sampling, and Logging Geophysical Seismic Surveys	173 177 185
REFERENCES			189

VOLUME II - APPENDICES

- APPENDIX A SPECIFIC IN SITU TEST DESCRIPTIONS, EVALUATIONS, AND RECOMMENDATIONS
- APPENDIX B PRESENT ASSESSMENT OF MEDIA/SITE SPECIFIC CHARACTERISTICS
- APPENDIX C EXISTING MEDIA/SITE SPECIFIC IN SITU TESTING RELATED TO HLW DISPOSAL

LIST OF FIGURES

No.	Title	Page
ES-1	Acceptable Level of Confidence in Satisfactory Performance	viii
ES-2	Task 2 Activity Flow Chart	xii
P-1	Definitions of "Compliance with Performance Criteria"	3
P-2	Maximizing the Level of Confidence in Satisfactory Performance	6
P-3	Repository Development	8
P-4	Acceptable Level of Confidence in Satisfactory Performance	10
P-5	Assessment of Level of Confidence in Satisfactory Performance (Example)	11
1.1	In Situ Test Program Within Repository Development	17
1.2	Task 2 Activity Flow Chart	21
3.1	Sensitivity of a Response Characteristic to an Assessment Variable (Example)	39
3.2	Scale Effect on Measured Values of a Response Characteristic	42
3.3	Probability Distribution for Response Characteristic . (Example)	46
3.4	Uncertainty in Estimation of Value of Characteristic	47
3.5	Process of Evaluating Significance of Characteristics . for In Situ Testing as They Relate to Design	51
5.1	Example In Situ Test Facility Configuration for Basalt	99
	Figure No. ES-1 ES-2 P-1 P-2 P-3 P-4 P-5 1.1 1.2 3.1 3.2 3.3 3.4 3.5 5.1	Figure No.TitleES-1Acceptable Level of Confidence in Satisfactory PerformanceES-2Task 2 Activity Flow ChartP-1Definitions of "Compliance with Performance Criteria"P-2Maximizing the Level of Confidence in Satisfactory PerformanceP-3Repository DevelopmentP-4Acceptable Level of Confidence in Satisfactory PerformanceP-5Assessment of Level of Confidence in Satisfactory PerformanceP-5Assessment of Level of Confidence in Satisfactory Performance (Example)1.1In Situ Test Program Within Repository Development1.2Task 2 Activity Flow Chart3.1Sensitivity of a Response Characteristic to an Assessment Variable (Example)3.2Scale Effect on Measured Values of a Response Characteristic3.3Probability Distribution for Response Characteristic (Example)3.4Uncertainty in Estimation of Value of Characteristic3.5Process of Evaluating Significance of Characteristics - for In Situ Test Facility Configuration for Basalt

List of Figures continued.

Figure No.	Title	Page
G-1	Idealized Creep Curve and Strain Recovery for a Rock Material	122
G-2	Strength of a Brittle Material	123
G-3	Borehole Extensometers	135
G-4	Two-Anchor Extensometer	137
G-5	Portable Borehole Inclinometer	139
G-6	Rod Type Multiple Position Borehole Deflectometer	141
G-7	Electrical Resistance Strain Gauges	143
G-8	Vibrating-Wire Strain Gauges	145
G-9	Linear Displacement Transducers	147
G-10	Mechanical Strain Gauge Reference Holes and Calibration Bar	149
G-11	Tunnel Convergence Gauges	151
G-12	Vibrating Wire Stressmeter	154
G-13	U.S. Bureau of Mines (USBM) Borehole Deformation Gauge	156
G-14	CSIR "Doorstopper" and Accompanying Setting Head	158
G-15	CSIR Triaxial Strain Cell	159
G-16	CSIRO Hollow Inclusion Triaxial Strain Cell and Installation Apparatus	160
G-17	Piezometer Installed from a Tunnel	164
G-18	Hydraulic Diaphragm Piezometer	167
G-19	Pneumatic Diaphragm Piezometer	168
G-20	Electrical Sensor Diaphragm Piezometers	170

List of Figures continued.

Figure No.	Title	Page
G-21	Drill Jumbo	174
G-22	Road Header Tunneling Machine	178
G-23	Underground Diamond Drill	179
G-24	Seismic Survey Techniques	187

LIST OF TABLES

Table No.	Title	Page
ES-1	Significant Characteristics	xiv
ES-2	Reasonable In Situ Test Programs	xvii
1.1	Present Status of DOE's Repository Site Screening/ Selection Program	18
2.1	Critical Repository Design and Construction Engineering Variables	28
2.2	Key Issues Affecting Choice of Values of Critical Shaft Engineering Variables	31
2.3	Key Issues Affecting Choice of Values of Critical Tunnel/Cavern Engineering Variables	32
2.4	Key Issues Affecting Choice of Values of Critical Waste Package Emplacement Hole Engineering Variables	33
3.1	Characteristics Which Impact the Resolution of Key Issues	36
3.2	Sensitivity of Response Characteristics to Assessment Variables	41
3.3	Acceptable Level of Uncertainty in the Assessment of Each Characteristic	53
3.4	Significance of Characteristics for In Situ Testing as they Relate to Design (Basalt-Hanford, Washington)	55
3.5	Significance of Characteristics for In Situ Testing as they Relate to Design (Tuff - Yucca Mountain, Nevada)	56
3.6	Significance of Characteristics for In Situ Testing as they Relate to Design (Domal Salt - Gulf Coast Sites)	57
3.7	Significance of Characteristics for In Situ Testing as they Relate to Design (Bedded Salt - Unspecified Site)	59
3.8	Significance of Characteristics for In Situ Testing as they Relate to Design (Granite - Unspecified Site)	60

List of Tables continued.

Table No.	Title	Page
4.1	Present Status of Site Specific In Situ Test Programs and Media Specific Test Facilities	62
4.2	NWTS Objectives of In Situ Test Facilities	63
4.3	Summary of Presently Proposed NWTS In Situ Test Program	64
4.4	Test Methods Available for Assessing Significant Characteristics	66
4.5	Limitations and Uncertainty in Assessment of Significant Characteristics by Available Test Methods	72
4.6	In Situ Test Methods Available to Resolve the Key Issues by Simulation	82
5.1	Testing Which is Expected to Precede In Situ Test Program	86
5.2	Tentative In Situ Test Program for Assessing Significant Characteristics	91
5.3	Tentative Media/Site Specific In Situ Test Programs	95
G-1	Common Rock Mass Response Characteristics	119
G-2	Summary of Strain and Displacement Monitoring Instrumentation	132
G-3	Summary of Stress Change Monitoring Instrumentation	153
G-4	Summary of Temperature Monitoring Instrumentation	162
G-5	Summary of Diaphragm Piezometers	166
G-6	Typical Costs of Data Acquisition System Components	172
G-7	Tunneling Cost Estimates	176
G-8	Drilling and Direct Logging Costs	180

PREFACE - LICENSING PERSPECTIVE

P.1 INTRODUCTION

It is the intent of this report to specifically address the requirements of in situ test programs for the development of deep geologic repositories for the permanent disposal of high level nuclear waste (HLW), so that the U.S. Nuclear Regulatory Commission (NRC) can adequately review U.S. Department of Energy (DOE) programs. In order to identify these requirements, it is first necessary to establish a <u>perspective</u> from which this will be accomplished. Too often, the perspective is not clear and disagreements regarding the conclusions are then often not easily resolved. In order to alleviate this problem, the perspective from which Golder Associates views repository development, and in situ testing in particular, is discussed in this Preface. The requirements of in situ test programs can then be identified and discussed on a logical basis.

The following topics are addressed within this Preface:

- Overview of high level nuclear waste program (Section P.2)
- Site selection (Section P.3)
- Repository design (Section P.4)
- Demonstrability (Section P.5)

P.2 OVERVIEW OF HIGH LEVEL NUCLEAR WASTE ISOLATION PROGRAM

High level nuclear waste (HLW) presently exists and in relatively large volumes. This waste is toxic, with its toxicity lasting for thousands of years. This toxicity may lead to adverse health effects and thus HLW must be isolated from the accessible environment during the period of its toxicity.

Compared to other forms of permanent disposal of HLW, a <u>deep geologic</u> repository appears to be optimum in terms of:

- Isolation (health effects)
- Feasibility
- Timeliness
- Cost-effectiveness
- Permanence, requiring no perpetual maintenance
- Public concensus and acceptability.

A federal-level approach is presently being taken to HLW disposal in a deep geologic repository, with provisions for the concurrence of the states involved. The U.S. Department of Energy (DOE) is expected to design, construct, and operate any HLW disposal facility. The U.S. Environmental Protection Agency (EPA) is the federal agency which is expected to establish the ultimate performance criteria that any facility must satisfy. These ultimate performance criteria are presently embodied in the draft rule 40-CFR-191. The U.S. Nuclear Regulatory Commission (NRC) is the federal agency which is expected to regulate HLW disposal, i.e., ensure that the DOE facility will achieve certain performance objectives related to public safety. In this regulatory role, the NRC has produced the draft rule 10-CFR-60 to help ensure DOE's compliance with the EPA criteria, as well as other performance objectives.

The <u>performance objectives</u>, specifically related to public safety, which deep geologic repositories for permanent disposal of HLW must achieve can be summarized as:

- Short term construction and operation objective (through decommissioning, about 100 years) of minimizing hazards jeopardizing the safety of the public and personnel during repository construction and operation (including possibly retrieval and decommissioning activities).
- Long term waste containment and isolation objective (postdecommissioning, from about 100 to 10,000's years) of minimizing radionuclide flux (rate/unit area) to accessible environment and thus minimizing hazards jeopardizing public safety after decommissioning. This objective dictates maintaining a waste retrieval capability for a specified period after waste emplacement and prior to decommissioning, thereby providing the opportunity for verifying a sufficiently high probability of satisfactory long-term performance and also providing a contingency plan for demonstrated non-verification.

Performance criteria must be established in order to define and, where possible, quantify the performance objectives. These critiera, such as the NRC criteria (10-CFR-60), which incorporate the EPA criteria (40-CFR-191), or other interim criteria established by the DOE or its contractor, can be given either as (see Figure P-1):

- Deterministic, in which an absolute numerical limit (Xc) is specified for each quantitative performance indicator (X), so that (if small X is good) X must be less than Xc for compliance
- 2) Probabilistic, in which an acceptable level of confidence (Lc) is specified for not exceeding the numerical limit (Xc) by the actual value of the quantitative performance indicator (X), so that the probability or likelihood of (X<Xc), or P(X<Xc), must be greater than Lc for compliance.



Where the criteria are given deterministically, a determination must still be made, although implicitly rather than explicitly, of the level of confidence in not exceeding the deterministic limit. Hence, it is Golder Associates' opinion that a probabilistic approach, which quantifies and explicitly incorporates the uncertainties in performance prediction, should be taken where possible.

In either case, for both technical and political reasons, it will be necessary to maximize the level of confidence (i.e., increase the probability or likelihood) that the actual performance will be satisfactory. There are two primary ways of maximizing this level of confidence:

- Selecting a suitable repository site
- Appropriately designing, constructing and operating the repository.

Three questions are apparent:

- How can a suitable site for a repository be selected so as to maximize the level of confidence in satisfactory performance?
- 2) Once a site has been selected, how can the repository be designed and constructed at that site so as to maximize the level of confidence in satisfactory performance?
- 3) How can an acceptably high level of confidence in satisfactory performance, i.e., compliance with the performance criteria, be demonstrated at each step in repository development?

Each of these questions is addressed in the following three sections.

P.3 SITE SELECTON

The first question, related to site selection, which needs to be addressed, is:

How can a suitable site for a repository be selected so as to maximize the level of confidence in satisfactory performance?

Repository site screening and selection requires the assessment of certain nonquantitative factors and quantitative parameters, which describe the site and its properties. These site characteristics can then be evaluated with respect to a particular set of site suitability criteria, which are based on how those characteristics are expected to affect repository system performance. In addition, these site characteristics can be input into performance models, which assess the probability of satisfactory performance. Both approaches could be used to compare sites. However, Golder Associates believes that the selection of an optimum site requires that several of the top ranking sites be compared against specific performance criteria. Furthermore, the use of performance models in the ranking process will significantly enhance the demonstrability of the selection process.

Site investigation for site screening and selection focuses primarily on large-scale features and does not generally require low uncertainty in the assessment of the characteristics. At this stage it thus consists of tests conducted from the surface, within limited boreholes, and in the laboratory on samples obtained from the site (e.g., core from boreholes), as well as prior information. As a result, the assessment of characteristics may have widely varying levels of uncertainty at this stage. It is possible that the assessment of certain characteristics at this stage could have low uncertainty and thus require no further information at subsequent stages.

P.4 REPOSITORY DESIGN

The second question, related to repository design, which needs to be addressed, is:

How can the repository be designed and constructed at the selected site so as to maximize the level of confidence in satisfactory performance?

Design and construction of a repository at the selected site will consist of selecting values for the engineering variables of each repository component, so as to optimize both the predicted performance and cost. Generally, once the site has been selected, the level of confidence in satisfactory performance will be maximized by a combination of (see Figure P-2):

- conservative design (increased margin of predicted performance over performance limits)
- decreased uncertainty in predicted performance (by decreasing uncertainty in predictive model and input, and by careful construction).

Both of these can be achieved, but only at increasing cost.

In order to optimize repository design and construction in terms of predicted performance and cost, it will be necessary to focus on those engineering variables which have a potentially significant impact on compliance with performance criteria at reasonable cost, and in addition can truly be varied (i.e., are not predetermined and set). An example of such an engineering variable is the excavation and associated support method, which may have a significant impact on performance. This is because such variables will affect the transport properties of a potentially important pathway for radionuclide escape. Once those significant engineering variables have been identified, key issues can be identified. Some of these key issues can be resolved by extrapolation of the results of prototype simulation (e.g., construction



(decreased uncertainty in predicted performance) P(x)dx [good bound] < $\leq \langle$ or p (x) dx D(X)dx [good bound] [good bound] (combinistion of decreased p (x)dx uncertainty in predicted performance and conservative [good bound] (conservative design) design)

38

Eng

78.07 .100

AP.3.463

Dwg. No.

6
of trial sections), whereas others require the assessment of specific site characteristics for input to predictive numerical models. Both approaches will be essential to the resolution of the key issues.

Significant characteristics of the site, i.e., those which resolve key issues affecting the choice of values of significant engineering variables, must generally be assessed with considerably greater certainty than for site screening/selection. However, if the uncertainty is already sufficiently low, then no additional effort may be required. Hence, site characterization should generally update the assessment of site characteristics for site screening/selection, especially by tests conducted in an in situ test facility. In addition to assessing site characteristics, this in situ test facility can serve as a simulation of the repository, and thereby resolve key issues.

In situ tests to be conducted in this facility are expected to reduce the uncertainty in the assessment of the significant characteristics. This reduction in uncertainty will be provided primarily by directly testing a representative volume of rock mass which is large enough to contain a significant number of discontinuities. Tests on a representative volume do not require scale-effect corrections in the determination of quasi-continuum rock mass characteristics, whereas smaller scale tests (e.g., borehole, laboratory, or small-scale in situ tests) assess either the intact rock or the discontinuity characteristics, and not the composite rock mass characteristics. Due to their generally high cost and long duration, however, the results of large-scale in situ tests must be used to develop and verify site-specific correlations with the results of tests which are less expensive and of short duration. Once reliable correlations have been determined, these simpler tests can be used with increased confidence in assessing the significant characteristics throughout the repository.

P.5 DEMONSTRABILITY

The third question, related to demonstrability, which must be addressed, is:

How can an acceptably high level of confidence in satisfactory performance, i.e., compliance with the performance criteria, be demonstrated at each step in repository development?

This question essentially defines repository development (see Figure P-3). The NRC must assess the level of confidence in satisfactory performance at various discrete decision points in repository development, especially at LA. If the confidence level is found to be acceptably high, the appropriate authorization or license amendment necessary to initiate the next phase of repository development is granted. However, if the level of confidence in compliance is found to be unacceptably low, then one of two approaches can be taken by the applicant:



- If the level of confidence can be increased by additional cost effective characterization or design modification, then that characterization or modification may be performed and the application updated
- (2) If the level of confidence cannot be cost effectively increased by further characterization or design modification, then repository development should be stopped at this site.

The discrete points in repository development are:

- Site characterization report (SCR) submittals, primarily DOE decision points for which NRC offers only opinions
- (2) License application (LA)
- (3) Updated application for license to operate (emplace waste)
- (4) License amendment to decommission.

The acceptable level of confidence in satisfactory performance, especially compliance with the ultimate (long-term) criteria, will increase with repository development, i.e., from initial SCR submittal to decommissioning (see Figure P-4). It is presently perceived that the acceptable level of confidence should be high at LA in order to provide reasonable assurance that the repository will be fully licensable, prior to extensive development and major expenditure of funds. If this were not a concern, then the acceptable level of confidence in satisfactory performance could be lower at LA, and the information gained after LA relied upon to improve the level of confidence sufficiently for decommissioning.

The assessment of the level of confidence in satisfactory performance, relative to the acceptable level, at each decision point will affect subsequent repository development activities (Figure P-3) and will determine the nature and extent of subsequent in situ testing and construction and operation monitoring. For example, if the level of confidence in satisfactory performance was high enough at the initial SCR submittal, due possibly to conservative and reliable engineered components and/or predictable geology/geohydrology, in situ testing prior to LA might not be necessary. Conversely (see Figure P-5), if the level of confidence in satisfactory performance at the time of the initial SCR submittal was perceived to be insufficent for granting construction authorization, in situ testing would be necessary to sufficiently improve the level of confidence in satisfactory performance. An exploratory shaft supplemented by surface tests might improve the level of confidence in satisfactory performance, but still not enough for granting a construction authorization. An underground test facility would thus be necessary, prior to LA, in order to sufficiently improve the level of confidence in satisfactory performance. The results of tests after LA, as well as construction and operation monitoring, could then sufficiently improve the level of confidence in satisfactory performance for subsequent licensing steps.





Dwg No. A813 (163

B,R

Eng

28

-0/

Dete

11

Acceptable levels of confidence may be established either:

- Implicitly, through progressive technical discussion between the regulator and applicant
- Explicitly, through rigorous decision-making utilizing quantitative risk assessment methodology (e.g., Roberds, 1981). Such a risk assessment includes explicit consideration of the consequences of not satisfying the performance objectives/criteria, as well as the risks in other alternatives for HLW disposal.

Although recognizing the difficulties in performing quantitative risk assessments and subsequently defending the results in the decisionmaking process (especially within the institutional framework), Golder Associates considers it important to be attempting to utilize uncertainty and probability assessments of the important parameters affecting repository performance. These assessments should be used as a guidance tool during the site characterization phase. While these probabilistic approaches are being established, the current approach of assessing acceptability implicitly through technical discussions needs to be continued.

The required demonstration of compliance with the performance criteria can be provided at any step in repository development by:

- Prediction (with inherent uncertainties) of future performance, based on numerical modeling or extrapolation of physical simulation
- Verification of past performance, based on monitoring performance.

Monitoring of repository system performance will not be possible, however, until later stages of repository development and, even then, complete verification of compliance with long-term performance criteria for waste isolation will not be possible. Hence, to demonstrate compliance with performance criteria, either numerical models, which rely on site characteristic assessments and the proposed repository design, or physical simulations must be utilized to predict repository performance, with some associated uncertainty.

The uncertainty in predicted performance must generally be decreased with each successive licensing step in order to demonstrate compliance. This can be accomplished by:

 Decreasing the uncertainty in the assessment of significant site characteristics

This can be accomplished by continually updating the assessment, e.g., by additional site investigation, by in situ testing, by construction as more areas are exposed, and by performance monitoring. Decreasing the uncertainty in predictive numerical models

This can be accomplished by comparing predicted large-scale in situ test results with the actual results and by comparing predicted performance of the in situ test facility or repository with the results of performance monitoring.

Improving the correlation between simulation tests and the prototype

The uncertainty in the extrapolation of results from a test case to the prototype will be a function of the degree of similarity between the two. Thus, as this similarity is increased (i.e., by making the test case as similar as possible to the prototype in terms of site characteristics, test conditions, and design/ construction), the correlation will be improved and the uncertainty reduced.

Updating performance predictions

The uncertainty in performance prediction increases with the time of projection. Performance monitoring allows for some previous predictions to be adjusted to agree with the actual performance at the time of measurement, and thus progressively eliminate measured errors and effectively reduce the projection time of subsequent projections. However, due to the relatively short operating period, this updating will be only marginally effective in reducing the uncertainty in predicting very long term performance.

P.6 SUMMARY

5

*

It is anticipated that certain testing/monitoring methods will need to be utilized in order to demonstrate an acceptably high level of confidence in satisfactory performance at each step of repository development (Figures P-3 and P-4). This perspective of the licensing process can be summarized as follows:

- For site screening/selection, as summarized in the initial SCR submittal, site investigation will focus generally on large-scale features and consists of surface, borehole and laboratory testing. This generally rough assessment of characteristics will allow for comparison between potential sites and for the possible demonstration of an acceptably high level of confidence in satisfactory performance based on a conceptual repository design.
- For a detailed repository design, as summarized in a license application (LA) for construction authorization, site characterization will focus on features adjacent to planned shafts and the repository horizon, as well as on large-scale features. This site characterization effort will consist primarily of the construction and operation of an in situ test facility. In addition to the improved assessment of site characteristics provided by in situ

testing and by monitoring of the in situ test facility, certain aspects of key issues may be adequately resolved by prototype simulation and predictive numerical models may be partially verified. The construction and operation of this in situ test facility may thus provide information for the detailed repository design and for the possible demonstration of an acceptably high level of confidence in satisfactory performance with that design.

- In an <u>updated application for license to operate (emplace waste)</u> subsequent to construction, the assessment of site characteristics will be further refined as more areas underground are exposed and as the performance of the repository is monitored. Additional verification of predictive numerical models and updating of performance predictions may be provided by monitoring. This monitoring may also provide information for modifications in design (if required) and for the possible demonstration of an acceptably high level of confidence in satisfactory performance by the constructed repository.
- Finally, in a license amendment to decommission, the assessment of site characteristics will be further refined by monitoring repository performance. Monitoring may also provide additional verification of predictive numerical models and updating of performance predictions. A final assessment of the level of confidence in satisfactory long-term performance will then be made, and a determination made by the NRC as to whether this likelihood is high enough. If it is determined that the likelihood is too low, modifications to the repository or retrieval of the waste will be required.

Hence, the assessment of site characteristics is a continual process, from site screening/selection to decommissioning, of reducing uncertainty in the assessment of those characteristics which have a significant impact on compliance with the performance criteria. Similarly, the verification of predictive numerical models is also a continual process, and consists of reducing their uncertainty.

INTRODUCTION

1.1 INTRODUCTION

This report represents the results of Task 2, "In Situ Test Programs Related to Design and Construction of High Level Nuclear Waste (HLW) Deep Geologic Repositories," of U.S. Nuclear Regulatory Commission (NRC) Contract NRC-02-81-037, "Technical Assistance for Repository Design."

The purpose of the complete project is to provide NRC with technical assistance for the following reasons:

- To enable the focused, adequate review by NRC of aspects related to design and construction of an in situ test facility* and final geologic repository, as presented in U.S. Department of Energy (DOE) Site Characterization Reports (SCR)
- To ascertain that the DOE site characterization will provide, as far as possible, all the information necessary to permit a review to be conducted by NRC of a License Application (LA).

This report presents and utilizes a defensible rationale to tentatively identify those in situ tests (if any) which constitute reasonable media/ site specific in situ test programs for adequately responding to the perceived information needs for construction authorization. These information needs are related to the additional information needed after initial SCR submittal (and site investigation) to adequately resolve the key issues of repository system performance for construction authorization at each site. It is expected that the in situ test program will evolve with time somewhat independently for each media/site considered as the perceived information needs and test capabilities develop. The program and design of specific tests must thus be flexible enough to take into account new information which becomes available during program performance, as presented in SCR updates. In addition. scoping recommendations have been made regarding (1) how specific in situ tests should be conducted, (2) research and development which may effectively improve the capabilities of the test program, and (3) the utilization of the Task 2 results by the NRC in their licensing review process.

It has been assumed that the in situ test program will be conducted within an in situ test facility, consisting of an exploratory shaft, extending from the surface to the prospective repository horizon possibly with test stations at various depths, and an underground test facility, consisting of appropriate tunnels and test rooms at that horizon. Plans for the program are expected to be presented in the initial SCR submittal and complete results presented in the LA.

*Specific terms utilized are defined in attached Glossary.

1.0

1.2 APPROACH

This study, within the context of the current licensing perspective (see Figure 1.1), involves the development of general recommendations for in situ testing, with variations for each of the specific media and sites under consideration. The media and sites specifically considered in this study include (see Table 1.1):

- Basalt at Hanford Reservation, Washington
- Tuff at Yucca Mountain, Nevada Test Site, Nevada
- Domai salt at Richton or Cypress Creek, Mississippi, or Vacherie, Louisiana
- Bedded salt at unspecified site (generic)
- Granite at unspecified site (generic).

The descriptions of domal salt and tuff media/sites were the subjects of Task 1 of this project (Golder Associates, 1982a and b, respectively) and the basalt, granite, and bedded salt media/sites were the subjects of other and earlier work (especially Golder Associates 1979a and c, and 1981).

This study is limited to consideration of tests conducted within an in situ test facility in the time frame between the initial SCR submittal and LA (i.e., site characterization). Testing preceding initial SCR submittal (i.e., site investigation during site screening/selection) is outside the scope of work and thus has not been evaluated; SCR requirements are given in the "Standard Format and Content of Site Characterization Reports for High-Level Waste Geologic Repositories" (NRC, 1981). Testing or monitoring subsequent to LA (i.e., during repository construction/operation) is also outside the scope of work.

A <u>defensible rationale</u> has been developed and utilized to tentatively select available tests to be included in the media/site specific in situ test programs. This rationale essentially consists of:

- Establishing the information needs for construction authorization at each site
- Assessing the capabilities of available tests to meet the specific information needs
- Matching the capabilities of specific tests to the perceived information needs at each site.

The information needs at any time result from the uncertainties in the prediction of repository system performance, and consist of the additional information needed in order to predict satisfactory performance with the required level of confidence. Information needs are determined as follows:

IN SITU TEST PROGRAM WITHIN REPOSITORY DEVELOPMENT

R

82

2



PRESENT	STATUS	OF DO	E'S RE	POSITORY	SITE
SCREENIN	G/SELE	CTION F	ROGR	AM	

Table 1.1

	Basalt Waste Isolation Project (BWIP), DOE prime contractor - Rockwell:
	The prospective repository horizon is in <u>basalt</u> flow (presumably the Umtanum, which is about 80 feet thick and 3800 feet deep) in Cold Creek Syncline at Hanford Reservation, Washington; SCR submittal is expected in late 1982, exploratory shaft is expected to be initiated in 1983.
•	Tuff at Yucca Mountain, Nevada Test Site, Nevada Nevada Nuclear Waste Storage Investigations (NNWSI), DOE prime contractor - Sandia:
	The prospective repository horizon is in <u>tuff</u> (possibly the Bullfrog formation, which is about 400 feet thick and 2100 feet deep, but 3 other formations are being studied to similar levels) under Yucca Mountain at Nevada Test Site (NTS), Nevada; SCR submittal is expected in 1983, exploratory shaft is expected to be initiated in 1983.
•	Domal Salt at Richton or Cypress Creek, Mississippi, or Vacherie, Louis Gulf Coast Domal Salt Investigation, DOE prime contractor - ONWI*:
	A choice between Vacherie Dome in Louisiana, Richton Dome and Cypress Creek Dome in Mississippi is expected to be made in 1983; a subsequent choice between the selected dome and the selected salt basin is expected to be made in 1983 for exploratory shaft initiation.
•	Bedded Salt at unspecified site (generic) Bedded Salt Investigation, DOE prime contractor - ONWI*:
	A choice between the Paradox Basin in Southeastern Utah and the Permian Basin in Northwest Texas is expected to be made in 1983; a subsequent choice between the selected basin and the selected salt dome is expected to be made in 1983 for exploratory shaft initiation.
	Granite at unspecified site (generic)
	Granite Investigation, DOE prime contractor - ONWI*:
	The investigation and site selection process for a suitable site in granite is expected to begin throughout the eastern portion of the U.S. in 1982; SCR submittals for as many as three selected sites are expected in about 1986.
* 01	WI - Office of Nuclear Waste Isolation is managed for DOE by Battelle-Columbus
N	lote: This perceived status is as of November 1982, and is subject to

ę

- Identify the existing information and assess the associated level of confidence in satisfactory repository system performance
- Compare the assessed level of confidence with the acceptable level, either implicitly or explicitly
- Determine what additional information is needed to raise the level of confidence in satisfactory performance to the acceptable level, by:
 - establishing the relationship between each component of the repository system and system performance (i.e., sensitivity)
 - identifying where the existing information regarding significant components of the system is insufficient and can be readily supplemented (i.e., where the existing uncertainty is large, but can be effectively reduced).

The information needs at any time are thus a function of:

- The significance of each component of the repository system (including site characteristics) with respect to system performance
- The currently available information, which in conjunction with the repository design determines the level of confidence in satisfactory repository system performance
- The acceptable level of confidence in satisfactory repository system performance for each licensing step.

Regarding the above, performance assessment (including sensitivity studies) is outside the scope, so that the significance of repository system components cannot be quantitatively evaluated and the level of confidence in satisfactory performance cannot be assessed. Also, the existing information is not constant, but continually being supplemented and updated. Determination of the acceptable level of confidence in satisfactory performance at each step is a part of the licensing perspective, and is outside the scope. Thus, for this study, tentative information needs must be established based on Golder Associates' premises, perceptions, and interpretations of available information.

The <u>capabilities</u> of <u>available tests</u>, with respect to responding to the information needs, must be assessed. These tests satisfy the information needs either by:

- Appropriately simulating various aspects of the actual repository (e.g., construction techniques) for extrapolation of results
- Adequately assessing identified media/site specific characteristics (e.g., hydraulic conductivity) to be used in numerical modeling

Sufficiently verifying predictive numerical models.

However, the development of new or hybrid tests (i.e., incorporating new concepts or combining available concepts) is outside the scope, so that only presently available tests and potential advancements to the state-of-the-art (i.e., expanding available concepts or improving technology) are considered. Areas are also pointed out where research and development might improve either individual test capabilities or the combined capabilities of a test program in better responding to the perceived information needs. Hence, test capabilities may improve with time.

From a comprehensive list of available tests and associated capabilities, appropriate in situ tests can be identified which adequately respond to the perceived information needs for construction authorization at each media/site considered. These tests must then be integrated into a reasonable in situ test program. An in situ test facility, which can accommodate such reasonable in situ test programs, can then be developed.

The actual media/site specific in situ test program will be a function of the information needs and test capabilities at that time. Hence, it is expected that the test program will evolve with time somewhat independently for each media/site considered as the perceived information needs and test capabilities develop. It is even possible that the information needs for construction authorization might be satisfied prior to the initial SCR submittal, precluding the need for most (if not all) in situ testing. The program must thus be flexible enough to take into account new information which becomes available during its performance, as presented in SCR updates. Also, the complete design and specifications for any particular test cannot be accomplished a priori without detailed descriptions of each individual test location, as well as identification of specific information needs. Hence, the recommendations for the conduct of specific tests are of a scoping nature only.

1.3 STUDY ACTIVITIES

The following activities have been undertaken in the course of developing recommendations for in situ testing (see Figure 1.2):

• A licensing perspective has been established (see Preface). The various licensing steps in repository development have been identified, based on the procedural rule of 10-CFR-60. At each step, the level of confidence in future satisfactory repository system performance should be assessed, and a determination made of whether that level of confidence is sufficiently high to allow further development (Figure 1.1). These steps are:



- (1) SCR submittals
- (2) License application (LA)
- (3) Updated application for license to operate (emplace waste)
- (4) License amendment to decommission.

The acceptable level of confidence at each step underlies the establishment of the information needs.

- Critical engineering variables, which in conjunction with the site characteristics will determine repository system performance, have been identified (see Chapter 2, Section 2.1). This has been accomplished by first identifying the primary engineering variables of repository design and construction (e.g., shaft dimension, shape, etc.). The impact of these primary engineering variables on the level of confidence in satisfactory performance has been qualitatively assessed and those perceived to have a significant potential impact, in conjunction with a capability for change at reasonable cost, have been considered to be critical. These critical engineering variables should be investigated by in situ testing, as well as emphasized during NRC's review process.
- Key issues, essentially related to compliance with the various aspects of the short-term construction/operation and the long-term waste containment/isolation performance critiera, have been identified (see Chapter 2, Section 2.2). These key issues impact the choice of values for the critical engineering variables and must be adequately resolved to demonstrate an acceptably high level of confidence in satisfactory performance. Hence, these key issues must be addressed in both the SCR and the LA review. Information needs are related to the adequate resolution of these key issues.

These key issues can typically be resolved either by extrapolating the results of appropriate prototype simulation or by adequately assessing specific site characteristics and then incorporating them in sufficiently verified numerical models.

• Characteristics of a site which are perceived to have a significant impact on the resolution of the key issues have been identified (see Chapter 3). The assessment of each characteristic entails some uncertainty, due to natural variability as well as due to the quality of the data base on which the interpretaton is based. The sources of uncertainty in this assessment have been identified. In addition, response characteristics may be anisotropic, scaledependent, time-dependent, and a function of present and past environmental conditions (i.e., stress level, pore pressure, temperature, radiation dose). Ignoring the effect of these conditions results in greater uncertainty. Hence, the relationship of each of the significant characteristics to these environmental conditions has been qualitatively determined, based on experience. Each of the characteristics has subsequently been evaluated in terms of its influence on satisfying the performance criteria. This evaluation has taken into account the following attributes of each characteristic:

- availability of cost-effective design and construction techniques which allow for a conservative assumption of the value of the characteristic
- uncertainty in the representation of the phenomenological laws of nature by the performance prediction model
- sensitivity of the performance prediction model to the value of the characteristic
- cost effectiveness and scheduling limitations of measures to reduce the uncertainty in the assessment of the characteristic.

From this evaluation, a qualitative determination has been made regarding the maximum acceptable level of uncertainty in the assessment of each of the characteristics. However, these levels are not independent of the magnitude of characteristic values (individually or combined), the sensitivity of performance to all the system components, or the acceptable level of confidence in satisfactory repository system performance.

- The present assessment of the value of significant characteristics at each media/site being considered, based on available information, has been summarized (see Chapter 3, Section 3.5 and Appendix B of Volume II). For tuff and domal salt sites, this has been based primarily on the results of Task 1 of this project (Golder Associates, 1982a and b). For basalt, bedded salt and granite sites, this has been based on earlier work (Golder Associates, 1979a, b, and c), complementary work (Golder Associates, 1981 and 1982c), work by others, and past experience. These assessments have been presented in terms of a best estimate and an indication of the level of uncertainty in that value.
- Test methods which are available to assess the significant characteristics have been identified (see Chapter 4). This has been accomplished by first listing all of the common test methods, including surface tests, borehole tests, laboratory tests, and in situ tests, available for assessing each characteristic (see Section 4.2). Existing repository-related in situ testing programs have been summarized to assist in the identification of available in situ tests (see Section 4.1 and Appendix C of Volume II).

The <u>capabilities</u> of each of the available test methods regarding its determination of the significant characteristics has been assessed (see Section 4.3). This assessment has been based partly on the test method's incorporation of those environmental conditions which affect the characteristic, as well as whether a representative volume of rock mass is tested. These test capabilities are not independent of either the magnitude of the characteristic value or media.

- Those in situ tests which can be used to resolve the key issues by simulation have been identified (see Chapter 4, Section 4.4). These in situ tests, as well as the in situ test facility itself, simulate various construction/ operation aspects of the repository, so that their results can be extrapolated to directly predict repository performance.
- The testing which will precede in situ testing has been anticipated (see Chapter 5, Section 5.1.1). The information obtained by this testing at sites where it is now incomplete will supplement the existing information, and the present assessments of significant characteristics may change prior to the initial SCR submittal and subsequent initiation of in situ testing.
- The information needs for construction authorization, which are perceived to exist at the time of the initial SCR submittal, have been identified (see Section 3.5.2). This has been accomplished by identifying the existing media/site specific assessment of significant characteristics and the anticipated additional site investigation, and comparing that level of information expected at initial SCR submittal with what is perceived to be required for construction authorization.
- Tentative media/site-specific in situ test programs have been recommended (see Chapter 5). These programs consist of available in situ test methods which best respond to the perceived information needs for each of the media/sites, either by simulation or by adequate assessment of the significant characteristics. Where no one appropriate method exists, a combination of independent methods which adequately responds to the perceived information needs has been identified.

It is expected, however, that these tentative programs will evolve, even during their performance, as information needs and even test capabilities develop.

An example in situ test facility which can accommodate these programs has been developed under Task 4 of this project (see Section 5.3).

• Scoping recommendations have been made regarding specific in situ tests (see Appendix A of Volume II). These tests have been investigated in detail, and a description, an evaluation, and a recommendation regarding methodology and utilization of results have been presented for each. However, it is expected that the design of each test will evolve as the information needs are defined, more information regarding the site becomes available, and testing technology improves. Potentially effective research and development of in situ testing techniques have been identified and recommended (see Chapter 6). These recommendations are primarily concerned with advancements to the state-of-the-art (i.e., expanding available concepts or improving technology), although the development of new or hybrid tests (i.e., incorporating new concepts or combining available concepts) and program integration have been addressed.

2.0 KEY ISSUES FOR REPOSITORY DESIGN AND CONSTRUCTION

2.1 CRITICAL REPOSITORY DESIGN AND CONSTRUCTION ENGINEERING VARIABLES

The repository must be designed, constructed, and operated so as to demonstrably satisfy established performance criteria (see Preface -Licensing Perspective). The repository system consists of engineered components and the geologic site, which has certain characteristics. It is the integration of engineered components and site characteristics which will determine the level of confidence in satisfactory performance.

The primary engineered components of the repository, which must be designed and constructed so as to optimize predicted performance and cost. are:

- Surface facilities
- Underground facilities, including engineered barriers (large-scale) - shafts

 - tunnels/caverns
 - (small-scale)
 - waste packages.

Surface facilities are outside the scope of this report, and only underground facilities, not including engineered barriers, have been considered. The waste package itself is also outside the scope of this report, but the waste package emplacement hole has been considered.

Once the conceptual repository design has been completed (as presented in the SCR), only certain aspects in the design and construction of each repository component can be varied to achieve optimization. Those engineering variables which are perceived to have a significant and cost effective impact on the level of confidence in satisfactory performance have been termed "critical." These critical engineering variables (see Table 2.1) should be focused on, for in situ testing, design, performance assessment, and NRC review. These variables have been assumed to be essentially media/site independent.

2.2 KEY ISSUES WHICH IMPACT THE CHOICE OF THE VALUES OF CRITICAL ENGINEERING VARIABLES

The key issues which will affect the choice of the value of each critical engineering variable (Table 2.1), and hence the level of confidence in satisfactory performance, can be separated into those related to:

 Purpose of each repository component, as defined by the repository conceptual design

2

	CI	RITICAL ENGINEERING V	ARIABLES			
	(Large scale)		(Small scale)			
REPOSITORY STAGE	SHAFTS	TUNNELS/CAVERNS	WASTE PACKAGE EMPLACEMENT HOLES			
Pre-construction (Design)	number location (spacing) orientation depth size (x-section) shape (x-section)	depth number location (spacing) orientation length size (x-section) shape (x-section)	location orientation spacing hole diameter hole depth			
Construction *	excavation method support method/ requirements liner requirements	excavation method support method/ requirements liner requirements (if any)				
	dewatering ventilation hoisting	dewatering ventilation transport				
Operation *			hole drilling hole lining. waste package emplacement hole backfill/plug waste package corrosion protection waste package retrieval			
Post-Operation (Decommissioning)	backfill plugging	backfill sealing				

Table 2.1

(Excluding surface facilities and waste package)

* Construction and operation are sequential for any panel, but they are concurrent for the total repository.

- Engineering variable interrelationships (if any)
- Site characteristics.

The purpose of the repository component is defined by the repository conceptual design, and includes the following key issues:

- Usage (type) definition, e.g.-
 - conveyance of men, equipment, muck from mining, or waste packages
 - provision of ventilation, utilities, or dewatering
 - storage of equipment, muck, or waste packages
 - operating life
- Capacity requirements, e.g.-
 - dimensions of equipment
 - conveyance capacity
 - ventilation capacity
- Isolation requirements, e.g., separation of mining and waste storage operations
- Layout requirements or interrelationships between components, e.g., repository level conceptual layout affecting shaft location.

Key issues which are related to <u>site characteristics</u> include the following:

- Constructability. Can the facility be constructed in a timely and safe fashion, and so that it will not jeopardize the long-term waste containment/isolation capability of the facility? Both the unavoidable creation of a disturbed zone of rock around underground openings and the construction of engineered barriers will have an effect on the response of the repository.
- Thermal Response. Can the temperature field be adequately predicted as a function of time to use as input to the mechanical, hydrological and geochemical models?
- Mechanical Response. Can the stability and deformation of underground openings (including waste package emplacement holes) be adequately predicted for the periods of short-term construction/ operation and long-term waste containment/isolation?
- <u>Hydrologic Response</u>. Can an adequate prediction be made regarding the resaturation time of the repository (post-closure) and of the groundwater flow through the repository over the long term, including the potential impact of shafts? Of lesser importance is the question of the amount of inflow into the repository during operation, i.e., over the short term.

-0

 Geochemical Response. Can an adequate prediction be made of the nature and extent of geochemical alteration of the engineered barriers and the rock? Can the quantity and rate of migration of specific radionuclides over the long term be adequately predicted?

The above key issues are essentially related to predicting the short-term construction/operation performance and the long-term waste containment/isolation performance. Generally, they are a function of both the constructed repository and the site characteristics. The site characteristics include those of the rock mass around underground openings which have been disturbed by excavation, as well as those of the undisturbed rock mass. The constructed repository will also include engineered barriers (e.g., backfill, seals, and plugs). The interface between the barriers and the rock mass must also be considered.

The various key issues affecting the choice of the value of each identified engineering variable for shafts, tunnels/caverns, and waste package emplacement holes (Table 2.1) have been identified (see Tables 2.2, 2.3, and 2.4, respectively). These key issues are considered to be essentially media independent.

The key issues must be resolved in order to:

- Design and construct a repository (i.e., choose values for the critical engineering variables) for any of the media/sites being considered
- Demonstrate an acceptably high level of confidence in satisfactory performance for that repository design and site.

These key issues can be resolved either by:

- Adequately assessing certain media/site specific characteristics, which are used as input in predictive numerical modeling, and sufficiently verifying these models.
- Appropriately simulating various aspects of the repository, in order to extrapolate the results.

These two approaches will be discussed separately. Information needs, to which in situ test programs must respond, are related to the adequate resolution of these key issues.

KEY ISSUES AFFECTING CHOICE OF VALUES OF CRITICAL SHAFT ENGINEERING VARIABLES

Table 2.2

			PRE-	CONST	RUCT	ION		CONS	TRUCT	TION	CONSTRUCTION /OPERATION			POST-OPER	
	KEY ISSUE	Number	Location	Orientation	Depth	Size (X-Section)	Shape (X-Section)	Excavation Method	Support Method/Requirements	Liner Requirements	Dewatering Requirements	Ventilation Requirements	Hoisting Requirements	Backfill	Plugging
OF	Usage									•		•			
AFTS	Capacity Requirements					٠					•	٠	•		
SH	Isolation Requirements		٠								•	٠	•	•	•
	Layout Requirements		٠	•			1				•	•	•	•	•
	Number	X	•			•		1			•	•			•
	Location		X								•	•			•
	Orientation		•	X		•	•	•	•		•	•		•	•
dIH	Depth			•	X			•			•	•		1.5	11
IONS	Size (X-Section)					X	•	•	•	•	•	•		•	•
LAT	Shape (X-Section)	1.				•	X	•	•_	•	-	e d	_	•	•
ERRE	Excavation Method	1	•	•		•	•	X	•	•		•		•	•
INI	Support Method/Requirements					0	•	•	X	•		•		•	•
ABL E	Liner Requirements				-	•	•	+	•	×	•	_		•	
ARIA	Dewatering Requirements									•	X			1	
>	Ventilation Requirements	1:	•			1		1				^	×	1 .	
	Hoisting Requirements	-·	-	•	-		-	١.			+		^	Y	-
	BackTITT					1	1	1	1	1		1			
	Constructability	-		-		-	-	-		-	-		-		
TIC	Thermal Response	1.													
TED	Mechanical Pernonse														
SI	Hydrologic Response														
CHAR	Geochemical Response					1									

CRITICAL SHAFT ENGINEERING VARIABLE (See Table 2.1)

• Issue affects choice of engineering variables

Eng. P.K.

Dete 182

Rev. 012-1635

X Issue of variable interrelationship does not exist (identical)

31

RE-CONSTRUCTION CONSTRUCTION CONSTRUCTION Usage XEY ISSUE Depth Number Location Joest Usage Capacity Requirements Depth Number Location Joest Lysue Requirements Number Location Space (X-Section) Space (X-Section) Support Method/Requirements X Number Location Space (X-Section) Support Requirements X Number Location Space (X-Section) Support Method/Requirements X Number Number Number Liner Requirements X Number Number Number Unrefit ation Requirements X Number Number Number Unrefit ation Requirements X Number Number Number Number Unrefit ation Requirements X Number Number Number Number Unrefit ation Requirements X Number Number Number Numer Unrefit ation Requiremen
Image: Second
•••• •
A A A A A A A A A A A A A A A A A

10.82

32

KEY ISSUES AFFECTING CHOISE OF VALUES OF CRITICAL WASTE PACKAGE EMPLACEMENT HOLE ENGINEERING VARIABLES

Table 2.4

			PRE-CONSTRUCTION OPERATIO								ON			
	KEY ISSUE	Location	Orientation	Spacing	Hole Diameter	Hole Depth	Hole Drilling	Hole Lining	Waste Package Emplacement	Hole Backfill/Plug	Waste Package Corrosion Protection	Waste Package Retrieval		
AGE HOLES	Usage													
PACK	Capacity Requirements									-				
TEM	Isolation Requirements										÷.			
MAS	Layout Requirements													
-	Location	X												
	Orientation		X											
IHSHI	Spacing			X										
1104	Hole Diameter				X									
RELA	Hole Depth					X								
ITERI	Hole Drilling	-					X					-		
-	Hole Lining							х						
ABL	Waste Package Emplacement	-							x					
VAR	Hole Backfill/Plug									x				
	Waste Package Corrosion Protection										X			
	Waste Package Retrieval											X		
5	Constructability													
STI	Thermal Response													
ATE	Mechanical Response													
REL	Hydrologic Response													
E	Geochemical Response													

CRITICAL WASTE PACKAGE EMPLACEMENT HOLE ENGINEERING VARIABLE (See Table 2.1)

• Issue affects choice of engineering variables

Eng. 3.R

Date 1-82

X Issue of variable interrelationship does not exist (identical)

SIGNIFICANT CHARACTERISTICS WHICH IMPACT THE RESOLUTION OF KEY ISSUES

3.1 SITE CHARACTERISTICS

3.0

Certain characteristics of a site must be assessed in order to adequately resolve the key issues (see Table 3.1). These characteristics (see Glossary) can be summarized as those which describe the:

- Geologic setting of the site
- Response or behavior of the site

The geologic setting constitutes the geometry and boundary/ field conditions of the site, as well as the "physical" characteristics of the materials.

The geometry consists of the present stratigraphy/ lithology and structure of rock mass units. Each rock mass unit is considered to be relatively homogeneous and consists of intact rock intersected by discontinuities, such as joints, shears, and fractures, with pore fluid (i.e., liquid and/or gas) contained in spaces within the rock mass. Boundaries of rock mass units can be defined by changes in lithology or structure, i.e., either large scale faults or changes in discontinuity patterns.

The boundary/field conditions include the pre-excavation:

- In situ stress field
- In situ hydraulic head field, which defines the direction and magnitude of hydraulic gradients
- In situ temperature field.

The "physical" characteristics which describe each rock mass unit include:

- Mineralogy, texture (including microcracking, bedding, schistosity), porosity, and density of the intact rock
- Spacing, orientation, persistence, roughness/planarity, aperture, and nature of infilling or surface materials of the discontinuities
- Content and composition of pore fluid.

Also, the potential changes in the boundary/field conditions and corresponding changes in structure (i.e., tectonics) or physical characteristics, which are unrelated to repository development, must be discussed as part of the geologic setting.

The "response" characteristics, i.e., those which describe the mechanical, thermal, hydrologic, and geochemical response to any loading, of the rock mass are strongly related to the physical

CHARACTERISTICS WHICH IMPACT THE RESOLUTION OF KEY ISSUES

KEY ISSUES Response Response Hydrologic Response Response Constructability Geochemical Mechanical [herma] CHARACTERISTICS GEOLOGIC SETTING Stratigraphic/structural* Tectonic . 0 . . 0 In situ stress field 0 0 . 0 0 In situ hydraulic head field . 0 0 . 0 In situ temperature field . . 0 0 0 RESPONSE (MECHANICAL) Strength . . 0 0 Deformation 0 0 Creep/fusing 0 0 . 0 (THERMAL) Thermal conductivity 0 . Heat capacity . 0 Linear thermal expansion . (HYDROLOGIC) Hydraulic conductivity 0 Effective porosity . Specific storage (GEOCHEMICAL) Dispersivity . Adsorption/retardation . Alteration/solubility 0 0 0 . .

 includes the physical and chemical characteristics of each rock mass unit (including pore fluid composition).

characteristic significantly impacts resolution of key issues.

o characteristic impacts resolution of key issues, but to a lesser degree

Table 3.1

New N8/3-1163 Date 10-82 Eng BR

characteristics of each rock mass unit. It is generally assumed that the response characteristics (or their functions) are unique for each rock mass unit because the physical characteristics are relatively constant throughout that unit. However, each rock mass unit is only approximately homogeneous, so that there is some variability in physical characteristics throughout the unit and some resulting variability in response characteristics as measured from point to point within the unit.

Volumes which are relatively homogeneous with respect to hydrologic response characteristics are often termed "hydrologic units." Such hydrologic units consist of one or more rock mass units, and will not necessarily be homogeneous with respect to physical characteristics or other response characteristics. The characteristics of rock mass units will thus be discussed here.

Discontinuities, as separations in the intact rock matrix, tend to dominate the response of the rock mass. Hence, two different approaches have been developed to assess the response of a rock mass (i.e., consisting of intact rock, discontinuities, and pore fluid). The rock mass can be treated as either:

- A discontinuum, in which the discontinuities are represented discretely and explicitly
- A continuum, in which the discontinuities are implicitly represented as part of a quasi-continuum rock mass.

The explicit treatment of each discontinuity in a <u>discontinuum</u>-type analysis necessitates the following:

- Description of <u>each discontinuity</u>. This includes determining the location, orientation, and planarity, as well as the response characteristics of each discontinuity. These response characteristics will be related to the physical characteristics of the discontinuity, i.e., persistence, aperture, roughness, nature of infilling or surface materials.
- Description of the intact rock. This includes determining the response characteristics of the intact rock, which will be related to the physical characteristics of the intact rock, i.e., its mineralogy and texture (including microcracking, bedding, and schistosity).

However, it is logistically very difficult to assess and model each discontinuity explicitly on a large scale and perform numerical analyses.

The implicit treatment of all the discontinuities in a continuum-type analysis necessitates the description of the rock mass. This includes determining the response characteristics of the rock mass, as a composite of the intact rock and discontinuities. These response characteristics will thus be related to the physical characteristics of the rock mass (i.e., the spacing, orientation, and planarity of the discontinuities), as well as the response characteristics of both the intact rock and the discontinuities. It is relatively easy to perform continuum-type analyses.

3.2 VARIABLES WHICH AFFECT ASSESSMENT OF RESPONSE CHARACTERISTICS

3.2.1 Sensitivity

The response characteristics for any volume of material are often:

- Anisotropic (i.e., vary with orientation)
- Scale-dependent (i.e., vary with the scale of the sample)
- Time-dependent (i.e., vary over time)
- A function of the present and past environmental conditions, including
 - stress level (orij, in tensor notation)
 - pore pressure (u)
 - temperature (Te)
 - radiation dose (R)

Thus, the value of an anisotropic parameter (P_{kl} , in tensor notation) for a volume of material at any location can be generally expressed as:

 $(P_{k1}) = f(V, T, \sigma_{1j}, u, T_{p}, R)$

where

V = scale or volume of the sample T = time of applicability

The <u>sensitivity</u> of the value of each response characteristic to each of the <u>assessment</u> variables can be investigated. For example, if the multivariable function "f" were reliably known, the sensitivity could be assessed by taking the partial derivative of "f" with respect to any assessment variable, e.g. (see Figure 3.1):

$$\frac{\delta(P_{XI})}{\delta Te} = \frac{\delta f(V, T, \sigma ij, u, Te, R)}{\delta Te}$$

The multivariable function "f" is not typically known, however, so that the sensitivity is often assessed by observing the change in P_{k1} with a change in one assessment variable, while maintaining the others constant. The observed sensitivity may change significantly, however, for another set of assessment variable values.



Parameter, Pki

.

Eng. DL

Dete.

Rev. No 2/ 0 1

10-82

Figure 3.1

Sensitivity of parameter to one assessment variable is for a given set of the other assessment variables (i.e., constant) and a given set of physical characteristics of the material.

Temperature, Te

Value of anisotropic parameter, P_{kl} (in tensor notation), is a function of various assessment variables:

 $(P_{kl}) = f(v, T, \sigma_{ij}, u, Te, R)$

here	V= scale or volume of sample
	T= time of applicability
	ofj=stress level (in tensor notstion)

u= pore pressure Te* temperature

R. radiation dose -

Environmental conditions (past and present)

This multivariable function is unique for a given set of physical characteristics of the material.

The general sensitivity of each of the response characteristics to the assessment variables has been qualitatively assessed, based on experience (see Table 3.2). However, the relationship, and thus sensitivity, of response characteristics to the assessment variables have not been explicitly established and, also, are not independent of media in many cases. Hence, these sensitivities provide a qualitative indication only of which assessment variables can be expected to affect the characteristic values.

3.2.2 Representative Sample

It is assumed that, for a given set of physical characteristics, the multivariable function "f" is unique. Thus, in order to adequately assess this function, it is necessary to obtain and test a representative sample, i.e., the sample must have a set of physical characteristics which is very similar to that of the material. However, the material is relatively homogeneous only within a certain dimensional range (see Figure 3.2):

- Intact rock is relatively homogeneous between the scales of textural features (i.e., typically fractions of an inch) and discontinuity spacing (i.e., up to tens of feet). A representative sample of intact rock can often be obtained by coring. Thus, the response characteristics of intact rock are relatively easy to assess.
- Discontinuities are relatively homogeneous between the scales of roughness (i.e., typically inches to feet) and planarity (i.e., often many feet). Although the physical characteristics of discontinuities within a set vary widely, it is often assumed that discontinuities within a joint set have similar physical characteristics, so that one joint is considered to be representative of the set. However, it is difficult to assess large-scale physical characteristics such as planarity and persistence. A representative sample of a discontinuity is difficult to obtain, except exposed in underground openings. Thus, the response characteristics of discontinuities are relatively difficult to assess.
- The rock mass is relatively homogeneous between the scales of discontinuity spacing (i.e., typically inches up to tens of feet) and stratigraphic/structural features (i.e., often many tens of feet). At smaller scales, either the intact rock or the discontinuity is tested, and not the composite rock mass. Conversely, at very large scales, the rock mass is no longer homogeneous, as diverse rock mass units are included in the sample. At an intermediate scale, however, the volume is sufficient to contain a significant number of representative discontinuities. This "representative" volume of the rock mass is typically difficult to achieve and thus the quasi-continuum response characteristics of the rock mass are often relatively difficult to assess. They can be assessed in one of two ways, either:

SENSITIVITY OF RESPONSE CHARACTERISTICS TO ASSESSMENT VARIABLES

Table 3.2

	ADDEDDMENT VARIADLED										
				ENVIRONMENTAL CONDITIONS (Past and Present							
RESPONSE CHARACTERISTICS (see Table 3.1)	Anisotropic	Scale-Dependent	I ime-Dependent	Stress Level	Pore Pressure	Temperature	Radiation Dose				
MECHANICAL											
Strength							0				
Deformation							0				
Creep/fusing	0	0	•	•	0	•	0				
THERMAL											
Thermal conductivity					0						
Heat capacity		0		0	0						
Linear thermal expansion	•	•	0	•	0		0				
HYDROLOGIC											
Hydraulic conductivity											
Effective porosity	0			0							
Specific storage	0			0	•	•					
GEOCHEMICAL											
Dispersivity			0		0						
Adsorption/retardation			0	0	0						
Alteration/solubility				0			•				

NENT UNDIADIES

response characteristic is typically sensitive to variable

24 Eng

-1.82

2

Dele.

Boo No. 23-4635

o responce characteristic may be sensitive to variable, but probably at a low level (if at all)

٩.

Note: The sensitivity of response characteristics to the assessment variables have been qualitatively assessed, based on experience. However, these sensitivities are <u>qualitative indicators only</u>, as the relationship, and thus the sensitivity, of response characteristics to the assessment variables have not been explicitly established and, also, are not independent of media in many cases.



- a large-scale sample, which contains a significant number of representative discontinuities, can be tested and the quasi-continuum response characteristics directly assessed. However, it is logistically very difficult to obtain and test a sufficiently large sample (i.e., often with dimensions of many feet), except with large-scale in situ tests.
- the quasi-continuum response characteristics can be indirectly determined as a result of explicitly modeling (numerically) representative discontinuities within a simulated volume of rock mass. Although it is relatively difficult to assess the characteristics of the discontinuities, it is relatively easy to adequately assess the response characteristics of the intact rock and to assemble a model to develop quasi-continuum response characteristics for the rock mass. This indirect assessment may contain significant uncertainty in the derived response characteristics.

Each rock mass unit is approximately homogeneous, however, so that the physical characteristics of even appropriately sized samples will vary from sample to sample. The response characteristics measured for these samples will thus exhibit some natural variability within each rock mass unit.

Other problems in obtaining a representative sample include changes in the physical characteristics due to sampling and with time. For example, the aperture of discontinuities typically changes with sampling. This causes a significant change in the response characteristics (e.g., strength, hydraulic conductivity, etc.). As another example, the intact rock mineralogy may change as alteration occurs with time. Similarly, the pore fluid content/composition and the nature of discontinuity infilling/surface coating may change with time.

The physical characteristics of the disturbed rock mass which will exist immediately around any underground openings will be unknown until after construction and may be significantly different from the undisturbed rock mass. This is especially related to the overstressing which occurs during excavation, and which leads to additional microcracking of intact rock and creation of additional discontinuities. Also, this zone may be dewatered and exposed to conditions which will accelerate alteration. The differences in the physical characteristics, and thus the response characteristics, may be significant between the disturbed and undisturbed rock mass, and will be a function of the excavation procedure. Prior to construction, these differences can only be predicted, whereas after construction they can be observed.

3.3 UNCERTAINTY IN THE ASSESSMENT OF CHARACTERISTICS

The estimate of each response characteristic is based on the interpretation of all available information, i.e., the data base. In addition to the inherent uncertainty in the predicted value of each

characteristic due to this interpretation, there will be some unknown natural variability throughout each rock mass unit. The level of uncertainty in the assessment of each characteristic will be related to:

- Range in physically possible values. Absolute upper and lower bounds on the value of each characteristic can be set by physical laws, e.g., shear strength, thermal conductivity, and hydraulic conductivity have lower bounds of zero. The larger the range between the upper and lower bounds, the larger the potential uncertainty is in the assessment of a given characteristic.
- Natural variability. There will exist some natural variability throughout each rock mass unit. This variability causes a residual uncertainty in the estimate of each characteristic, even for an ideal data base. The larger this variability, the larger the residual (or minimum possible) uncertainty.
- Quality of data base. The data base must contain sufficient information to draw conclusions regarding the mean value and natural variability of each significant characteristic. Thus, the data base must be sufficiently large, and the correlation between each data point and the actual value must be high. This correlation will be a function of random errors and/or systematic errors (or biases) in the determination of each data point. These errors, in turn, are a function of:
 - Representation of the material by the sample. The physical characteristics of the sample tested may be different than the rock mass unit of interest, e.g., due to scale problems, sampling disturbance, or a change in pore fluid. Also, the physical characteristics may change with time, e.g., as alteration or solution occurs. In the case of the potential disturbed zone, the physical characteristics are a function of the excavation procedure. Where a numerical model is used to represent the coupling of individual components of a rock mass unit, there will be additional uncertainty in the derived data point.
 - Representation of the environmental conditions by the test. The test may not establish the complete relationship between the response characteristic and all of the environmental conditions. For example, one of the environmental conditions (e.g., temperature) might be held constant as the others are varied. If the characteristic is sensitive to that variable, then there will be significant uncertainty in the resulting assessment.
 - Accuracy in individual test results. The results of each test will have some inherent uncertainty due to possible measurement errors/ biases and assumptions/interpretations in the analysis. With care, this uncertainty can often be significantly reduced.
As the quality of the data base improves, e.g., incorporating appropriate data from carefully controlled in situ tests under operating conditions, the uncertainty in the assessment of a characteristic decreases.

Where the characteristic is <u>quantifiable</u> and thus the probability of values is continuously distributed, e.g., hydraulic conductivity, the uncertainty in the assessment can be expressed in terms of confidence levels, e.g., those bounding values for which there is a 90% chance that the actual value will lie between (see Figure 3.3). Higher uncertainty is reflected by larger ranges for a given confidence level (see Figure 3.4). Where the characteristic is <u>non-quantifiable</u> (or has integer values) and thus the probabilities are discretely distributed, e.g., the presence of faults, the uncertainty can be expressed in terms of probabilities for each possibility (or scenario) in a comprehensive, mutually exclusive set.

It must be emphasized that the probability distributions for each characteristic are assessed based on the interpretation of available information (i.e., the data base). This probability distribution is not equivalent to the statistical distribution of data contained within the data base, but should take into account <u>all</u> components of uncertainty.

3.4 PRESENT ASSESSMENT OF MEDIA/SITE SPECIFIC CHARACTERISTICS

3.4.1 Basis for Present Assessment

The values of significant characteristics have been roughly assessed, as discussed in previous sections, for each of the media/sites under consideration (see Appendix B of Volume II). This assessment has been based on readily available information. That is, a data base has been assembled for each media/site and subsequently interpreted. This present assessment of media/site specific characteristics consists of a best estimate and an indication of the present level of uncertainty in that value (e.g., 90% confidence levels). This uncertainty includes the natural variability of the characteristic.

Although this assessment is considered to be sufficient for the purposes of this study, i.e., illustrating the development of, and subsequent response to, information needs, it should not be construed as being sufficient for accurate performance assessment/licensing purposes. As additional information becomes available, this assessment will change.

3.4.2 Present Media/Site Specific Assessments

3.4.2.1 Basalt

In accordance with the previously described basis (see Section 3.4.1), the present assessment of characteristics of the site in basalt at the Hanford Reservation in Washington has been accomplished (see Appendix B





,

of Volume II, Table B-2). The distinctive features of the site include:

- Relatively complex geology (flow structures)
- Potential for tectonic activity
- High horizontal in situ stresses
- High in situ temperatures
- Proximity to a major water resource
- Highly fractured rock mass, especially vertical cooling joints; the intact rock is relatively strong, brittle, abrasive, impermeable, and thermally conductive, but the fractures dominate rock mass strength, stiffness, hydraulic conductivity, effective porosity and adsorption/retardation (similar to granite, see Section 3.4.2.5).

3.4.2.2 Tuff

In accordance with the previously described basis (see Section 3.4.1), the present assessment of characteristics of the site in tuff at Yucca Mountain at the Nevada Test Site has been accomplished (see Appendix B of Volume II, Table B-3). The distinctive features of the site include:

- Relatively complex geology (flow structures)
- Potential for tectonic activity
- Deep water table
- Very porous fractured rock mass; the rock mass is relatively weak and may have high hydraulic conductivity, although it may be highly adsorptive
- Susceptibility of rock to alteration, especially with elevated temperatures.

3.4.2.3 Domal Salt

In accordance with the previously described basis (see Section 3.4.1), the present assessment of characteristics of the potential sites in domal salt along the Gulf Coast has been accomplished (see Appendix B of Volume II, Table B-4). The distinctive features of these sites include:

Relatively complex geology (folding)

- Possibly ongoing diapirism (dome building)
- Relatively weak and plastic rock mass, which exhibits creep and self-healing/fusing (minimal fracturing); mechanical characteristics degrade rapidly with increasing temperature
- Relatively impermeable rock mass, but the rock is soluble and the pore fluid is corrosive (similar to bedded salt, see Section 3.4.2.4).

3.4.2.4 Bedded Salt

In accordance with the previously described basis (see Section 3.4.1), the present assessment of characteristics at any potential site in bedded salt has been accomplished (see Appendix B of Volume II, Table B-5). The distinctive features of these sites include:

- Possible existence of continuous, porous interbeds
- Relatively weak and plastic rock mass, which exhibits creep and self-healing/fusing (minimal fracturing); mechanical characteristics degrade rapidly with increasing temperature
- Relatively impermeable rock mass, but the rock is soluble and the pore fluid is corrosive (similar to domal salt, see Section 3.4.2.3).

3.4.2.5 Granite

In accordance with the previously described basis (see Section 3.4.1), the present assessment of characteristics at any potential site in granite has been accomplished (see Appendix B of Volume II, Table B-6). The distincitve features of these sites include:

 Rock mass with widely spaced joints; the intact rock is relatively strong, brittle, abrasive, impermeable, and thermally conductive, but the fractures dominate rock mass strength, stiffness, hydraulic conductivity, effective porosity and adsorption/retardation (similar to basalt, see Section 3.4.2.1).

3.5 SIGNIFICANCE OF EACH CHARACTERISTIC

3.5.1 Process of Evaluation

The uncertainty in the assessment of each characteristic must be reduced to certain levels in order to resolve the issues and demonstrate an acceptably high level of confidence in satisfactory performance. However, the required level of confidence in the prediction of each characteristic will not necessarily be media/site independent, but may be a function of its expected value and the values of other significant characteristics. This is because repository performance is a function of many characteristics, and acceptable repository performance can be predicted with the required level of confidence by various combinations of characteristic values.

In order to establish the approximate required level of confidence in the prediction of each characteristic, the significance of each characteristic, in terms of its influence on the short-term construction/operation and long-term waste containment/isolation performance, must first be established. This evaluation should take into account the following general attributes of the characteristic:

- Availability of cost-effective design and construction techniques which allow for a conservative assumption of the value of the characteristic
- Uncertainty in the representation of the phenomenological laws of nature by the performance prediction model
- Sensitivity of the performance prediction model to the value of the characteristic
- Cost effectiveness and scheduling limitations of measures to reduce the uncertainty in the assessment of the value of the characteristic.

The process used in this evaluation of the significance of each characteristic regarding compliance with each of the two summary performance criteria (short-term construction/operation and long-term waste containment/isolation) was to ask the following three questions in sequence and respond based on experience (see Figure 3.5):

(1) Is this characteristic significant to the particular performance criteria?

In other words, must the characteristic be assessed in order to design a critical engineering variable and/or demonstrate compliance with the performance criteria? If there is no significant relationship between the characteristic and the performance criteria, then the characteristic is determined to be of minor significance and should have the lowest priority for assessment and review. For example, such a lack of significance could be because either:

- (a) a conservative assumption for the value of the characteristic can easily be designed around and thus has little cost impact
- (b) the predictive model used in performance assessment and demonstration of compliance with performance criteria is not very representative of the real world and/or that model is relatively insensitive to the characteristic, so that a reduction in the uncertainty in the assessment of the characteristic does not result in a significantly improved performance assessment.



,

If there is a significant relationship between the characteristic and the performance criteria, further evaluation is required.

(2) Can anything significant be learned regarding this characteristic during the in situ test program?

In other words, can the uncertainty in the assessment of the characteristic be potentially reduced in a cost-effective manner subsequent to the initial SCR submittal and prior to license application (LA)? If nothing realistic can be done, then the characteristic is determined to be significant, but not for in situ testing. For example, this might be because either:

- (a) there is no test available to better assess the characteristic during the in situ test program
- (b) there is low uncertainty in its present assessment.

If the uncertainty in the assessment can be realistically reduced, further evaluation is required.

(3) Does the in situ test program, as it pertains to the assessment of this characteristic, result in a large reduction in the uncertainty in predicted performance, as required for a demonstration of compliance with the performance criteria?

In other words, does the predictive model used in performance assessment and demonstration of compliance with the performance criteria adequately represent the real world and also is that model sensitive to the characteristic? If the model is very representative (i.e., has low uncertainty) and is also highly sensitive to the characteristic, so that the in situ test program results in a major improvement in the performance assessment, then that characteristic is determined to be of major significance. If the model is only moderately representative and/or only moderately sensitive to the characteristic, so that the in situ test program results in only a moderately better performance assessment, then that characteristic is determined to be of major significance. If

Characteristics with major significance need to be assessed with relatively low uncertainty and should have highest priority for in situ testing. Characteristics with moderate significance can be assessed with greater uncertainty and should have next highest priority for in situ testing. Characteristics with minor significance can be assessed with relatively large uncertainty and should have low priority for in situ testing. Characteristics which are significant, but cannot be assessed during in situ testing must be assessed with relatively low uncertainty subsequent to repository construction.

The maximum acceptable level of uncertainty in the prediction of each characteristic which will sufficiently resolve the key issues (Table 3.1) has been subjectively assessed (see Table 3.3) based on experience

ACCEPTABLE LEVEL OF UNCERTAINTY IN THE ASSESSMENT OF EACH CHARACTERISTIC

Table 3.3

	ACCEPTABLE LEVEL OF UNCERT (See Figure 3.4)	AINTY
CHARACTERISTIC (see Table 3.1)	Very High High Moderate Low	Very Low
GEOLOGIC SETTING		
Stratigraphic/structural**	-	
Tectonic		
In situ stress field	-	
In situ hydraulic head field		
In situ temperature field		
RESPONSE		
Strength		
Deformation		
Creep/fusing	•	
(THERMAL)	+	
Thermal conductivity		
Heat capacity		
Linear thermal expansion	•	
(HYDROLOGIC)		-
Hydraulic conductivity		
Effective porosity		
Specific storage	•	
(GEOLHEMICAL)	T	
Dispersivity		
Adsorption/retardation		
Alteration/solubility		

Eng. B.A

out th

Dete.

A 917-1163

No

R.v.

*Note: Acceptable levels of uncertainty have been subjectively assessed, based on experience and a particular licensing perspective, especially regarding the acceptable level of confidence in satisfactory performance (see Preface). However, these levels are qualitative indicators only, as the sensitivity of performance to all system components has not been explicitly determined. Also, the significance of these characteristics is not independent of megnitude of the characteristic value, of each other or of media.

 Maximum acceptable level of uncertainty for undisturbed rock mass characteristic (and disturbed, unless otherwise indicated)

o Maximum acceptable level of uncertainty for disturbed (due to excavation) rock mass characteristic

**Includes the physical and chemical characteristics of each rock mass unit (including pore fluid composition).

and a particular licensing perspective, especially regarding the acceptable level of confidence in satisfactory performance (see Preface). However, until the sensitivity of performance to all system components (including site characteristics) has been determined and the licensing perspective clarified, these maximum acceptable levels of uncertainty in the prediction of each characteristic can be considered as qualitative indicators only. Also, the significance of characteristic value, of each other or of the media.

The significance of the various characteristics, specifically as they relate to short- and long-term performance, has been determined utilizing the above procedure (Figure 3.5) for each of the media/sites under consideration. This assessment of significance is thus based on the present assessment of media/site specific characteristics (including magnitude and uncertainty) (see Section 3.4 and Appendix B of Volume II), and also the perceived reliability of models, general test capabilities, and sensitivity of performance to each characteristic.

3.5.2 Media/Site Specific Evaluations

3.5.2.1 Basalt

The significance of the characteristics, as they relate to the shortand long-term performance criteria, has been evaluated for the site in basalt at the Hanford Reservation in Washington (see Table 3.4). This evaluation has been based on the previously discussed procedure (see Section 3.5.1), the present assessment of characteristics (see Section 3.4.2.1 and Appendix B of Volume II, Table B-2), and experience.

3.5.2.2 Tuff

The significance of the characteristics, as they relate to the shortand long-term performance criteria, has been evaluated for the site in tuff at Yucca Mountain at the Nevada Test Site (see Table 3.5). This evaluation has been based on the previously discussed procedure (see Section 3.5.1), the present assessment of characteristics (see Section 3.4.2.2 and Appendix B of Volume II, Table B-3), and on experience.

3.5.2.3 Domal Salt

The significance of the characterisitcs, as they relate to the shortand long-term performance criteria, has been evaluated for the potential sites in domal salt along the Gulf Coast (see Table 3.6). This evaluation has been based on the previously discussed procedure (see Section 3.5.1), the present assessment of characteristics (see Section 3.4.2.3 and Appendix B of Volume II, Table B-4), and on experience.



	PERFORMANC	E CRITERIA
CHARACTERISTICS (see Table 3.1)	SHORT-TERM (up to 100 years) COMSTRUCTION/OPERATI	LONG-TERM (100-10,000's Years WASTE CONTAINMENT) ON ISOLATION
GEOLOGIC SETTING Stratigraphic/structural*		
Tectonic		
In situ stress, hydraulic hea temperature fields RESPONSE (MECHANICAL) Strength	d and	
Deformation		
Creep		
(THERMAL) Thermal conductivity		
Heat capacity		
Linear thermal expansion		
(HYDROLOGIC) Hydraulic conductivity		
Effective porosity		
Specific storage		
(GEOCHEMICAL) Dispersivity		
Adsorption/retardation		
Alteration		
*Includes the physical and che (including pore fluid composi Characteristic with	mical characteristics tion) major significance	of each rock mass un
Characteristic with	moderate significance	
Characteristic with	minor significance	
Characteristic is s	ignificant, but not du	ring in situ testing

71.5

		PERFORMANC	E CRITEPIA
CHARACTERISTIC (See Table 3.1		SHORT-TERM (up to 100 years) DNSTRUCTION/OPERATI	LONG-TERM (100-10,000's Years WASTE CONTAINMENT/ ION ISOLATION
GEOLOGIC SETTING Stratigraphic/str	uctural *		
Tectonic			
In situ stress, h temperature fie RESPONSE (MECHANICAL)	nydraulic head an elds	d	
Strength			
Deformation			
(THERITAL) Thermal conductiv	vity		
Heat capacity			
Linear thermal e	xpansion		
(HYDROLOGIC) Hydraulic conduc	tivity		
Effective porosi	ty		
Specific storage			
(GEOCHEMICAL) Dispersivity			
Adsorption/retar	dation		
Solubility			
*Includes the physi (including pore fl	cal and chemical uid composition)	characteristics of	f each rock mass unit
Charact	eristic with maj	or significance	
Charact	eristic with mod	erate significance	
Charact	eristic with min	or significance	

3.5.2.4 Bedded Salt

The significance of the characteristics, as they relate to the shortand long-term performance criteria, has been evaluated for potential sites in bedded salt (see Table 3.7). This evaluation has been based on the previously discussed procedure (see Section 3.5.1), the present assessment of characteristics (see Section 3.4.2.4 and Appendix B of Volume II, Table B-5), and on experience.

3.5.2.5 Granite

The significance of the characteristics, as they relate to the shortand long-term performance criteria, has been evaluated for potential sites in granite (see Table 3.8). This evaluation has been based on the previously discussed procedure (see Section 3.5.1), the present assessment of characteristics (see Section 3.4.2.5 and Appendix B of Volume II, Table B-6), and on experience.

	PERFORMANCE	CRITERIA
CHARACTERISTICS (see Table 3.1)	SHORT-TERM (up to 100 years) CONSTRUCTION/OPERATIO	LONG-TERM (100-10,000's Years WASTE CONTAINMENT/ N ISOLATION
GEOLOGIC SETTING Stratigraphic/structural*		
Tectonic		
In situ stress, hydraulic he temperature fields RESPONSE (MECHANICAL)	ead and	
Strength		
Deformation		
Creep/fusing		
(THERMAL) Thermal conductivity		
Heat capacity		
Linear thermal expansion		
(HYDROLOGIC) Hydraulic conductivity		
Effective porosity		
Specific storage		
(GEOCHEMICAL) Dispersivity		
Adsorption/retardation		
Solubility		
*Includes the physical and ch (including pore fluid compos	nemical characteristics of sition)	each rock mass uni
Characteristic with	major significance	
Characteristic with	moderate significance	
Characteristic with	n minor significance	
Characteristic is s	significant, but not durin	a in citu toctina

28.0

SIGNIFICANCE OF CHARACTERISTICS FOR IN SITU TESTING AS THEY RELATE TO DESIGN Table 3.8 (GRANITE - UNSPECIFIED SITE)

	PERFORMANCE	CRITERIA
CHARACTERISTICS (See Table 3.1)	SHORT-TERM (up to 100 years) (CONSTRUCTION/OPERATIO	LONG-TERM 100 to 10,000's Year WASTE CONTAINMENT, N ISOLATION
GEOLOGIC SETTING Stratigraphic/structural*		
Tectonic		
In situ stress, hydraulic hea temperature fields RESPONSE (MECHANICAL)	d and	
Deformation		
Creen		
(THEDMAL)		
Thermal conductivity		
Heat capacity		
Linear thermal expansion		
(HYDROLOGIC) Hydraulic conductivity		
Effective porosity		
Specific storage		
(GEOCHEMICAL) Dispersivity		
Adsorption/retardation		
Alteration/solubility		
*Includes the physical and cher (including pore fluid composi	mical characteristics of tion)	each rock mass unit
Characteristic with	major significance	
Characteristic with	moderate significance	
Characteristic with	minor significance	
Characteristic is si	gnificant, but not durin	a in situ testing
te: Significance of characteristi process illustrated in Figure	cs was subjectively eval 3.5 and discussed in Se	uated using the ction 3.5.1.

Eng. DR

28.91 Dele /-42

Rev. 477- 463 C. Dwe No. 4/3 //630

AVAILABLE TEST METHODS

4.1 EXISTING REPOSITORY-RELATED IN SITU TEST PROGRAMS

4.1.1 Significance of Existing Programs

The DOE site screening/selection program is progressing rapidly (Table 1.1). In situ test plans have been or will be developed by DOE (or their contractors) for responding to the perceived information needs not only in a generic sense, but specifically for each of the sites under consideration. In addition, test facilities have been developed or are planned for various media to assess specific test techniques and verify certain aspects of predictive models; however, these media specific test facilities have generally been limited in scope and not intended for site characterization purposes.

The existing site specific in situ test programs and media specific test facilities (see Table 4.1) have been identified (see Appendix C of Volume II) although they are subject to update and revision. The evaluation of these existing programs is not within the scope of this study. However, these programs serve as examples of what has been considered by others to be appropriate for responding to perceived information needs. These existing programs also identify some available test methods.

4.1.2 Existing Media Independent Programs

A generic in situ test program has been developed by DOE (DOE/NWTS, 1981), and summarized in "NWTS Program Strategy and Guidelines for the Development of Test Facilities at Candidate Repository Sites" (January 8, 1982).* This program is subject to update and revision, but presently consists of the following facilities (see Table 4.2):

- Exploratory shaft
 - Phase I to determine the suitability of the site for a test and evaluation facility (TEF),
 - Phase II for site characterization
- At-depth test facility (ADTF) for site characterization
- Test and evaluation facility (TEF) for operational procedures.

The TEF, and its associated activities, are outside the scope of work, and thus have not been considered further in this study.

A variety of tests (see Table 4.3) is expected to be performed within these facilities in order to satisfy the stated objectives; the actual tests are subject to update and revision.

^{*}F.E. Coffman, DUE-Washington D.C., personal communication to J.B. Martin, NRC-Washington D.C., January 28, 1982.

PRESENT STATUS OF SITE SPECIFIC IN S'TU TEST PROGRAMS AND MEDIA SPECIFIC TEST FACILITIES

Table 4.1

MEDIA	SITE SPECIFIC IN SITU TEST PROGRAM PLAN	MEDIA SPECIFIC TEST FACILITIES
BASALT	Cold Creek Syncline at Hanford, Washington.	Near Surface Test Facilit at Hanford, Washington.
TUFF	Yucca Mountain at Nevada Test Site.	G-Tunnel at Nevada Test Site.
DOMAL SALT	None.	Conceptual Experimental Test Facility in South Texas.
		Avery Island Mine in Louisiana.
		Asse Mine in West Germany.
BEDDED SALT	None. (Although Waste Isolation Project Plant-WIPP in	Conceptual Experimental Test Facility in West Texas.
	Carlsbad, New Mexico is an example)	Carey Salt Mine in Hutchinson, Kansas.
		Project Salt Vault in Lyons, Kansas.
		WIPP in Carlsbad, New Mexico
GRANITE	None.	Stripa Mine in Sweden.
		Colorado School of Mines Experimental Mine in Colorado.
		Climax Stock at Nevada Test Site.
		UKAEA at Cornwall, England.
		Underground Research

Rev. ______ Dwg. No. A813-1163 Date /0-82 Eng. BR

63

Exploratory	shaft	At-Depth Test Facility	Test and Evaluation Facility				
Phase I	Phase II	(ADIF)	(TEF)				
 Access to horizon for Phase II Testing Allow decision on TEF site suitability 	 Allow decision on site suitability for repository Geotechnical design verification data for TEF (selected site only) Provide information necessary to finalize SCR submittal to NRC 	 Geotechnical design optimization and verification data for repository scaling technology thermal response thermomechanical response excavation methods ground control water control Expanded verification of Reference Repository conditions 	 Verification of waste handling, emplacement, and retrieval technology and procedures (not site-specific) Data base for occupational exposure Verification of ventilation system design Verification of instrumentation and control system design Evaluation of equipment performance Develop scaling factors to full repository Develop operational procedures for both routine and abnormal operations Operator training and certification Reduced uncertainty in design bases 				

Note: These objectives are subject to update and revision.

from "NWTS Program Strategy and Guidelines for Conduct Development of Test Facilities at Candidate Repository Sites" (January 8, 1982) by F.E. Coffman, DOE-Washington, D.C., personal communication to J.B. Martin, NRC-Washington, D.C., January 28, 1982.

Table 4.

N

SUMMARY OF PRESENTLY PROPOSED NWTS IN SITU TEST PROGRAM

Table 4.3

- EXPLORATORY SHAFT PHASE I Coreholes (limited)
 Exploratory excavation (limited)
 Determine rock mass structure
 Determine in situ stress field
 Determine rock mass strength
 Monitor drainage into excavation
 Construction monitoring
- EXPLORATORY SHAFT PHASE II
 Coreholes
 Exploratory excavation
 Exposure mapping
 Groundwater sampling
 Determine rock mass structure
 Determine in situ stress field
 Determine rock mass deformation moduli/Poisson's ratio
 Determine horizontal and vertical permeability of rock mass

 AT-DEPTH TEST FACILITY Exploratory excavation
 Exposure mapping
 Seismic monitoring
 Groundwater sampling
 Mine-by test
 Heater test
 Determine geochemical interactions
 Monitor drainage into excavation
 Monitor construction and performance of excavations which simulate repository (including associated rock disturbance/ damage)
 Borehole sealing test
 Shaft sinking/sealing test
 Backfill test

Note: This summary represents Golder Associates' perception of the current NWTS program, which is subject to update and revision.

The test and evaluation facility (TEF) is outside the scope of work, and has not been considered.

from "NWTS Program Strategy and Guidelines for the Development of Test Facilities at Candidate Repository Sites" (January 8, 1982) by F.E. Coffman, DOE-Washington, D.C., personal communication to J.B. Martin, NRC-Washington, D.C., January 28, 1982.

4.2 TEST METHODS AVAILABLE TO ASSESS SIGNIFICANT CHARACTERISTICS

There are typically a variety of test methods available for assessing the previously identified significant characteristics (see Chapter 3). These methods can be generally categorized as:

- Surface tests
- Borehole tests
- Laboratory tests
- In situ tests.

-

The test methods within each category which are generally available to assess each significant characteristic have been identified (see Table 4.4), based on experience as well as existing repository-related in situ test programs (see Section 4.1 and Appendix C of Volume II). This is a relatively comprehensive list of available tests; it should not be construed that all of these tests are being recommended. The selection of appropriate tests from this list will be a function of the perceived information needs and test capabilities.

It must be recognized that there are no surface or borehole tests available to assess the characteristics of the disturbed zone surrounding an underground opening, except those which are used for the undisturbed rock mass and whose results must be modified to estimate the disturbed zone's characteristics. Also, any laboratory or in situ tests used to assess the response characteristics of the disturbed zone should be performed on samples with physical characteristics similar to those which will exist in the repository. Hence, laboratory samples should be obtained from, and in situ tests performed in, underground openings which are excavated and supported using similar techniques as are contemplated for the repository.

4.3 COMPARISON OF CHARACTERISTIC ASSESSMENT BY SPECIFIC TEST METHODS

The various test methods available for assessing each significant characteristic (Table 4.4) do so with various limitations and levels of uncertainty. As previously discussed (see Section 3.3), the uncertainty in the assessment of each characteristic is due to (1) range in physically possible values, (2) natural variability and (3) quality of data base, which in turn is a function of the size of the data base and random/systematic errors in each datum; these errors are a function of the environmental conditions by the test, and (c) accuracy in individual test results.

The level of uncertainty in the determination of each characteristic, due to quality of the data base, roughly varies with categories of test methods, generally decreasing as follows:

and the second second				
HARACTERISTIC see Table 3.1)	SURFACE	AVAILABLE TEST METHOD (see BOREHOLE (from surface)	e notes() at end of Table ABORATORY IN SI	e) TU (from subsurface excavation)
GEOLOGIC SETTING)			
tratioraphic/ structural (includes the physical and cnemical char- acteristics of each rock mass unit, including bore fluid composition)	o geologic mapping (1) o surface geophysics (2).	o cutting evaluation of o coreholes (3) of o geophysic/seismic (x-hole, surface- borehole) ogroundwater sampling (5a) of -gas detector.	<pre>core logging determination of rock sample mineralogy (thin section, x-ray diffraction) determination of groundwater sample composition and age hydrochemical analysis of rock sample.</pre>	<pre>o exploratory excavations (including excosure mapping) o coreholes (3, 4) o rock mass sampling (5b) o geophysic/seismic (x-hole, exposure-borehole) o exposure geophysics (radar scanning) o groundwater sampling (4,1a) - gas detector (4) - gas detector in mine.</pre>
fectonic	<pre>o geologic mapping (1) o surface geophysics (2) o seismic monitoring - monitor surface disolacements/ tilt - monitor elec- trical poten- tial/current/ discharge.</pre>	 acoustic emission monitoring monitor stress changes in rock mass monitor dis- placements in rock mass monitor temperatures in rock mass monitor pore pressures in rock mass monitor gas emission in rock mass monitor elec- trical notential/ current/ discharge in rock mass. 		 acoustic emission monitoring seismic monitoring monitor stress changes in rock mass (4) monitor stress changes in supports monitor displacements in rock mass (4) monitor temperatures in rock mass (4) monitor temperatures in rock mass (4) monitor pore pressures in rock mass (4) monitor pore pressures in rock mass (4) monitor drainage into excavation monitor gas emission in rock mass (4) monitor gas emission in excavation monitor gas emission in excavation monitor electrical potential current/discharge.
In situ stress field (stress tensor) (6)		o rock mass sampling (5b) o hydrofracturing (16) o borehole jacking o overcoring.	<pre>o core observation (stress relief, core discing).</pre>	 o rock mass sampling (4, 5b) o hydrofracturing (4) o borehole jacking (4) o dvercoring (4) o flat jack test o overcoring (exposure) (7) monitor stress changes in rock mass (4, 7) monitor displacements in rock mass (4, 7) monitor displacements at exposure (7) monitor stress changes and strains in supports
In situ hydrauli head field fore pressure (5)	c o hydrologic mapping - climatic monitoring.	o monitor pore pressures - monitor bore- hole inflow - groundwater sampling (5a),	 determination of groundwater sample composition and age. 	 o monitor pore pressures in rock mass (4) monitor borehole drainage (4) monitor drainage into excavation (7) groundwater sampling (5a).
In situ tempera- ture field 'temperature'	(6)	o monitor temperatures.		o monitor temperatures in rock mass (4) - monitor temperature in excavation (7).
(MECHANICAL) Strength (8)		o rock mass sampling (5b) - borehole jacking (10a) - hydrofracturing.	<pre>0 index tests on rock samples (9) 0 simple strength tests on rock sample (sliding test, Brazilian test, point load test, beam test, tensile strength test, fracture toughness test) (10a) 0 direct shear test on discontinuity sample (w/or w/o neat) (10b, 14) 0 unconfined compression test on rock core (w/or w/o heat) (10a) 0 triaxial test on</pre>	<pre>o index tests on excosures (?) o rock mass sampling (5b) o shear jacking (10b, 11) o pillar test (w/or w/o heat) (1 o mine-by test o monitoring fracturing in rock mass around excavation (e.d., acoustic emission monitoring) (7) - borehole jacking (4, 10a) - hydrofracturing (4) - plate test (13)</pre>

TEST METHODS AVAILABLE FOR ASSESSING SIGNIFICANT CHARACTERISTICS

Table 4.4

2 of 4

	CHARACTERISTIC (see Table 3.1)	AVAILABLE TEST METHOD SURFACE BOREHOLE (from surface)	(see notes() at end of Tabl LABORATORY IN SI	e) TU (from subsurface excavation)
	Deformation (8)	o surface geophysics (2) o rock mass sampling (5b) o borehole jacking (10a) o geophysic/seismic (x-hole, surface- borehole).) o index tests on rock sample (9) o unconfined com- pression test on rock core (w/or w/o neat) (10a) o triaxial test on rock core (w/or w/o heat) (10a,b) o true triaxial test on large rock sample (w/or w/o heat) (10a,b) o direct shear test on discontinuity sample (w/or w/o heat) (10b, 14) o sonic velocity test on rock core (w/or w/o stress) (10a).	<pre>o index tests on exposures (4) o rock mass sampling (5b) o borehole jacking (4, 10a) o geophysic/seismic (x-hole, exposure-borehole) o exposure geophysics o flat jack test o plate test (13) o pillar test (w/or w/o heat) (12 o block test (w/or w/o heat) (12 o block test (w/or w/o heat) o chamber test w/displacement monitoring (w/or w/o heat) o shear jacking (10b,11) o mine-by test o monitor stress changes in rock mass as excavation occurs (4,7) o monitor displacements in rock mass as excavation occurs (4,7) o monitor displacements at exposure as excavation occurs(7 o monitor stress changes and strains in supports (7).</pre>
	Creep (B)	<pre>o rock mass sampling (5b) - borehole jacking, long term (10a) - monitor deformation of hole after boring w/soft inclusion (7,10a) - monitor stress changes around hole after boring w/hard inclusion (7,10a).</pre>	 o unconfined compression test on rock core, long term (w/or w/o heat) (lOa) o triaxial test on rock core, long term (w/or w/o heat) (lOa) o true triaxial test on larce rock sample, long term (w/or w/o heat) (lOa,b) o direct shear test on discontinuity sample, long term (w/or w/o heat) (lOb,l4) core observation (monitor deformations as stress relieved by coring) (7, lOa). 	<pre>o rock mass sampling (5b) o flat jack test, long term o plate test, long term (13) o pillar test, long term (13) o pillar test, long term (w/or w/o neat) (12) o block test, long term (w/or w/o neat) o chamber test w/displacement monitoring, long term (w/or w/o neat) o mine-by test o monitor stress changes in rock mass after excavation (4,7) o monitor displacements in rock mass after excavation (4,7) o monitor stress changes and strains in supports (7) o borehole jacking, long term (4,10a) monitor deformation of hole after boring w/soft inclusion (4,7, 10a).</pre>
10-82 ENO 28	(THERMAL) Thermal :onductivity (8) Heat capacity (8)	o rock mass sampling (5b) o thermal probe (single borehole small scale heater test) (10a) o heater test (x-hole small scale).	<pre>o heated rock sample (10a) o unconfined com- pression test on rock core w/heat (10a) o triaxial test on rock core w/heat (10a).</pre>	<pre>0 rock mass sampling (5b) 0 thermal probe (single borehole small scale heater test) (4,10a) 0 heater test (x-hole small scale) (4) 0 heater test (large scale) 0 block test w/heat (w/or w/o stress) 0 chamber test w/heat (12) 0 chamber test w/heat (12) 0 chamber test w/heat water 0 monitor temperature in rock mass and excavation (w/ventilation/ cooling) (7).</pre>
*8 No. 22/3//43 Date	Linear thermal expansion(3)	o rock mass sampling (5b),	<pre>o heated rock sample (10a) o unconfined com- pression test on rock core w/heat (10a) o triaxial test on rock core w/heat (10a).</pre>	<pre>o rock mass sampling (5b) o heater test (large scale) o block test w/heat (w/or w/o stress) o pillar test w/heated vater and w/displacement monitoring o monitor temperatures and displacements in rock mass and excavation (7).</pre>
		67		

TEST METHODS AVAILABLE FOR ASSESSING SIGNIFICANT CHARACTERISTICS

35 Eng.

28.01

Pe,

Table 4.4

No.



TEST METHODS AVAILABLE FOR ASSESSING SIGNIFICANT CHARACTER/STICS

Table 4.4 4 of 4

Notes: o Primary test method - Possible secondary test method

- <u>Geologic mapping includes a geodetic survey and exposure mapping, as well as possibly aerophotography (by airplane or satellite, black-and-white, color, or infrared).</u>
- (2) <u>Surface geophysics</u> include possibly gravity and magnetic surveys (land or air based), electrical survey, and seismic surveys (reflection and refraction).
- (3) <u>Coreholes</u> include coring and core logging, as well as possibly borehole surveying, caliper logging, oriented coring, integral sampling, impression packer, borehole TV/camera, borehole radar, and geophysical well logging (electrical, acoustic, and nuclear).
- (4) Tests are conducted in boreholes drilled from underground.
- (5) a) Groundwater sampling implies subsequent laboratory determination of groundwater composition and age.
 b) Rock mass sampling, either coring or large block samples, implies subsequent laboratory tests.
- (6) The in situ (i.e., pre-excavation or virgin) stress, hydraulic, and temperature fields can be indirectly assessed or inferred from the stratigraphy/structure and tectonics (e.g., in situ stress field can be inferred from the geomorphology and tectonics of the site).
- (7) Monitoring performance implies associated analysis to assess characteristic.
- (8) The response characteristics refer to the rock mass, which consists of intact rock, discontinuities and pore fluid. These
 - The response characteristics refer to the rock mass, which consists of infact rock, discontinuities and pore fluid. Inese response characteristics can be assessed either: O Directly by testing a large scale sample which contains a significant number of discontinuities o Indirectly by separately assessing the response characteristics of the intact rock, discontinuitis, and pore fluid, and then assembling by a model. Hence, the rock mass response characteristics can often be inferred from the stratigraphy/structure
- Index tests do not assess the characteristic directly, but by empirical correlations (e.g., use of a Schmidt hammer on rock core or on an exposure is an index test whose results can be roughly correlated with the modulus of deformation, based on experience). There are too many index tests, with varying reliability, to list. (9)
- (10) Tests assess the response characteristics of only the a) intact rock or b) discontinuity, and does not directly assess the response characteristics of the rock mass (see note 8)
- Shear jacking is very similar to torsion jacking. Because of these similarities, only shear jacking will be discussed, although torsion jacking might be a suitable alternative. (11) Shear

BE

Eng

28.01

Dete

4313-1163

No -----

- (12) <u>Pillar test</u> can consist of either: o Jacking an isolated pillar or unconfined block to failure (i.e., essentially an unconfined plate test or an axially loaded unconfined block test) Reducing the dimensions of a pillar (and thus increasing stresses) until failure occurs.
- (13) <u>Plate test</u> is very similar to two other tests: <u>o</u> <u>Cable</u> jacking test, in which the reaction is provided by an anchor in the rock mass rather than the opposite wall of the excavation
 - Radial jacking, in which the entire circumference of the opening is jacked using, for example, several plate jack 0 systems

Because of these similarities, only the plate test will be discussed, although cable jacking or radial jacking might be suitable alternatives.

- shear test on discontinuity samples can be of various scales and is also very similar to torsional shear tests. of these similarities, only the direct shear test will be discussed, although torsional shear test might be a (14) Direct shear Because suitable alternative.
- (15) Tests are constant head injection, constant head withdrawal, constant flow rate withdrawal, pulse injection, or gas injection permeability test. Multiple borehole permeability tests are often called "pump" tests.
- (16) Hydrofracturing must be very carefully performed in order to control the extent of fractures which are generated.

This is a relatively comprehensive list of test methods which are available for assessing site characteristics. It should not be construed that all of these available tests are being recommended.

- Estimate based on typical values for similar materials and conditions
- (2) Interpretation of the results of surface tests
- (3) Interpretation of the results of borehole or laboratory tests
- (4) Interpretation of the results of in situ tests
- (5) Interpretation of the monitored response of prototypes or prototype simulations.

The differences in uncertainty in the assessment of characteristics between the various test methods are due primarily to the quality of the data generated:

- <u>Surface tests</u>, although testing material with minimal sampling disturbance and under existing environmental conditions, generally have the following limitations:
 - characteristics assessed at the surface must be extrapolated to depth with significant uncertainty
 - characteristics of a volume of rock mass, which is often larger than the representative volume (i.e., nonhomogeneous), are often interpreted with poor resolution and no range in environmental conditions.
- Borehole tests, although testing the material in place, generally have the following limitations:
 - small-scale sample, so that either the intact rock or individual discontinuity, and not a representative volume of the rock mass, is tested
 - some disturbance of the sample tested occurs due to drilling (i.e., a change in physical characteristics)
 - only a very limited range of environmental conditions can be evaluated
 - poor control of test due to remoteness.
- <u>Laboratory tests</u>, although often allowing a full range in environmental conditions to be applied with good test control, generally have the following limitations:
 - small-scale sample, so that either the intact rock or possibly an individual discontinuity, and not a representative volume of rock mass, is tested

- significant disturbance of the sample occurs due to drilling and change in environmental conditions, such as stress relief or dessication.
- In situ tests, although testing a representative volume of material in place, generally have the following limitations:
 - some minimal disturbance of the sample due to excavation and instrumentation installation
 - limited range of environmental conditions can be practically evaluated.

The assessment variables (see Section 3.2.1) which can be incorporated in each of the available test methods have been identified (see Table 4.5), based on standard test methodologies. Clearly, in order to adequately assess a characteristic, a test method must be able to incorporate those assessment variables which that characteristic is sensitive to (Table 3.2).

The potential minimum level of uncertainty in the assessment of each of the significant characteristics by the various available test methods has thus been subjectively evaluated (see Table 4.5), based solely on the resulting quality of the data:

- (1) General sample representativeness
- (2) Applied range in environmental conditions
- Typical test reliability.

Clearly, however, the sample representation and test reliability are not necessarily media/site independent. Also, the level of uncertainty is not independent of the absolute magnitude of the value measured, i.e., a test may be appropriate for a given range of values only and very inaccurate outside of that range. Other factors, i.e., the range in possible values, natural variability, and size of the data base, also impact the level of uncertainty. The subjective evaluation of the potential minimum level uncertainty in the assessment of each characteristic by various available test methods (Table 4.5) thus provides a qualitative indication only of individual test capabilities.

4.4 IN SITU TEST METHODS AVAILABLE WHICH SIMULATE THE REPOSITORY

As previously discussed, rather than resolving key issues by adequately assessing certain characteristics, in situ tests can be used to resolve key issues by appropriately simulating various aspects of the repository. The in situ test facility is itself a simulation of various construction and operation aspects of the repository. The uncertainty in the extrapolation of results from a test case (i.e., a situ test or test facility) to the prototype (i.e., repository) will be a function of the degree of similarity between the two. This correlation is improved, and the uncertainty thus reduced, by making the test case as similar as

LIMITATIONS AND UNCERTAINTY IN ASSESSMENT OF SIGNIFICANT CHARACTERISTICS BY AVAILABLE TEST METHODS

Table 4.5

1 of 9

		L	TES	VAR	SSE	SSM ES	ENT	_		UNCERTAINTY IN					
					ENV	IRON DND1	E IN MEN TION	TAL		AS CHA	SES	SME	 15T	OF IC	
	TEST METHOD (See Notes () at end of Table)	Anisotropy	Scale (Rep. Volume)	Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	Low	Very LOW	
	Hydrologic mapping	-	•	-	-	X	-	-	In situ hydraulic field	F	4				
(SURFACE TEST METHODS)	Geologic mapping (1)		•	-					Stratigraphic/structural Tectonic	F	-				
	Surface geophysics (2)	-	• • •	- • ×	×	×	×		Stratigraphic/structural Tectonic Deformation	II					
(S	Seismic monitoring	-		x	-	-	-	-	Tectonic	-					
	Cutting evaluation	-	x	-	-	-	-	-	Stratigraphic/structural	-	4				
	Coreholes (3)	-	•	-	-	-	-	-	Stratigraphic/structural	-		-			
(BOREHOLE TEST METHODS)	Rock mass sampling (5b) (see laboratory tests)	1001001001001	• × × × × × • 1 • 1 × × • •	0111111100011	11111001000001	1100111100110	110000000000000	111111111000	(Pore fluid composition) In situ stress field Strength Deformation Creep/fusing Thermal conductivity Heat capacity Linear thermal expansion Hydraulic conductivity Effective porosity Specific storage Dispersivity Adsorption/retardation Alteration/solubility						
	Groundwater sampling (5a) (see laboratory tests)	-××-	• × × •		-		1000	1000	(Pore fluid composition) Dispersivity Adsorption/retardation Alteration/solubility						

72

B...

No. 4813-1163

Eng BR

Date 10-92

		Γ	TES	T A	SSE	SSM	ENT	T		T				
					ENV	RANG 1ROM	E IN	TAL		UN AS CHA	CER SES RAC	TAIN	NTY NT IST	101010
(54	TEST METHOD ee Notes () at end of Table)	Anisotropy	Scale (Rep. Volume)	Duration	Stress level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	LOW	Voeu Lou
	Geophysical well logging	×	×××		×	×××	×××		Hydraulic conductivity Effective porosity Specific storage	TTT				
	Geophysic/seismic (x-hole, surface-borehole)	-	•	×	×	- x	- ×	-	Stratigraphic/structural Deformation		H			
	Borehole jacking (10a)	X X	××	- 0	•	- ×	- ×	-	In situ stress field Deformation					
1	Overcoring	0	X	-	•	-	-	-	In situ stress field	-	-	-		
	, Hydrofracturing (16)	X	0	-	•	-	-	-	In situ stress field	-	4			
(00H)	Permeability test (single borehole) (15)	× - -	000		×	••••	××××		Hydraulic conductivity Effective porosity Specific storage	E	111			
OLE TEST ME	Multiple borehole permeability test (15)	•	000		×	•	×××		Hydraulic conductivity Effective porosity Specific storage			111		
(BOREH	Tracer test	:	000	-	-	-	××	:	Dispersivity Adsorption/retardation	F				
	Thermal probe (single borehole small scale heater test) (10a)	× -	•	-	× -	-	•	-	Thermal conductivity Heat capacity	TT				
	Heater Test (x-hole small scale)	•	0	-	×	-	•	-	Thermal conductivity Heat capacity					
	Acoustic emission monitoring	-	•	X	-	-	-	-	Tectonic	H				T
	Monitor pore pressure	-		-	-		-	-	In situ hydraulic field	-	-	4		Γ
	Monitor temperature	-	•	-	-	-	•	-	In situ temperature field	-	-	-		
	Monitor borehole inflow (7)	X - -	0 0 0		× - -	××××	× × ×		Hydraulic conductivity Effective porosity Specific storage					
	Monitor stability of borehole	-			-	X	X	X	Alteration/solubility	L	4			Γ

ė

73

-

LIMITATIONS AND UNCERTAINTY IN ASSESSMENT OF SIGNIFICANT CHARACTERISTICS BY AVAILABLE TEST METHODS

Table 4.5

			TES	T A	SSE	SSM	ENT			T				
					ENV	RANG	E IN	TAL		UN AS CHA	SES	SME TEP	KTY KT (IST)	IN OF IC**
(5)	TEST METHOD ee Notes () at end of Table)	Anisotropy	Scale (Rep. Volume)	Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	Low	Very Low
	Core observation (stress relief, core discing)	×	x	-	0	-	-	-	In situ stress field	F	-			
(LABORATORY TEST METHODS)	Core logging	-	• •	- 0	-	×	×	- ×	Stratigraphic/structural Alteration/solubility	E	-	-		
	Determination of rock sample mineralogy (thin section, x-ray diffraction)	- × × -	0 X X 0	1 1 1 0		X	- × × ×	I ×××	Stratigraphic/structural Dispersivity Adsorption/retardation Alteration/solubility	LIII	-			
	Hydrochemical analysis of rock sample	-	•	-	-	-	-	-	(Pore fluid composition)	F				+
	Determination of ground- water sample composition and age	- × × -	• × × •	×		- - ×	- × × ×	- × × ×	(Pore fluid composition) Dispersivity Adsorption/retardation Alteration/solubility				-1	-
	Index tests on rock sample (9)	0 0 0 1 1	× × × × ×	× ×	×××	× × × × × ×	× × × × ×		Strength Deformation Hydraulic conductivity Effective porosity Specific storage	IIIII				
	Sonic velocity on rock core (with or without stress) (10a)	°	X	X	0	X	X	-	Deformation	F				
	Simple strength tests on rock sample (10a)	0	X	X	X	X	X	-	Strength	F		T		T
	Unconfined compression test on rock core (with or without heat) (10a)	0 0	××	0 0	Xo	××	00	-	Strength Deformation	-	T			
	long term (with or without heat) (10a)	-	-	•	0	-	0	-	Creep/fusing	E	+			
	with heat (10a)	0 -	0	-	0 1	-	•		Thermal conductivity Heat capacity	E		+		
		0	0	-	0	-		-	Linear thermal expansion	F	-	-		

**Note: This is only a qualitative indication of uncertainty in the assessment of each characteristic, due only to quality of the data base which will be media/site specific and a function of the magnitude of the value measured (see Figure 3.4).

No. 4122-162 Date 10.12 Eng.

.....

S.C.

	and the second								Lange Carlo						
TEST AS VARI					SSE	SSM	ENT								
						RANG IRON DND I	E IN	AL S		UNCERTAINTY I- ASSESSMENT OF CHARACTERISTIC					
15	TEST METHOD	Anisotropy	Scale (Rep. Volume)	Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	Low	Very LOW	
100	Triaxial test on rock core (with or without	•	Х	0	•	0	0	-	Strength	F	-	+		-	
	heat) (10a, b)	•	X	0	•	0	0	-	Deformation		=	=	-	-	
	without heat (10a)		-	-	-	-	•	-	Creep/fusing			Ľ	_	L	
	with heat (10a)	•	0 1 0		• - •		• • •		Thermal conductivity Heat capacity Linear thermal expansion	E					
	True triaxial test on large rock sample (with or without heat) (10a,b)	•	X	0	•	0	0	-	Deformation	F	F	-		F	
	long term (with or without heat) (10a)	-	-	•	•	-	0	-	Creep/fusing	F	+	F	Γ	-	
METHOD)	Direct shear test on discontinuity sample (with or without heat) (10b, 14)	•	××	0 0	•	0 0	0 0		Strength Deformation	F	-				
TEST	long term (with or without heat) (10b, 14)	-	-	•	•	-	0	-	Creep/fusing	F	T			1	
LABORATORY	Heated rock sample (10a)	•	0 1 0		× - ×		•		Thermal conductivity Heat capacity Linear thermal expansion	E					
)	Axial permeability test on rock core (with or without heat, stress) (10a, 15)	0	××××		•	•	0 0 0		Hydraulic conductivity Effective porosity Specific storage	Ē	TTT				
	Radial permeability test on rock sample (with or without heat, stress) (10a, b, 15)	•	×××		0 1 1	•	0 0 0	1 1 1	Hydraulic conductivity Effective porosity Specific storage		111				
	Tracer test on rock sample (with or without heat, stress) (10a, b)	00	××	-	-	-	000	•	Dispersivity Adsorption/retardation	F		4			
	Slaking or accelerated weath- ering test on rock sample (with heat, stress)	-	•	•	-	x	•	X	Alteration/solubility	-			-		
	Solubility test on rock sample (with or without heated water)	-	•	•	-	X	•	X	Alteration/solubility	-	F	F	-	T	

		TEST ASSESSMENT VARIABLES					ENT	٦		T				
				ENV	ANGI LRON ND11	E IN MENT TION	TAL S		UNCERTAINTY I ASSESSMENT OF CHARACTERISTIC					
(54	TEST METHOD ee Notes () at end of Table)	Anisotropy	Scale (Rep. Volume)	Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	Low	Very Low
T	Monitor alteration/ solutioning of rock sample	-	•	•	-	X	Х	Х	Alteration/solubility	F	_	-		
+	Exploratory excavations	-		-	-	-	-	-	Stratigraphic/structural	F			-	
(IN SITU TEST METHODS)	Exposure mapping	-	•	1 0		- ×	- ×	×	Stratigraphic/structural Alteration/solubility	E		T	T	
	Index tests on exposure (9)	0 0	0 0	X X	X X	××	××	1 1	Strength Deformation	F	TT			
	Coreholes (3, 4)	-	•	-	-	-	-	-	Stratigraphic/structural			-	-	
	Rock mass sampling (5b) (see laboratory tests)	110010101011001	0 0 0 ×× 1 0 1 0 × ×× ×× 0	11100011111110	110000010011111	11100111100111	1110000000000000	11111111111000	Stratigraphic/structural (Pore fluid composition) In situ stress field Strength Deformation Creep/fusing Thermal conductivity Heat capacity Linear thermal expansion Hydraulic conductivity Effective porosity Specific storage Dispersivity Adsorption/retardation Alteration/solubility					
	Groundwater sampling (4, 5a) (see laboratory tests)	-××-	• × × •		-	×	000	1000	(Pore fluid composition) Dispersivity Adsorption/retardation Alteration/solubility					
	Geophysical well logging (4)	×	XXX		X	XXX	XXXX		Hydraulic conductivity Effective porosity Specific storage	E				

g

			TEST ASSESSMENT VARIABLES							Т					
					RANGE IN ENVIRONMENTAL CONDITIONS					UNCEPTAINTY 14 ASSESSMENT OF CHARACTERISTIC					
(Se	TEST METHOD	Anisotropy	Scale (Rep. Volume)	Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	Low	Very Low	
	Geophysic/seismic (x-hole, exposure- borehole)	-	•	- ×	×	- x	×	-	Stratigraphic/structural Deformation	F	-				
	Exposure geophysics (radar screening)	- 0	•	×	×	×	×	-	Stratigraphic/structural Deformation	E					
	Borehole jacking (4, 10a)	××	××	1 0	•	×	×	-	In situ stress field Deformation	E	TT				
	Overcoring (4)	0	Х	-	•	-	-	-	In situ stress field	-	-		-		
	Hydrofracturing (4)	X	0	-	•	-	-	-	In situ stress field	-	4				
	Permeability test (single borehole with or without heated water) (4, 15)	×	0 0 0	-	×	•	0 0 0		Hydraulic conductivity Effective porosity Specific storage	E	TTT				
T METHODS)	Multiple borehole permeability test (with or without heated water) (4, 15)	•	0 0 0		×	• •	0 0 0	-	Hydraulic conductivity Effective porosity Specific storage	E		III			
SITU TES	Tracer test (with or without heated water) (4)	:	0 0	-	-	-	0 0	•••	Dispersivity Adsorption/retardation	F					
(IN	Thermal probe (single borehole small scale heater test) (4, 10a)	×	0	-	× -	-	•	-	Thermal conductivity Heat Capacity	E					
	Heater test (x-hole small scale) (4)	1:	•	=	X	=	:	-	Thermal conductivity Heat capacity	E	E				
	Heater test (large scale)	•	•		× - ×		•••••	111	Thermal conductivity Heat capacity Linear thermal expansion	F					
	Övercoring (exposure) (7)	0	•	-		X	-	X	Alteration/solubility In situ stress field	-=	E	-	-	-	
	Flatjack test	•	0 0	- 0	•	- ×	x	-	In situ stress field Deformation	E		-			
	long term	1-	-	•	•	-	X	-	Creep/fusing		F		- 1	-	
	Plate test (13)		0	0	•	X	X	-	Deformation	-	-	-			
	long term	-	-	•	•	-	X	-	Creep/fusing	F	E	+		_	
	Shear jacking (10b, 11)		0	0	•	X	X	-	Strength	F	-	+			

Ber

		Г	TES	T A	SSE	SSM ES	ENT							
					ENV	RANGE IN ENVIRONMENTAL CONDITIONS				UNCEPTAINTY IN ASSESSMENT OF CHARACTERISTIC				
5	TEST METHOD ee Notes () at end of Table)	Anisotropy	Anisotropy Scale (Rep. Volume) Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High High Moderate Low Very Low					
T	Block test (with or without heat)	•	0	0	•	X	0	-	Deformation					
I	long term (with or without heat)	-	-	•	•	-	0	-	Creep/fusing					
-	with heat (with or without stress)	•	0 1 0		0 1 0		•		Thermal conductivity Heat capacity Linear thermal expansion					
-	with multiple bore- hole permeability test (with or without heat, stress)	-	0 0 0		0 1 1	•	0 0 0		Hydraulic conductivity Effective porosity Specific storage					
-	with tracer test (with or without heat, stress)	•	0 0	-		-	000	•••	Dispersivity Adsorption/retardation					
1	Pillar test (with or without heat) (12)	0	0 0	0 0	•	XX	0 0	-	Strength Deformation					
	long term (with or without heat (12)	-	-	•	•	-	0	=	Creep/fusing					
	with heat (12)	0 - 0	0 1 0	-	• - •	-		-	Thermal conductivity Heat capacity Linear thermal expansion	E				
C31 0110 H	Mine-by test	•	•	•	×××	××-	×××	-	Strength Deformation Creep /fusing					
-	Chamber test (with or without heated water) (15)	•	••••	1110	×	•	0000		Hydraulic conductivity Effective porosity Specific storage Alteration/solubility					
(IN SITU TEST MET	with deformation monitoring (with or without heated water)	•	•	0	X	•	0	-	Deformation					
	long term with defor- mation monitoring (with or without heated water)	-	-	•	×	-	0	-	Creep/fusing					
	with heated water	•	•		X	-		-	Thermal conductivity Heat Capacity					
	with heated water and displacement monitoring	1	•	-	X	-	-	-	Linear thermal expansion					
	with tracer injection and monitoring (with or without heated water)	:		-	-	-			Dispersivity Adsorption/retardation					

		F	TEST ASSESSMENT VARIABLES									TA 1 1	TY			
					E			RANG	TION	TAL		ASSESSMENT			NT IST	0
(5	TEST METHOD See Notes () at end of Table)	Anisotropy	Scale (Rep. Volume)	Duration	Stress Level	Pore Pressure	Temperature	Radiation Dose	CHARACTERISTICS ASSESSED BY TEST METHOD (see Table 4.4)	Very High	High	Moderate	Low			
	Acoustic emission monitoring	-	•	X	-	-	-	-	Tectonic	F						
	Seismic monitoring	-	•	x	-	-	-	-	Tectonic	-	4			1		
	Monitor pore pressures in rock mass (4)	-	•	-	-	•	-	-	In situ hydraulic field	-			-			
	and drainage into excavation (7)	•	••••		×	×××	××××		Hydraulic conductivity Effective porosity Specific storage			1111				
(500	Monitor temperatures in rock mass (4) and excavation (with ventilation/ cooling) (7) and displacements in rock mass and excavation (7)	-	•		- × - ×		• × × ×		In situ temperature field Thermal conductivity Heat capacity Linear thermal expansion				T			
ST METH	Monitor stress changes and/o displacements in rock mass and excavations (7)		•	•	××	×	××	-	Deformation Creep/fusing	E			11	-		
I SITU TE	Monitor stress changes and strains in supports (7)	•	•	•	××	× -	X X	1 2	Deformation Creep/fusing	E		1 1				
(1)	Monitor fracturing in rock mass around excavation (e.g., by acoustic emission monitoring) (7)	•	•	•	×	×	×	-	Strength	F		7				
	Monitor alteration/ solutioning of exposure	-	•	0	-	x	×	X	Alteration/solubility	-		-				
	Monitor borehole drainage (4,7)	0	0 0 0		×	×××	×××		Hydraulic conductivity Effective porosity Specific storage		TTT					

LIMITATIONS AND UNCERTAINTY IN

LIMITATIONS AND UNCERTAINTY IN ASSESSMENT OF SIGNIFICANT CHARACTERISTICS BY AVAILABLE TEST METHODS

NOTES

- Assessment variable is adequately incorporated in test method.
- Assessment variable can be adequately incorporated in test method (optional).
- Assessment variable is not adequately incorporated in test method.
- Characteristic is relatively insensitive to assessment variable (see Table 3.2), and thus the assessment variable is X not relevant.
- Geologic mapping includes a geodetic survey and exposure mapping, as well as possibly aerophotography (by airplane or satellite, black-and-white, color, or infrared).
- (2) <u>Surface geophysics</u> include possibly gravity and magnetic surveys (land or air based), electrical survey, and seismic surveys (reflection and refraction).
- (3) <u>Coreholes</u> include coring and core logging, as well as possibly borehole surveying, caliper logging, oriented coring, integral sampling, impression packer, borehole TV/camera, borehole radar, and geophysical well logging (electrical, acoustic, and nuclear).
- (4) Tests are conducted in boreholes drilled from underground.
- a) <u>Groundwater sampling</u> implies subsequent laboratory determination of groundwater composition and age. b) <u>Rock mass sampling</u>, either coring or large block samples, implies subsequent laboratory tests. (5) a) Groundwater
- (6) The in situ (i.e., pre-excavation or virgin) stress, hydraulic, and temperature fields can be indirectly assessed or inferred from the stratigraphy/structure and tectonics (e.g., in situ stress field can be inferred from the geomorphology and tectonics of the site).
- (7) Monitoring performance implies associated analysis to assess characteristic.
- (8) The response characteristics refer to the rock mass, which consists of intact rock, discontinuities and pore fluid. These

 - The response characteristics refer to the fock mass, which consists of interesting the discontinuities response characteristics can be assessed either: O Directly by testing a large scale sample which contains a significant number of discontinuities o Indirectly by separately assessing the response characteristics of the intact rock, discontinuitis, and pore fluid, and then assembling by a model. Hence, the rock mass response characteristics can often be inferred from the stratigraphy/structure.
- (9) Index tests do not assess the characteristic directly, but by empirical correlations (e.g., use of a Schmidt harmer or rock core or on an exposure is an index test whose results can be roughly correlated with the modulus of deformation, based on experience). There are too many index tests, with varying reliability, to list.
- (10) Tests assess the response characteristics of only the a) intact rock or b) discontinuity, and does not directly assess the resconse characteristics of the rock mass (see note 8).
- Their jacking is very similar to torsion jacking. Because of these similarities, only shear jacking will be discussed, although torsion jacking might be a suitable alternative. (11)
- (12) Pillar test can consist of either:

35

Eng

10-82

2 Dai

AB13-1163

N. ---

- O Jacking an isolated pillar or unconfined block to failure (i.e., essentially an unconfined plate test or an axially loaded unconfined block test)
- Reducing the dimensions of a pillar (and thus increasing stresses) until failure occurs.
- (13) Plate test is very similar to two other tests:
 - Cable jacking test, in which the reaction is provided by an anchor in the rock mass rather than the opposite wall of 0 the excavation
 - Radial jacking, in which the entire circumference of the opening is jacked using, for example, several plate jack 0 systems

Because of these similarities, only the plate test will be discussed, although cable jacking or radial jacking might be suitable alternatives.

- ar test on discontinuity samples can be of various scales and is also very similar to torsional snear tests. These similarities, only the direct shear test will be discussed, although torsional snear test might be a (14) Direct shear of suitable alternative.
- (15) Tests are constant head injection, constant head withdrawal, constant flow rate withdrawal, pulse injection, or gas injection permeability test. Multiple borehole permeability tests are often called "pump" tests.
- (16) Hydrofracturing must be very carefully performed in order to control the extent of fractures which are generated.

This is a relatively comprehensive list of test methods which are available for assessing site characteristics. It should not be construed that all of these available tests are being recommended.
possible to the prototype in terms of site characteristics, test conditions, and design/construction.

Available simulation test methods, which resolve key issues, have been identified (see Table 4.6). These tests attempt to duplicate a construction or operation method, which is being considered for repository development, and then observe and extrapolate the results to the repository.

The tests with appropriate capabilities can subsequently be selected from this relatively comprehensive list of available tests, in order to respond to the perceived information needs. For those key issues which have already been adequately resolved, or will be by other means, simulation tests may not be necessary. Only those key issues which need additional resolution need to be addressed. Hence, it should not be construed that all of these available tests (Table 4.6) are being recommended.

IN SITU TEST METHODS AVAILABLE TO RESOLVE THE KEY ISSUES BY SIMULATION

Table 4.6 1 of 2

DNSTRUCTABILITY	 Shaft or tunnel test section in which the excavation/ support procedures are varied: determine advance rates and cost for each procedure monitor blast vibrations (if drill and blast) and acoustic emissions for each procedure
	 determine advance rates and cost for each procedure monitor blast vibrations (if drill and blast) and acoustic emissions for each procedure
	 monitor blast vibrations (if drill and blast) and acoustic emissions for each procedure
	cintastons for coord proceedie
	 determine corresponding extent and nature of disturbed zone after excavation
	 monitor performance (i.e., stress, displacement, stability) of opening for each procedure
	 observe rate of weathering, alteration, or solutioning of exposure with time
	 determine effectiveness of ground and groundwater control procedures
	- determine characteristics of as-constructed disturbed zone.
	 Borehole, shaft, or tunnel test section in which the backfill/ seal/plug materials and placement procedures are varied:
	- determine rate of placement and cost of engineered barrier
	- monitor performance of engineered barrier
	 observe rate of alteration, weathering, or solutioning with time
	 determine characteristics of as-placed engineered barrier and rock/barrier interface.
	• Grout injection tests.
	 Evaluate procedures for emplacing and retrieving waste packages.
	 Evaluate operating procedures (i.e., hoisting/transporting, ventilating, drainage/pumping) in shaft/tunnel test sections.*
HERMAL RESPONSE	 Monitor thermal response of tunnel test section (with or without test backfill/seals/plugs) due to emplaced (actual or simulated) waste packages.
MECHANICAL RESPONSE	 Monitor mechanical response (i.e., deformations) of shaft/ tunnel test section* (with or without test backfill/seals/ plugs) in which excavation/support procedures are varied.
	 Rock bolt pull out tests.
HYDROLOGIC RESPONSE	 Monitor hydrological response (i.e., pore pressures and flow) of shaft/tunnel test sections* (with or without test backfill/seals/plugs.)
	지수 물건을 가지 않는 것을 하는 것을 가지 않는 것을 가지 않는 것을 하는 것을 하는 것을 했다.
	THERMAL RESPONSE MECHANICAL RESPONSE

28.01

IN SITU TEST METHODS AVAILABLE TO RESOLVE THE KEY ISSUES BY SIMULATION

Table 4.6

KEY ISSUE

IN SITU TEST METHOD

GEOCHEMICAL RESPONSE

Eng. 85

Dete 1.12

No. 813-1/4 18

B...

- Observe exposures over long term after excavation of shaft/tunnel test sections.*
- Monitor geochemical response (i.e., waste-rock interaction/corrosion/alteration and radionuclide migration) of tunnel test section (with or without test backfill/seals/plugs) due to emplaced waste packages.

* Notes: The entire in situ test facility constitutes shaft/tunnel test sections.

This is a relatively comprehensive list of in situ test methods which are <u>available</u> to resolve the key issues by simulation. It should <u>not</u> be construed that all of these available tests are recommended.

REASONABLE IN SITU TEST PROGRAMS

5.1 PROGRAM CONSIDERATIONS

5.1.1 Testing Which is Expected to Precede In Situ Test Program

Site investigation is expected to precede the in situ test program. This precedent testing, including surface tests and tests within a limited number of boreholes, as well as laboratory tests on samples obtained from those boreholes, has been anticipated (see Table 5.1). The results of actual site investigation will be presented in the Site Characterization Report (SCR), as delineated in "Standard Format and Content of Site Characterization Reports for High-Level Waste Geologic Repositories" (NRC, 1981).

The information needs, to which the in situ test program must respond, will develop as the present assessment of characteristics at each site changes with site investigation prior to initial SCR submittal. It is possible that some characteristics may be adequately assessed for construction authorization prior to initial SCR submittal. In most cases, however, in situ tests will be required to reduce the uncertainty in the assessment of significant characteristics to acceptable levels and adequately resolve the key issues.

5.1.2 In Situ Test Program Objectives

The in situ test program must adequately respond to the information needs for construction authorization which have developed by the time of initial SCR submittal at a given site. This is accomplished by utilizing appropriate tests which:

Reduce the uncertainty in the assessment of significant characteristics to acceptable levels. The uncertainties are reduced by improving the quality of the data base. Large-scale in situ tests, including predicting/monitoring the performance of the in situ test facility itself, assess the characteristics of a representative volume of a rock mass unit which is large enough to contain a significant number of discontinuities. Tests on such a representative volume do not require scale-effect corrections in the determination of quasi-continuum rock mass characteristics, whereas smaller scale tests (e.g., borehole, laboratory, or small-scale in situ tests) assess either the intact rock or the discontinuity characteristics, and not the composite rock mass characteristics. Due to their generally high cost and long duration, however, the results of large-scale in situ tests should be used to develop and verify site-specific correlations with the results of tests which are less expensive and of short duration. Once reliable correlations have been determined, these simpler tests can be used with improved confidence to build a large quality data base. The magnitude and natural variability of characteristics can thus be assessed over a large area.

5.0

TESTING WHICH IS EXPECTED TO PRECEDE IN SITU TEST PROGRAM

8.8 Eng

38

Nor

Date

Rev.

Table 5.1 1 of 3



86

TESTING WHICH IS EXPECTED TO PRECDED IN SITU TEST PROGRAM

Table 5.1 2 of 3

.

~ うちょうしょ

(see Table 3.1)	SURFACE	BOREHOLE (from surface) L	ABORATORY
Creep (9)		• rock mass sampling (5b)	 unconfined com- (No in situ tests are anticipated pression test on rock core, long term (w/or w/o heat) (10a) triaxial test on rock core, long term (w/or w/o heat)(10a)
(THERMAL) Inernal Conductivity (8) Heat catacity (8)		 rock mass sampling (Sb) 	 heated rock sample (10a)
Linear thermal exhansion(3)		 rock mass sampling (5b). 	• heated ruck sample (10a)
(HYDROLOGIC) Hydraulic conductivity (8) Effective porosity (8) Specific storage (8))	 rock mass sampling (5b) geophysical well logging permeability test (single borehole) (15) 	 index tests on rock sample (9) axial permeability test on rock core (w/or w/o heat, stress) (10a,15)
(SEOCHEMICAL) Dispersivity (8) Adsorption/ retardation (8)		 rock mass sampling (5b) groundwater sampling (5a) 	 determination of groundwater sample composition and age tracer test on rock sample (w/or w/o heat, stress) (10a,b) determination of rock sample mineralogy.
Alteration/ solubility	• geologic mapping (1).	• rock mass sampling (5b) • groundwater sampling(5a)	 determination of groundwater sample composition and age core logging determination of rock sample mineralogy slaking or accelerated weathering test on rock sample (w/heat, stress) solubility test on rock sample (w/or w/o heated water)

TESTING WHICH IS EXPECTED TO PRECEDE IN SITU TEST PROGRAM

Table 5.1 3 of 3

NOTES:

CX.

. Test which is expected to precede in situ test program.

- <u>Geologic mapping includes a geodetic survey and exposure mapping, as well as possibly aerophotography (by airplane or satellite, black-and-white, color, or infrared).</u>
- Surface geophysics include possibly gravity and magnetic surveys (land or air based), electrical survey, and seistic surveys (reflection and refraction). (2)
- Corencles include coring and core logging, as well as possibly borehole surveying, caliper logging, oriented corino, integral sampling, impression packer, borehole TV/camera, borehole radar, and geophysical well logging (electrical, acoustic, and nuclear). (3)
- (4) Tests are conducted in boreholes drilled from underground.
- (5) a) Groundwater sampling implies subsequent laboratory determination of groundwater composition and age.
 b) Rock mass sampling, either coring or large block samples, implies subsequent laboratory tests.
- (6) The in situ (i.e., pre-excavation or virgin) stress, hydraulic, and temperature fields can be indirectly assessed or inferred from the stratigraphy/structure and tectonics (e.g., in situ stress field can be inferred from the geomorphology and tectonics of the site).
- (7) Monitoring performance implies associated analysis to assess characteristic.
- (8) The response characteristics refer to the rock mass, which consists of intact rock, discontinuities and pore fluid. These
 - response characteristics can be assessed either: O Directly by testing a large scale sample which contains a significant number of discontinuities o Indirectly by separately assessing the response characteristics of the intact rock, discontinuitis, and pore fluid, and then assembling by a model. Hence, the rock mass response characteristics can often be inferred from the stratigraphy/structure.
- (9) Index tests do not assess the characteristic directly, but by empirical correlations (e.g., use of a Schmidt hammer or rock core or on an exposure is an index test whose results can be roughly correlated with the modulus of deformation, based on experience). There are too many index tests, with varying reliability, to list.
- (10) Tests assess the response characteristics of only the a) intact rock or b) discontinuity, and does not directly assess the response characteristics of the rock mass (see note 8).
- Shear jacking is very similar to torsion jacking. Because of these similarities, only shear jacking will be discussed, although torsion jacking might be a suitable alternative. (11) Shear
- (12) Pillar test can consist of either:
 - Jacking an isolated pillar or unconfined block to failure (i.e., essentially an unconfined plate test or an axially 0 loaded unconfined block test)
 - o Reducing the dimensions of a pillar (and thus increasing stresses) until failure occurs.
- (13) $\frac{\text{Plate test}}{\text{O}}$ is very similar to two other tests: $\frac{13}{\text{O}}$ Cable jacking test, in which the reaction is provided by an anchor in the rock mass rather than the opposite wall of the excavation
 - Radial jacking, in which the entire circumference of the opening is jacked using, for example, several plate jack 0 systems

Because of these similarities, only the plate test will be discussed, although cable jacking or radia! jacking might be suitable alternatives.

- (14) Direct shear test on discontinuity samples can be of various scales and is also very similar to torsional shear tests. Because of these similarities, only the direct shear test will be discussed, although torsional shear test might be a suitable alternative.
- (15) Tests are constant head injection, constant head withdrawal, constant flow rate withdrawal, pulse injection, or gas injection permeability test. Multiple borehole permeability tests are often called "pump" tests.

(16) Hydrofracturing must be very carefully performed in order to control the extent of fractures which are generated.

3E

Eng

- Partially verify predictive numerical models, by comparing the measured results of large-scale in situ tests with predictions. This includes comparing predicted performance of the in situ test facility with monitored performance.
- Simulate important aspects of the repository. By monitoring the performance of the in situ test facility, some of the data collected during construction and operation of this facility can be extrapolated to predict repository construction/operation performance. The validity of this extrapolation will be a direct function of how well the in situ test facility represents the repository, both in site characteristics and design/construction.

The in situ test program is assumed to consist of tests which are available (Tables 4.4 and 4.6), and are conducted within an in situ test facility in the time frame between initial SCR submittal and LA. This in situ test facility is assumed to consist of an exploratory shaft, extending from the surface to the prospective repository horizon possibly with test stations at various depths, and an underground test facility, consisting of appropriate tunnels and test rooms at that horizon.

The exploratory shaft should be constructed to:

- Allow description of the site below the surface (including the repository horizon) near the shaft
- Provide access to the repository horizon
- Provide information on shaft design and construction engineering variables
- Provide information on shaft operation (i.e., ventilation, hoisting, etc.) by monitoring
- Not adversely impact repository performance.

The underground test facility should be constructed, if the site has not been disgualified after construction of the exploratory shaft, to:

- Allow description of the repository horizon
- Provide information on tunnel design and construction engineering variables
- Provide information on tunnel operation (i.e., ventilation, transportation, etc.)
- Not adversely impact repository performance.

It is expected that plans for this in situ test program will be presented in the initial SCR submittal and that the complete results will be presented in a LA. Testing preceding the initial SCR submittal (Table 5.1) will affect the in situ test program, as the uncertainty in the assessment of some characteristics is reduced and the information needs correspondingly change. Similarly, testing or monitoring subsequent to LA (i.e., during repository construction/operation) may be required to further increase the level of confidence in satisfactory performance to acceptable levels. This subsequent testing has not been considered, however.

5.1.3 Tentative Media Independent In Situ Test Program

Tests which should generally be considered for inclusion in an in situ test program have been identified (see Table 5.2, in conjunction with Table 4.6). This selection has been based on the objectives of the in situ test program (see Section 5.1.2), especially based on:

- The perceived information needs for construction authorization which will typically exist when the initial SCR is submitted, which in turn depend on:
 - The maximum acceptable level of uncertainty in the assessment of the significant characteristics (see Section 3.5)
 - The present assessment of significant characteristics (see Section 3.4)
 - The tests which are expected to precede in situ testing (see Section 5.1.1)
- The assessed capabilities of each test in addressing these information needs, which in turn depend on:
 - The typical level of uncertainty in the assessment of each significant characteristic by available test methods (see Section 4.3)
 - In situ tests which are available to resolve the key issues by simulation (see Section 4.4)

Specific in situ test methods, which comprise this program and are especially unique and important, have been investigated in detail (see Appendix A of Volume II). These methods have been individually described and evaluated, and Golder Associates has made recommendations regarding test methodology, utilization of test results, and potential research/ development.

As previously discussed, however, neither the information needs for construction authorization nor the test capabilities at the time of initial SCR submittal can be firmly established at this time. Also, neither are independent of media or site. Thus, this tentative program (Tables 5.2 and 4.6) provides only a preliminary indication of an adequate in situ test program.

TENTATIVE IN SITU TEST PROGRAM FOR ASSESSING SIGNIFICANT CHARACTERISTICS

Table 5.2

CHARACTERISTIC (see Table 3.1)	TEST METHOD	(see noter:) at end of Tat) LASORATORY IN ST	e) (See Table 4.4) TJ (from subsurface excavation)
(GEOLOGIC SETTING) Stratioraphic/ structural (includes the physical and chemical char- acteristics of each rock mass unit, including pore fluid composition)	(Refer to Table 5.1 for tests anticipated to be conducted from surface or in bore- holes drilled from surface prior to in situ test program.)	 core logging determination of rock sample mineralogy (thin section, x-ray diffraction) determination of groundwater sample composition and age hydrochemical analysis of rock sample. 	 exploratory excavations (including exposure mapping) coreholes (3, 4) rock mass sampling (5b) geophysic/seismic (x-hole, exposure-borehole)
Tectonic In situ stress field (stress tensor) (6)			 acoustic emission monitoring seismic monitoring hydrofracturing (4) overcoring (4) flat jack test
In situ hydraulic head field "hore hrecsure)			 monitor pore pressures in rock mass (4)
In situ tempera- ture field "temperature"(6)			 monitor temperatures in rock mass (4)
(MECHANICAL) Strength (8)		 index tests on rock samples (9) simple strength tests on rock sample (sliding test, Brazilian test, point load test, beam test, tensile strength test, fracture toughness test) (loa) direct shear test on discontinuity sample (w/or w/o heat) (lob, 14) unconfined compression test on rock core (w/or w/o heat) (loa) triaxial test on rock core (w/or w/o heat) (loa,b). 	<pre>index tests on exposures (9) rock mass sampling (5b) mine-by test monitoring fracturing in rock mass around excavation (e.g., by acoustic emission monitoring) (7)</pre>
Deformation (8)		 index tests on rock sample (9) unconfined com- pression test on rock core (w/or w/o heat) (10a) triaxial test on rock core (w/or w/o heat) (10a,b) true triaxial test on large rock sample (w/or w/o heat) (10a,b) direct shear test on discontinuity sample (w/or w/o heat) (10b, 14) sonic velocity test on rock core (w/or w/o stress) (10a). 	 index tests on exposures (4) rock mass sampling (5b) geophysic/seismic (x-hole, exposure-borehole) flat jack test plate test (13) block test (w/or w/o heat) mine-by test monitor displacements in rock mass as excavation occurs (4,7) monitor displacements at exposure as excavation occurs(7)

TENTATIVE IN SITU TEST PROGRAM FOR ASSESSING SIGNIFICANT CHARACTERISTICS

0

N.

Table 5.2 2 of 3

CHARACTERISTIC TEST METHOD (see notes() at end of Table) (See Table 4.4) LABORATORY IN SITU (from subsurface excavation) (see Table 3.1) rock mass sampling (55)
flat jack test, long term
plate test, long term (13)
block test, long term 'w/or with (Refer to Table 5.1 for tests anticipated to be conducted from surface or in bore-holes drilled from surface prior to in · unconfined com-Creep (8) pression test on rock core, long term (w/or w/o situ test program.) heat) (10a) • triaxial test on heat) • mine-by test triaxial test on mine-by test
 rock core, long term (w/or w/o heat)(l0a)
 true triaxial test on an area rock sample, long term (w/or w/o monitor displacements in rock mass after excavation (4,7) exposure after excavation [4," heat) (10a,b)
 direct shear test on discontinuity sample, long term (w/or w/o heat) (10b,14) • rock mass samp'ind [55 (THERMAL) · heated rock Thermal sample (10a) • unconfined com-• neater test (x-hole small conductivity (8) scale) (4)
• heater test (large scale) Heat cabacity (8) pression test on rock core w/heat · block test w/heat (w/or w/o stress)
monitor temperature in rock mass (10a) • triaxial test on and excavation (w/ventilation cooling) (7). rock core w/heat (10a). • rock mass sampling (5b) · heated rock Linear thermal sample (10a)
unconfined comheater test ('arge scale)
block test w/heat (w/or w/o expansion(3) pression test on stress) rock core w/heat • monitor temperatures and displacements in rock mass and excavation (7). (10a) · triaxial test on rock core w'heat (10a) rock mass sampling (5b)
 permeability test (strole borehole) (w/or w/o heated water) (4,15) (HYOROLOGIC) • index tests on rock Hydraulic sample (9) • axial permeability conductivity (8) Effective porosity (8 Specific storage (8) test on rock core (w/or w/o heat, stress) (10a,15) a radial permeability • multiple borehole permeability test (w/or w/o heates water) (4,15) test on rock sample (w/or w/o heat, stress)(10a,b, 15). · block test w/multip'e boreno'e permeability test w or w/o heat, stress) · chamber test (w/or w/o heated water) (15) e monitor drainade into excavation and pore pressure in rock mass determination of groundwater sample composition and age tracer test on rock
 tracer test on rock
 tracer test (w/or w/o neated sample (w/or w/o heat, water) (4)
 stress) (10a,b)
 block test w/tracer test (w/ or w/o neat, stress) (GEOCHEMICAL) x Dispersivity (8) Adsorption/ 80 retardation (8) Eng 28 mineralogy. NON · determination of · exposure mapping Alteration/ groundwater sample composition and age • rock mass sampling (55) Date. sclubility composition and age groundwater sampling (4,5a) core logging heater test (large scale) determination of chamber test (w/or wio neat) rock sample mineralogy monitor alteration solutioning slaking or of exposures. core logging
determination of · slaking or accelerated weathering test on rock sample A 813-1163 (w/heat, stress) e solubility test on rock sample (w/or w/o heated water) No. Dwg. 92

TENTATIVE IN SITU TEST PROGRAM FOR ASSESSING SIGNIFICANT CHARACTERISTICS

Table 5.2 3 of 3

Notes:

- Test which should be specifically considered in a reasonable in situ test program. The actual selection of tests will be a function of the information needs for a specific site, as well as the test capabilities regarding those needs. Hence, it is not to be expected that all of the indicated tests will be required at each site.
- <u>Geologic mapping includes a geodetic survey and exposure mapping, as well as possibly aerophotography (by airplane or satellite, black-and-white, color, or infrared).</u>
- (2) <u>Surface geophysics include possibly gravity and magnetic surveys (land or air based)</u>, electrical survey, and seismic surveys (reflection and refraction).
- Coreholes include coring and core logging, as well as possibly borehole surveying, caliper logging, oriented coring, integral sampling, impression packer, borehole TV/camera, borehole radar, and geophysical well logging (electrical, acoustic, and nuclear). (3)
- (4) Tests are conducted in boreholes drilled from underground.
- a) <u>Groundwater sampling</u> implies subsequent laboratory determination of groundwater composition and age. b) <u>Rock mass sampling</u>, eitner coring or large block samples, implies subsequent laboratory tests. (5)
- (6) The in situ (i.e., pre-excavation or virgin) stress, hydraulic, and temperature fields can be indirectly assessed or inferred from the stratigraphy/structure and tectonics (e.g., in situ stress field can be inferred from the geomorphology and tectonics of the site)
- (7) Monitoring performance implies associated analysis to assess characteristic.
- (8) The response characteristics refer to the rock mass, which consists of intact rock, discontinuities and pore fluid. These response characteristics can be assessed either
 - o Directly by testing a large scale sample which contains a significant number of discontinuities o Indirectly by separately assessing the response characteristics of the intact rock, discontinuitis, and pore fluid, and then assembling by a model. Hence, the rock mass response characteristics can often be inferred from the stratigraphy/structure.
- (9) Index tests do not assess the characteristic directly, but by empirical correlations (e.g., use of a Schmidt hammer on rock core or on an exposure is an index test whose results can be roughly correlated with the modulus of deformation, based on experience). There are too many index tests, with varying reliability, to list.
- (10) Tests assess the response characteristics of only the a) intact rock or b) discontinuity, and does not directly assess the response characteristics of the rock mass (see note 8)
- Shear jacking is very similar to torsion jacking. Because of these similarities, only shear jacking will be discussed, although torsion jacking might be a suitable alternative. (11) Shear
- (12) Pillar test can consist of either:

BE

Eng

10-82

0.1e

4313-1165

No

- Jacking an isolated pillar or unconfined block to failure (i.e., essentially an unconfined plate test or an axially loaded unconfined block test) 0
- Reducing the dimensions of a pillar (and thus increasing stresses) until failure occurs. 0
- (13) <u>Plate test</u> is very similar to two other tests: <u>0</u> Cable jacking test, in which the reaction is provided by an anchor in the rock mass rather than the opposite wall of the excavation
 - o Radial jacking, in which the entire circumference of the opening is jacked using, for example, several plate jack systems.

Because of these similarities, only the plate test will be discussed, although cable jacking or radial jacking might be suitable alternatives.

- (14) Direct shear test on discontinuity samples can be of various scales and is also very similar to torsional shear tests. Because of these similarities, only the direct shear test will be discussed, although torsional shear test might be a suitable alternative.
- (15) Tests are constant head injection, constant head withdrawal, constant flow rate withdrawal, pulse injection, or gas injection permeability test. Multiple borehole permeability tests are often called "pump" tests.

(16) Hydrofracturing must be very carefully performed in order to control the extent of fractures which are generated.

5.2 TENTATIVE MEDIA/SITE SPECIFIC IN SITU TEST PROGRAMS

Tests which should be specifically considered for inclusion in an in situ test program for each of the media/sites under consideration have been identified (see Table 5.3). These selections of tests are derived from the tentative in situ test program (see Section 5.1.3), taking into additional consideration:

- The present assessment of characteristics for each media/site (see Section 3.4.2 and Appendix B of Volume II)
- The significance of characteristics for each media/site (see Section 3.5.2)
- The applicability of each test to the media/site specific conditions (see Appendix A of Volume II).

The media and sites considered include (Table 1.1):

- Basalt at Hanford Reservation, Washington
- Tuff at Yucca Mountain, Nevada Test Site, Nevada
- Domal salt at Richton or Cypress Creek, Mississippi, or Vacherie, Louisiana
- Bedded salt at unspecified site (generic)
- Granite at unspecified site (generic).

Golder Associates believes that the tentative media/site specific in situ test programs (Table 5.3) offer the best available opportunity for adequately responding to the presently perceived information needs, and thus resolving the key issues related to repository design and construction, prior to LA.

However, the information needs may change between now and when the initial SCR is submitted, especially for granite and to a lesser extent domal salt, bedded salt, and tuff. This is due to the additional data which might be generated by surface, borehole and laboratory testing performed prior to initial SCR submittal, and thus preceding in situ testing, at these sites. This is also due to possible clarification of the licensing perspective. Similarly, the capabilities of specific tests will generally improve in the future and even new or hybrid tests, which have not been considered herein, may be developed having significantly different capabilities.

Due to the possible changes in both information needs and test capabilities, the reasonable media/site specific in situ test programs (Table 5.3) are preliminary in nature and will evolve with time. In situ test programs must thus incorporate some flexibility in order to easily adapt to these changes, as presented in SCR updates.

TENTATIVE MEDIA/SITE SPECIFIC IN SITU TEST PROGRAMS

Table 5.3

1

1 of 3

IN SITU TEST METHOD (Appendix, Section):	CHAP	RAC'	TERI	STIC (see	S AS	SESS 5.2	ED B	Y TE	ST	ASPE SIMU BY T (see Table	CTS LATED EST: e 4.6)	MININ NUMBI OF TI PERFO IN:	NUM ER ESTS ORMED	MEI WHI REC	OIA/ ICH OMN See	SIT TES IEND able	E F T I ED	OR
	Geo	108	tic 1	Sett	ing	()	()	(2	(2)									
	Stratigraphic/ Structural (1a)	Tectonics	In Situ Stress Field (1b)	In Situ Hydraulic Head Field (1b)	In Situ Tempera- ture Field (1b)	Mechanical Characteristics (2	Thermal Characteristics (2	Hydrologic Characteristics (2	Geochemical Characteristics (2	Constructability	Repository Performance	Exploratory Shaft	Underground Test Facility	Basalt (3)	Tuff (4)	Domal Salt (5)	Bedded Salt (6)	Grantto (7)
Plate Test (9) (A.1)						•						-	a	•	•			•
Block Test (A.2)						•	•	•	•			-	2	•	•		•	•
Chamber Test (A.3)									•		•	-	1	•	•			
Mine-By Test (A.4)						•				•	•	-	1	•	•	•	•	
Heater Test (Large Scale) (A.5)							•		•	•	•	-	1	•	•	•	•	
Heater Test (Small Scale) (A.5)							•					a	a	•	•	•	•	
Tracer Test (A.6)			T	•				•	•			a	a	•	•			
Multiple Borehole Permeability Test (10) (A.7)								•				a	a		•			
Overcoring (A.8)												b	b	•				
Flatjack Test (A.9)			•			•						a	a	•	•	•	•	
Acoustic Emission Monitoring (A.10)										•	•	c	c	•				
Exposure Mapping (A.11)												c	с			•		

Dwg No 28.7.463 Date 10 82

Eng 2/8

95

Rev.	_				
Dwg.	No.	1813-1163	Dete	10.82	Eng DR

Operation	Construction Monitoring	Monitor Alteration, Solutioning of Exposures	Monitor Drainage into Excavation(16)	Monitor Pore Pressures in Rock Mass (16)	Monitor Displace- ments in Rock Mass and Excavations (16)	Monitor Temperature in Rock Mass and Excavations(16)	Seismic Monitoring	Geophysic/Seismic (X-hole, Exposure-borehole)	Index Tests on Exposures (15)	Laboratory Tests (14)	Hydro- fracturing (17)	Permeability Test (Single Borehol (10)	Groundwater Sampling (13a)	Rock Mass Sampling (13b)	Coreholes (12)	Exploratory Excavations (11)			IN SITU TEST KETHOD (Appendix, Section):
			1							•				•			Stratigraphic/ Structural (Ia)	Ge	CH
	-		-														Tectonics	010	RAC
			1				-										In Situ Stress Field (1b)	810	TER
-			+-														In Situ Hydraulic	Se	(se
				•			<u>.</u>										Head Field (1b)	11	* 1C
						•											In Situ Tempera- ture Field (1b)	ng a	ble AS
					•			•	•	•		1.14		•			Mechanical Characteristics (2)	SSES
					•	•				•				•			Thermal Characteristics (2	,	SED
				•	-				-1	•		•		•			Hydrologic Characteristics (2	,	3Y T1
		•								•			•	•			Geochemical Characteristics (2	,	EST
	•		•		•	•		•	•	•					•	•	Constructability		ASPE SINU BY T BY T
•	•	•	•	•	•	•	•									•	Repository Performance		CTS LATED EST: e 4.6
n	0	0	0	0	0	0	0	σ	σ	σ	8	σ	σ	σ	σ	*	Exploratory Shaft		MININ NUMB OF TI PERFY
0	0	0	0	0	0	0	0	σ	σ	σ	σ	σ	σ	σ	σ	*	Underground Test Facility		NUM ER ERSTS ORMED
		•	•	•	•			•	•	•		•		•	•	•	Basalt (3)		REE
•	•		•	•		•	•	•	•	•	•	•	•	•	•	•	Tuff (4)		DIA ICH COM
•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	Domal Salt (5)		APIe
•	•		•		•	•	•	•	•	•	•	•	•	•	•	•	Bedded Salt (6)		E TE
			•	•	•	•		•	•			•			•	•	Granite (7)		SA

.

10

TENTATIVE MEDIA /SITE SPECIFIC Table 5.3 IN SITU TEST PROGRAMS 3 of 3 Notes: a = several (≤ 10) b = numerous (>10) C = continual ★ = Exploratory shaft and underground test facility are exploratory excavations. (1) a) The stratigraphy/structure includes the physical and chemical characteristics of each rock mass unit (including poor fluid composition). b) The in situ (i.e., pre-excavation or virgin) stress, hydraulic, and temperature fields can be indirectly assessed or inferred from the stratigraphy/structure and tectonics (e.g., in situ stress field can be inferred from the geomorphology and tectonics of the site). (2) The response characteristics refer to the rock mass, which consists of intact rock, discontinuities and pore fluid. These reponse characteristics can be assessed either: Directly by testing a large scale sample which contains a significant number of discontinuities Indirectly by separately assessing the response characteristics of the intact rock, discontinuities, and pore fluid, and then assembling by a model. Hence, the rock mass response characteristics can often be inferred from the stratigraphy/structure. (3) Baselt at Hanford, Washington (See Section 3 and Appendix 8 of Volume II) (8) (4) Tuff at Yucca Mountain, Nevada (See Section 3 and Appendix B of Volume 11)(8) (5) Domal Salt at Gulf Coast Sites (See Section 3 and Appendix 8 of Volume 11) (8) (6) Bedded Salt at unspecified site (See Section 3 and Appendix 8 of Volume II)(8) (7) Granite at unspecified site (See Section 3 and Appendix 8 of Volume [1](8) (8) Significance of characteristics for in situ testing, as they relate to design, for each media/site subjectively evaluated as shown in Figure 3.5. (9) Plate test is very similar to two other tests: Cable jacking test, in which the reaction is provided by an anchor in the rock mass rather than the opposite wall of the excavation. Radial jacking, in which the entire circumference of the opening is jacked using, for example, several plate jack systems. Because of these similarities, only the plate test will be discussed, although cable jacking or radial jacking might be suitable alternatives. (10) Tests are constant head injection, constant head withdrawal, constant flow rate withdrawal, pulse injection, or gas injection permeability test. (11) Exploratory excavations include the exploratory shaft and underground test facility. (12) Coreholes include coring and core logging, as well as possibly borehole surveying, caliper logging, oriented coring, integral sampling, impression packer, borehole TV/camera, borehole radar, and geophysical well logging (electrical, acoustic, and nuclear). (13) a) Groundwater sampling implies subsequent laboratory determination of groundwater composition and age. b) Rock mass sampling, either coring or large block samples, implies subsequent laboratory tests. (14) Laboratory tests are performed on rock mass or groundwater samples (See Table 5.2 for a complete listing). (15) Index tests do not assess the characteristic directly, but by empirical correlations (e.g., use of a Schmidt hammer on rock core or on an exposure is an index test whose results can be roughly correlated with the modulus of deformation, based on experience). There are too many index tests, with varying reliability, to list. (16) Monitoring performance implies associated analysis to assess characteristic. (17) Hydrofracturing must be very carefully performed in order to control the extent of fractures which are generated.

BR

Eng

28.01

0.1.

163

A.BL3.

Dwg No.

97

.

5.3 EXAMPLE IN SITU TEST FACILITY FOR BASALT AT HANFORD, WASHINGTON

The configuration, schedule, and cost of a reasonable in situ test facility which can accommodate the tentative in situ test program has been developed in Task 4, "Relationship of an In Situ Test Facility to a Deep Geologic Repository for High Level Nuclear Waste" (Golder, 1982d), of this project. The development of the example in situ test facility has utilized the geologic setting, the existing exploratory shaft design, and the preconceptual repository design for the Basalt Waste Isolation Project (BWIP) at Hanford, Washington. BWIP was chosen as an example case because the repository program was the most advanced and more information was available, as compared to other media/sites.

The example configuration of a reasonable underground test facility for basalt (see Figure 5.1) has been primarily influenced by the following factors:

- Accommodate example in situ test program
- Construct certain test and access tunnels to repository dimensions and orientations to evaluate full scale opening response
- Spatially separate experiments to reduce test interaction
- Confirm the extent of the proposed repository host rock mass and evaluate the variability of characteristics in two orthogonal directions (long, near-horizontal boreholes can be added to extend this investigation)
- Provide sufficient working area, access and utilities for quality assurance and safety.

The schedule for the development of the example in situ test facility for basalt has been estimated to be about 66 months, from mobilization to completion of the test program; however, this could possibly be accelerated to about 47 months.

The cost of the example in situ test facility construction, in situ testing, and facility operation has been estimated to be about \$60 million, of which \$4 million and \$19 million have been estimated for testing in the shaft and the underground facility, respectively.

The configuration, schedule and cost of an in situ test facility at other potential repository sites will probably vary from the example developed for basalt at Hanford, Washington. The test program, and corresponding configuration, schedule and costs of an example test facility, have been developed based on presently perceived information needs and test capabilities. Either of these may change with time, so that the in situ test program and facility may also change, and thus should be considered tentative and remain flexible.



RECOMMENDED IN SITU TESTING RESEARCH

6.1 INTRODUCTION

6.0

Although the tentative media/site specific in situ test programs (see Section 5.2), in conjunction with the recommendations regarding specific in situ tests (see Appendix A of Volume II), provide the best available means for resolving the key issues, there are some remaining limitations. In order to reduce or mitigate these remaining limitations, Golder Associates recommends that research be conducted in three broad areas:

(1) Existing tests

Recommended areas of specific research (especially in monitoring or measurement) in tests already selected (see Appendix A of Volume II).

(2) New or hybrid tests

Research and development of new or hybrid in situ tests which will improve the understanding of fundamental laws and assist in the development and verification of complex coupled predictive models. (An example would be the coupling of thermal effects with bulk modulus properties, as measured in a Plate Test.)

(3) Program integration

Of considerable importance to schedules and cost in construction of a geologic repository is to establish how many, how extensive, where, and when in situ tests should be performed during the process of repository development.

Each of these research topics is discussed in the following subsections. Also, research needs have been previously discussed by others for many areas of rock mechanics (e.g., National Research Council, 1981) and specifically for repository development (e.g., LBL, 1979).

6.2 POTENTIAL IMPROVEMENTS TO EXISTING IN SITU TESTS

6.2.1 Plate Test

The recommendations for research and development regarding the plate test (see Appendix A of Volume II, Section A.1) include:

- Development of improved modeling of the plate test for determination of various deformation parameters
- Development of improved correlations of plate test results with other simple tests

- Modification of the plate test to incorporate variables which affect the deformational characteristics, e.g.:
 - temperature
 - radiation
 - water saturation/pore pressure
 - confining stress
- Development of improved waterproofing of electrical instrumentation and improved corrosion resistance for all equipment
- Development of more sensitive, accurate, and stable electronic deformation monitoring instrumentation, especially for long-term creep tests
- Development of more sensitive, accurate, and stable application of boundary conditions, especially for long-term creep tests
- Development of more accurate determination of boundary conditions
- Elimination of MPBX anchor slippage and stress-induced flowage at the edge of the loaded area, especially in salt.

6.2.2 Block Test

The recommendations for research and development regarding the block test (see Appendix A of Volume II, Section A.2) include:

- Development of improved modeling (i.e., discontinuum) of the block test for determination of various thermal, thermomechanical and mechanical parameters
- Development of improved correlations of block test results with other in situ tests
- Development of improved waterproofing of electrical instrumentation and improved corrosion resistance for all equipment
- Development of more convenient, sensitive, accurate, and stable electronic or optical subsurface deformation/strain monitoring instrumentation, especially for long-term creep tests
- Development of improved joint permeability measurement methods and related analytical models
- Improvement in the methods of stress tensor determination for use in the block test heating phases.

1

6.2.3 Chamber Test

The recommendations for research and development regarding the chamber test (see Appendix A of Volume II, Section A.3) include:

- Investigation of flow in natural fractures. Statistical methods can
 possibly be developed to characterize the aperture, orientation,
 attitude, and persistence of fractures in geologic materials. Such
 efforts may result in the development of practical, discontinuum
 numerical models for flow in fractured media.
- Investigation of the relationship between the dimensions of fracturing and the size of a representative elementary volume (REV). This would allow a better assessment of the use of porous media models for describing flow in fractured media.
- Refinement of algorithms for automatic calibration of hydrologic parameters in groundwater models (inverse models)
- Modification of the chamber test to incorporate other variables which affect the hydrologic response of a rock mass:
 - stress environment
 - temperature
 - rock and fracture deformability
 - anisotropy
- Development of improved numerical modeling of the chamber test for determination of various hydrologic, thermal, and deformational parameters.

6.2.4 Mine-by Test

The recommendations for research and development regarding the mine-by test (see Appendix A of Volume II, Section A.4) include:

- Development of a borehole instrument that will provide continuous monitoring of both longitudinal and transverse deformations at 1 m intervals along the borehole
- Development of a borehole instrument that can monitor stress changes in three dimensions
- Development of improved modeling of the mine-by test for determination of various deformational parameters
- Further development of improved correlations of mine-by test results with other simple tests and empirical relations.

6.2.5 Heater Test

The recommendations for research and development regarding heater tests (see Appendix A of Volume II, Section A.5) include:

- Developments to improve sensitivity, reliability, and durability for all instruments which must operate for long periods at high temperatures and under adverse moisture conditions
- Development of numerical modeling techniques to predict rock mass thermomechanical response. An understanding must be achieved of coupled reactions, such as the effects of temperature and stress on the thermal and thermomechanical properties, and heat transfer by conduction and convection in an anisotropic, nonhomogeneous media.
- Development of correlations of the thermal and thermomechanical rock mass parameters derived from the heater tests with those derived from laboratory tests, as well as correlations with those derived from heated block tests.

6.2.6 Tracer Test

The recommendations for research and development regarding tracer tests (see Appendix A of Volume II, Section A.6) include:

- Investigation of solute transport in natural fractures. Statistical methods can possibly be developed to characterize the aperture, orientation, attitude, and continuity of fractures in geologic materials. Such efforts may result in the development of practical, discrete, numerical models of solute transport in fractured media.
- Investigation of the relationship between the dimensions of fracturing and the size of a representative elementary volume (REV). This would allow a better assessment of the use of porous media models for describing solute transport in fractured media.
- Modification of the tracer test methodology and analysis to incorporate other variables (such as temperature, rock and fracture deformability and anisotropy) which affect the hydrologic response of a rock mass
- Development of laboratory techniques for determination of the equilibrium distribution coefficient to verify calculated values from tracer tests
- Investigation to define the variance of a measured value and to determine the relationship of the value measured to the scale of testing.

6.2.7 Multiple Brrehole Permeability Test

The recommendations for research and development regarding multiple borehole permeability tests (see Appendix A of Volume II, Section A.7) include:

 Investigation of flow in natural fractures. Statistical methods can possibly be developed to characterize the aperture, orientation, attitude, and persistence of fractures in geologic materials. Such efforts may result in the development of practical discontinuum numerical models for flow in fractured media.

.

- Investigation of the relationship between the dimensions of fracturing and the size of a representative elementary volume (REV). This would allow a better assessment of the use of porous media models for describing flow in fractured media.
- Refinement of algorithms for automatic calibration of hydrologic parameters in groundwater models (inverse models)
- Modification of the multiple borehole permeability test to incorporate other variables which affect the hydrologic reponse of a rock mass:
 - temperature
 - rock and fracture deformability
 - anisotropy
- Development of improved numerical modeling of the multiple borehole permeability test for determination of various hydrologic, thermal, and deformational parameters.

6.2.8 Overcoring

The recommendations for research and development regarding overcoring (see Appendix A of Volume II, Section A.8) include:

- Improvement of adhesives, i.e., maximizing reliability and strength and minimizing setting time and time-dependent characteristics, for the adverse conditions found in drilling.
- Improvement of present designs of installation devices to make them more convenient to use and more reliable
- Reduction in the cost of in situ measurements. These efforts should include reducing the cost of each measurement, reducing the number of unsuccessful measurements, and reducing the number of successful measurements necessary to establish a local stress field with confidence.

- Improvement of strain gauges and their accuracy. Specifically, some work needs to be performed on eliminating or reducing the effects of temperature, length of electrical cables, or other phenomena which adversely impact the reliability of strain readings. These improvements might easily be effected by manufacturers.
- Continued investigation into methods to improve data reduction and instrument calibration
- Continued improvement in the measurements of the in situ mechanical properties of the rock mass required in analysis
- Development of refined analysis methods for nonlinear or anisotropic rock
- Development of techniques to extend the depth to which in situ stress measurements can be made by overcoring techniques.

6.2.9 Flatjack Test

The recommendations for research and development regarding the flatjack test (see Appendix A of Volume II, Section A.9) include:

- Investigation, possibly coordinated with repository geotechnical investigations, into large scale flatjack testing
- Development of the flatjack test for use deeper in the rock mass
- Continued investigations into numerical and analog techniques for improved interpretation of flatjack measurements.

6.2.10 Acoustic Emission Monitoring

The recommendations for research and development regarding acoustic emission monitoring (see Appendix A of Volume II, Section A.10) include:

- Development of more sensitive seismometers to detect lower amplitude emissions
- Development of telemetry and fiber-optics systems to transmit signals more efficiently from remote seismometers to a central recording station
- Development of more sophisticated analysis techniques for event location and source parameter determination.

6.2.11 Exposure Mapping

The recommendations for research and development regarding exposure mapping (see Appendix A of Volume II, Section A.11) include:

- Development of a standard scratch hardness tester that would apply a constant force to the specimen. This would help eliminate operator variability. The tester would be similar to a pocket penetrometer used for soils, but would have a sharpened tip for scratching the specimen. A standard scratch hardness tester is already available for laboratory use.
- Development of a "full perimeter tunnel camera." This would be similar to a borehole camera, but would simultaneously photograph the crown, walls, and invert of a tunnel.
- Further development of computer methods for storing, analyzing, and plotting geologic information
- Further investigation of statistical analysis of discontinuity characteristics

6.3 DEVELOPMENT OF NEW OR HYBRID TESTS

Some of the available tests selected for resolving the key issues of repository development are aimed at establishing the relationship of a characteristic with the assessment variables. However, in many cases, only a certain range (which is often insufficient) in the assessment variables can be evaluated by any one test presently available. The extension of this range to cover the conditions expected during the repository life should be attempted. For example, the chamber test could be further complicated by the addition of a fluid at a variety of elevated temperatures. This would also enable a further clarification of the laws relating heat and fluid flow. Such research, with similar improvements in modeling of such tests, may be the only way in which model uncertainty in specific site performance evaluations can be adequately reduced. The minimum acceptable level of confidence in satisfactory performance will control whether or not such research is required for a particular site. Stated in another way, research should be carried out at a particular site only if there is a clear benefit in establishing or predicting performance. If justified, R&D may need to be accelerated in order to meet the constraining schedules of repository development.

It is not within the scope of this project to detail research needs under the general heading of improving fundamental knowledge. It is thus only important to establish the areas where, for specific sites, the current knowledge may be unacceptably low. These areas include:

- (1) The interaction between heat and fluid transport in a fractured medium, which cannot be adequately characterized by porous flow behavior. This coupling of flow through fractures under a variety of heat loads has been studied in the laboratory, but very little work has been carried out in situ.
- (2) The relationship between induced stress and fracture aperture/ deformation, and the resulting effect on hydraulic conductivity. This is only partially understood from limited field and laboratory experiments.
- (3) The transport of specific radionuclides. The effect of assessment variables (e.g., temperature), in conjunction with the material type and pore fluid composition, on adsorption/retardation of various radionuclides is not well understood. At present, the determination of the equilibrium distribution coefficient (K_d) , which dominates repository performance assessment, is very uncertain.

6.4 INTEGRATION OF IN SITU TEST PROGRAM IN REPOSITORY DEVELOPMENT

The integration of the in situ test program in repository development, i.e., number, extent, location, and timing of in situ tests relative to the repository configuration and schedule, is a critical question which demands site specific consideration. This topic has been addressed under Task 4 of this project (Golder Associates, 1982d) (see Section 5.3). Exploration by means of subsurface development, i.e., a system of tunnels (or boreholes) and an access shaft, is a most important step in deciding on whether further investment should be made at a particular site. This decision must be based on the observed variability of the geological setting, as well as on the results of tests and hydrologic and mechanical response of the test facility due to excavation of subsurface openings. The possibility of modular development of the in situ test facility, as well as of the repository, with explicit intermediate decision points, requires further detailed investigation for each potential repository site.

SUMMARY AND CONCLUSIONS

The Task 2 results include the general recommendation of those tests which should be specifically considered in designing media/site specific in situ test programs. It has been assumed that these tests will be conducted within an in situ test facility, consisting of an exploratory shaft, extending from the surface to the prospective repository horizon possibly with test stations at various depths, and an underground test facility, consisting of appropriate tunnels and test rooms at that horizon. Plans for the program are expected to be presented in the initial SCR submittal and the complete results presented in the license application (LA). The media and sites considered include (1) basalt at Hanford, Washington; (2) tuff at Yucca Mountain, Nevada Test Site; (3) domal salt at specific Gulf Coast sites; (4) bedded salt at an unspecified site; (5) granite at an unspecified site.

A defensible rationale has been developed and utilized to tentatively select available tests to be included in the media/site specific in situ test programs. This rationale essentially consists of:

- Establishing the information needs for construction authorization developed by the time of initial SCR submittal at each site
- Assessing the relevant capabilities of available tests
- Matching the capabilities of specific tests to the perceived information needs at each site.

The information needs at any time result directly from the uncertainties in the prediction of repository system performance, and consist of the additional information needed to adequately demonstrate satisfactory performance. Information needs are determined as follows:

- Establish a licensing perspective, by:
 - developing repository system performance criteria to quantify performance objectives
 - identifying the steps during repository development at which compliance with the criteria must be demonstrated
 - establishing (either implicitly or explicitly) the acceptable level of confidence in satisfactory repository system performance at each step which constitutes adequate demonstration of compliance
- Identify the existing information and assess (either implicitly or explicitly) the associated level of confidence in satisfactory repository system performance
- Compare the assessed level of confidence with the acceptable level (either implicitly or explicitly)

7.0

- Determine what additional information, if any, is needed to raise the level of confidence to the acceptable level, by:
 - establishing the sensitivity of system performance to each component of the repository system
 - identifying where the existing information regarding significant components of the system is insufficient and can be efficiently supplemented (i.e., where the existing uncertainty is large, but can be effectively reduced).

The information needs at any time are thus a function of:

- The significance of each component of the repository system (including site characteristics) to system performance
- The existing information related to system performance
- The acceptable level of confidence in satisfactory repository system performance for each licensing step.

Performance assessment (including sensitivity studies) is outside the scope, so that the significance of repository system components has only been qualitatively evaluated and the level of confidence in satisfactory repository system performance, based on existing information, has not been assessed. The existing information for each media/site considered has been summarized herein; descriptions of tuff at NTS and domal salt at specific Gulf Coast sites were generated under Task 1 of this project, whereas descriptions of the other media/sites were derived from previous work. Also, the testing expected to precede in situ testing, and thus supplement the existing information, has been identified. The acceptable level of confidence in satisfactory performance, specifically for construction authorization, has not been explicitly established for each stage of repository development. This acceptable level has in the past, and may continue to be, established implicitly through progressive technical discussions between NRC and DOE (see Preface - Licensing Perspective). In the absence of a specified acceptable level, the information needs for construction authorization cannot be firmly established at this time. Hence, information needs for construction authorization which are perceived to develop by the time of initial SCR submittals at each media/site have been utilized to illustrate the rationale for test selection.

Tests which are available and respond to the specific information needs have been identified, and their capabilities assessed. Specific in situ tests, which might be included in an in situ test program and are especially important or unique, have been investigated in detail. Because design and specifications for each test require prior definition of information needs and a detailed description of each test location, recommendations regarding the conduct of each test and the utilization of test results are of a scoping nature only. Potential advancements to the state-of-the-art have been suggested, and areas pointed out where research and development might improve test capabilities in response to the perceived information needs.

From a comprehensive list of available tests and their relevant capabilities, in situ tests have been identified which adequately respond to the perceived information needs for construction authorization at each media/site considered (Table 5.3). These tests satisfy the information needs either by:

- Simulating various aspects of the repository for extrapolation of results
- Assessing identified media/site specific characteristics to be used in numerical modeling
- Verifying predictive numerical models.

An in situ test facility which can accommodate such reasonable in situ test programs has been developed for basalt at Hanford under Task 4 of this project and summarized here for illustration of program integration.

The actual selection of tests which should be included in a media/site specific in situ test program will be a function of the information needs and test capabilties at that time. Additional information can be expected to be obtained during ongoing or future site investigation prior to initial SCR submittal for some media/sites. Also, new or hybrid tests and modeling techniques, with improved capabilities, can be expected to be developed with time. Hence, it is expected that the in situ test program will evolve somewhat independently with time for each media/site considered as the perceived information needs and test capabilities develop. The program and the design of specific tests must be flexible enough to take into account new information which becomes available during program performance, as presented in SCR updates.

Golder Associates believes that this report presents defensible recommendations, based on currently available knowledge, regarding those in situ tests which should be specifically considered in designing a reasonable in situ test program conducted within an in situ test facility prior to construction authorization at any site. These tests adequately respond to the perceived information needs, and thus sufficiently resolve the key issues related to short-term construction/operation and long-term waste containment/isolation performance (as given in the current drafts of NRC's 10-CFR-60 and EPA's 40-CFR-191), for construction authorization at each media/site.

Although some of the assessments made within the rationale for the tentative selection of these in situ tests are necessarily subjective and the licensing perspective may not be universally shared, the rationale is clearly outlined so that specific areas of technical disagreement can be readily identified and these disagreements (if any) resolved.

Golder Associates thus recommends that the NRC should, accordingly, identify the information needs for each site and then focus on (1) the plans of the in situ test program in their review of an SCR and (2) the results of this program, and the appropriate incorporation of these results in design and performance assessment, in their review of a license application.

GLOSSARY

G.1 GENERAL TERMS

Advancements in Expanding available concepts or improving available state-of- technology.

Characteristic: Aspect of ground or environment describing repository site, and being either quantitative ("parameter") or qualitative ("factor").

Characterize: Describe the set of "characteristics," based on testing results or measurement.

Critical "Engineering variable" which has a significant im-Engineering pact on the "level of confidence in satisfactory Variable: performance."

Engineering Engineering aspect of repository design or construc-Variable: Engineering aspect of repository design or construction which can be altered by the engineer (e.g., size, shape, and orientation of underground openings).

Environmental Those conditions which may affect the assessment of a "characteristic," including stress level, pore pressure, temperature and radiation dosage.

Factor: Nonguantitative "characteristic."

Geologic Setting: Description of the geometry and boundary/field conditions of the site, as well as the physical (nonresponse) characteristics of the materials. The geometry consists of the stratigraphy/lithology and structure of rock mass units. Boundary/field conditions include the pre-excavation in situ stress field, in situ hydraulic head field and in situ temperature field. The potential changes in the geometry or boundary/field conditions unrelated to repository development (i.e., tectonics) are also part of the geologic setting.

Hydrologic Unit: Volumes which are relatively homogeneous with respect to hydrologic "response characteristics". Such hydrologic units consist of one or more "rock mass units," and will not necessarily be homogeneous with respect to physical characteristics or other "response characteristics."

(Note: all terms in quotation marks are defined separately herein.)

Information Needs: Additional information needed to improve the "level of confidence in satisfactory performance" to acceptable levels, i.e., sufficiently reduce the uncertainty in predicted repository system performance. The information needs at any time are thus a function of:

- The significance of components of the "repository system" (including site characteristics), with respect to system performance
- The existing information, which determines the "level of confidence in satisfactory performance"
- The "acceptable level of confidence in satisfactory performance".

The information needs can be responded to by:

- Appropriately simulating various aspects of the actual repository (e.g., construction techniques) for extrapolation of results
- Adequately assessing identified media/site specific "characteristics" (e.g., hydraulic conductivity) to be used in predictive numerical modeling
- Sufficiently verifying predictive numerical models.

In Situ Test Facility: Consists of an exploratory shaft (or drift), extending from the surface to the repository horizon possibly with test stations at various depths, and an underground test facility, consisting of appropriate tunnels and test rooms at that horizon. The plans for this facility will be presented in the initial SCR and the complete results presented in the license application.

In Situ Test Program: Consists of a suite of appropriate in situ tests conducted within an "in situ test facility" (after initial SCR submittal and prior to construction authorization) to adequately respond to the "information needs" and thus resolve "key issues," either by simulation or by assessing "significant characteristics" with acceptable uncertainty. Key Issue: Question, essentially related to short-term construction/operation and long-term waste containment/ isolation performance, which influences the design of "critical engineering variables" and which must be adequately resolved to demonstrate compliance with "repository performance criteria."

Level of Confidence in Satisfactory Performance: Probability that the repository system performance will comply with the "performance criteria". This probability is based on the "performance assessment."

Acceptable Level of Confidence in Satisfactory Performance: Required "level of confidence in satisfactory pertable "level of confidence in satisfactory performance" should increase during "repository development" from L_1 at the time of initial submittal of the Site Characterization Report, to L_2 at the time of construction authorization, to L_3 at the time of issuing the license to emplace waste, to L_4 at the time of decommissioning. Thus, $L_4>L_3>L_2>L_1$.

New or Hybrid Test: A test which incorporates new concepts or combines available concepts in a significant manner, as opposed to simply expanding available concepts or improving technology.

Parameter: Quantitative "characteristic."

Performance Assessment: The prediction of various aspects of performance, i.e., the determination, either explicitly or implicitly, of a probability distribution for a given performance indicator (X). For example, X might be the radionuclide release rate at some specific location.

Performance Technical requirements or performance standards Criteria: which quantify short-term construction/operation and long-term waste containment/isolation performance objectives. These criteria are given either:

> Deterministically, in which an absolute numerical limit (Xc) for each quantitative performance indicator (X) is specified, so that (if small X is good) X must be less than Xc for compliance.

Probabilistically, in which an acceptable level of confidence (Lc) that the numerical limit (Xc) will not be exceeded by the actual value of the quantitative performance indicator (X) is, specified, so that the probability of (X<Xc), o P(X<Xc), must be greater than Lc f compliance.

Repository The process of proceeding through the various "stages" of repository life from site screening, Development: selection to decommissioning.

Repository System: Consists of all components which contribute to performance, i.e., the site characteristics and engineered components.

Response "Parameters" which describe the response or behavior Characteristics: (i.e., mechanical, thermal, hydrologic and/or geochemical) of a volume of material. These "characteristics" are strongly related to the physical characteristics of the material, and also may be:

- Anisotropic (i.e., vary with orientation) Scale-dependent (i.e., vary with the scale) Time-dependent (i.e., vary over time)
- .
- A function of the present and past environmental . conditions, including
 - stress level

.

- pore pressure
- temperature
- radiation dose.

Rock Mass Unit:

Volume of rock which is relatively homogeneous with respect to its physical characteristics (i.e., intact rock, discontinuities, and pore fluid). Boundaries of rock mass units can be defined by changes in lithology, structure (i.e., large faults or changes in discontinuity patterns), or pore fluid. Due to this relative homogeneity, the "response characteristics" (or their functions) are relatively constant throughout each unit. However, each rock mass unit is only approximately homogeneous, that there is some variability in physical characteristics throughout the unit and some resulting variability in "response characteristics" as measured from point to point.

Dimensional aspects of a repository, as follows (in increasing size order):

- Waste package (very near field)
- Room (near field) .
- Repository (3 sq mi underground) ٠
- Site (10 sq mi) (far field)
- Location (30 sq mi)
- Area (1000 sq mi)
- Basin
- Region (multi-state)
- Nation (U.S.).

Significant Characteristic:

Scale:

"Characteristic" which has a significant impact on the resolution of "key issues," and thus influences design of "critical engineering variables." These "characteristics", in conjunction with the engineered components, will determine repository system performance.

- Site Characteriza-The program of exploration and research, both in the tion: laboratory and in the field, undertaken to establish the "characteristics" of a particular site. Site characterization includes borings, surface excavations, excavation of exploratory shafts, limited subsurface lateral excavations and borings, and in situ testing at depth needed to determine the suitability of the site for a geologic repository. but does not include preliminary borings and geophysical testing needed to decide whether site characterization should be undertaken.
- Site Investigation: The program of exploration, both in the laboratory and in the field (e.g., preliminary borings and geophysical testing) conducted during site screening and selection, prior to SCR submittal.

Stage:

Distinct period of activity during repository life, as follows (in chronological order):

- Site screening .
- . Site selection
- Detailed site investigation (followed by initial . submittal of SCR)
- In situ testing and repository design
- . Repository construction (preceded by license application and construction authorization)
- Repository operation (preceded by updated license application and operating license)
- .
- Waste retrieval (if required) Decommissioning (preceded by license amendment . and decommissioning authorization)
- Post-decommissioning.

G.2 COMMON ROCK MASS RESPONSE CHARACTERISTICS

Various parameters (i.e., quantifiable response characteristics) are used to describe the mechanical, thermal, thermomechanical, hydrologic, and geochemical response of a rock mass. Each of the common response characteristics discussed in this report (see Table G-1) is briefly defined in the following sections. Other less common characteristics are defined as they are used in the report.

G.2.1 MECHANICAL

Young's Modulus

Young's modulus, or modulus of elasticity, E, is defined as the ratio of the change in stress to the corresponding strain increment (in uniaxial compression or tension), for linear elastic behavior of the material:

$$E = \frac{\Delta \sigma x}{\Delta \epsilon x} (ML^{-1} T^{-2})$$

where

 $\Delta \sigma x = \text{stress change } (ML^{-1} T^{-2})$ $\Delta \epsilon x = \text{strain change } (\text{dimensionless})$

Young's modulus, in conjunction with one other elastic parameter (typically Poisson's ratio), describes the elastic behavior of an isotropic material.

Deformations within rock masses, however, typically consist of a combination of elastic and inelastic behavior, and may be nonlinear. The term "deformation modulus" (analogous to Young's modulus) is often used to describe the elastic/inelastic, as well as nonlinear, behavior of rock masses.

These moduli are often anisotropic and scale dependent. Their values often vary with stress history, stress level, strain rate, temperature and possibly radiation dosage.

Poisson's Ratio

Poisson's ratio, ν , is defined as the negative ratio of lateral strain to axial strain produced by an applied axial load, for elastic behavior of the material:

 $\nu = \frac{\Delta \epsilon_L}{\Delta \epsilon_A}$ (dimensionless)

where

 $\Delta \epsilon_{L}$ = lateral strain (dimensionless) $\Delta \epsilon_{A}$ = axial strain (dimensionless)

Poisson's ratio, in conjunction with one other elastic parameter (typically Young's modulus), describes the elastic behavior of an
CHARACTE	RISTICS			Table G-1
	RESPONSE CHARACTERISTIC	COMMON SYMBOL	DIMENSIONS	COMMON UNITS
Mechanical	Young's Modulus	F	MI -1 -2	psi. MPa
	Poisson's Ratio	2	(dimensionless)	-
	Creen	- <u>-</u>	T-1	hr ⁻¹
	Strength	_	ML -1 -2	psi, MPa
	Joint Normal Stiffness	K,	ML-2T-2	MPa/mm
	Joint Shear Stiffness	Ks	ML-2T-2	MPa/mm
Thermal				
	Thermal Conductivity	к	MLT ⁻³ Te ⁻¹	Watt/meter ^O C
	Mass Heat Capacity	C _m	L ² T ⁻² Te ⁻¹	Calorie/gm ^O C
	Volumetric Heat Capacity	C,	'ML ⁻¹ T ⁻² Te ⁻¹	Calorie/cm ^{3 o} C
	Specific Heat	C_	(dimensionless)	-
	Thermal Diffusivity	ĸ	L ² T ⁻¹	cm ² /sec
Thermo-Mechanical				
nermo-nechanica i	Coefficient of Thermal			
	Expansion: linear	a	Te ⁻¹	°F ⁻¹ , °C ⁻¹
	volumetric	ß	Te ⁻¹	°F-1, °C-1
ydrologic				
	Hydraulic Conductivity	к	LT-1	fps, cm/sec
	Intrinsic Permeability	k	L ²	cm ²
	Specific Storage	S.	L ⁻¹	cm ⁻¹
	Effective	, n	(dimensionless)	

COMMON R CHARACTE	ROCK MASS RESP	ONSE		Table G-1 2 of 2
	RESPONSE CHARACTERISTIC	COMMON SYMBOL	DIMENSIONS	COMMON UNITS
Geochemical			1.19	
	Dispersivity	a		cm
	Equilibrium Distribution Coefficient	к _d	L ³ M ⁻¹	m1/gm
	Sorption Ratio	Rd	L ³ M ⁻¹	m1/gm
	Retardation Factor	R	(dimensionles	ss) -

Key:	М	=	mass
	L	=	length
	Т	=	time
	Te	=	temperature

Rev. Dwg. No. 213 46 46 42 42 Date / 22 Eng. 44

isotropic material. Poisson's ratio is also often used to describe the typical elastic/inelastic, as well as non-linear, behavior of rock masses.

Poisson's ratio may be anisotropic and scale dependent. In addition, the value may vary with stress level, temperature, strain rate, and possibly radiation dosage.

Creep Parameters

Creep parameters express the viscous behavior of a material i.e., the strain which occurs with time at constant stress or the stress relaxation which occurs with time and no deformation. There is presently no well-accepted formulation for creep behavior in all media. However, the following characteristics are sometimes exhibited under constant stress conditions (see Figure G-1):

- Primary creep, in which initial creep strain occurs at a decreasing rate with time, and unloading produces complete strain recovery
- Secondary creep, in which the creep strain approaches a steady state (i.e., constant creep rate) and unloading does not result in complete strain recovery (i.e., permanent set); secondary creep is exhibited only at relatively high stress levels
- Tertiary creep, in which the creep strains become very large and the rate of strain increases to rupture; tertiary creep occurs only at stress levels approaching the strength of the material.

Creep behavior is often anisotropic and scale dependent. In addition, the behavior is often a function of stress level, temperature, and possibly radiation dosage.

Strength

The strength of a material expresses the limiting state of stress (ML-1 T-2) at which failure occurs. In this sense, failure consists of fracturing or sliding along existing discontinuities in brittle material, or of large deformations in ductile material. In brittle material, peak and residual strengths may be defined, where peak strength is the maximum stress state (at which failure occurs) and residual strength is the post-peak steady-state (i.e., frictional) stress state (see Figure G-2).

A strength criterion for a material can be defined as a unique combination of stress parameters which define the boundary (surface) between stable and unstable conditions. Strength is termed tensile strength, compressive strength, or shear strength depending on the loading and constraint conditions.



W.W. Revision #13-11638 46001

28-1



Strength is often anisotropic and scale dependent, especially in discontinuous media. In addition, due to its partly frictional nature, strength is strongly a function of confining stress, as well as temperature, stress history, strain rate, pore pressure, and possibly radiation dosage.

Joint Normal Stiffness

The joint normal stiffness, K_n , is defined as the normal displacement across the joint caused by a unit change in stress normal to the joint. The joint normal stiffness is a function of the stress level, stress history, strain rate and temperature.

Joint Shear Stiffness

The joint shear stiffness, K_s , is defined as the relative shear displacement along the joint caused by a unit change in shear stress parallel to the joint. The joint shear stiffness is a function of the stress level, stress history, strain rate and temperature.

G.2.2 THERMAL

Thermal Conductivity

The coefficient of thermal conductivity, K, is defined as the rate of heat transferred by conduction per unit area normal to the direction of heat flow per unit thermal gradient in the direction of heat flow, under steady-state conditions:

$$\kappa = \frac{Q}{At} \left(\frac{\delta T}{\delta x}\right)^{-1} (MLT^{-3} Te^{-1})$$

where

Q = amount of heat energy (ML² T⁻²) A = cross-sectional area normal to flow (L²) t = time of flow (T) $\frac{\delta T}{\delta x}$ = thermal gradient, or change in temperature per un t δx length in the x-direction (TeL⁻¹)

Thermal conductivity may be anisotropic and scale dependent. It may vary with temperature, state of stress, and pore fluid composition.

Mass Heat Capacity

The mass heat capacity, C_m , is defined as the amount of heat energy necessary to change the temperature of a unit mass of material by one degree:

$$C_{m} = \frac{Q}{m \Delta T} (L^{2} T^{-2} Te^{-1})$$
$$= \frac{C_{v}}{\rho_{d}}$$

where

Q = amount of heat energy (ML² T⁻²) m = mass of material (M) $\triangle T$ = change in temperature (T_e) δ_d = dry density of material (ML⁻³) C_v = volumetric heat capacity (ML⁻¹ T⁻² Te⁻¹)

The mass heat capacity, in conjunction with thermal conductivity, is needed to describe transient heat flow.

The mass heat capacity may be scale dependent. It may vary with temperature, state of stress, and pore fluid composition.

Volumetric Heat Capacity

The volumetric heat capacity, C_V , is defined as the amount of heat energy necessary to change the temperature of a unit volume of material by one degree:

$$C_v = \frac{Q}{V\Delta T} (ML^{-1} T^{-2} Te^{-1})$$
$$= C_m P_d$$

where

Q = amount of heat energy (ML² T⁻²) V = volume of material (L³) ΔT = change in temperature (T_e) P_d = dry density of material (ML⁻³) C_m = mass heat capacity (L² T⁻² Te⁻¹)

The volumetric heat capacity, in conjunction with thermal conductivity, can be used to express transient heat flow.

The volumetric heat capacity may be scale dependent. It may vary with temperature, state of stress, and pore fluid composition.

Specific Heat

Specific heat, C_p, expresses the ratio of the mass heat capacity of a material to that of water. The specific heat, in conjunction with thermal conductivity, can be used to express transient heat flow.

Specific heat may be scale dependent. It may vary with temperature, state of stress, and pore fluid composition.

Thermal Diffusivity

Thermal diffusivity, κ , expresses the rate at which a change in temperature spreads by conduction for one-dimensional transient heating.

It is related to thermal conductivity and volumetric heat capacity as follows:

$$\kappa_{x} = \frac{\kappa_{x}}{C_{y}} (L^{2} T^{-1})$$

where

 K_x = thermal conductivity in the x-direction (MLT-3 Te-1) C_y = volumetric heat capacity (ML-1 T-2 Te-1)

Thermal diffusivity may be anisotropic and scale dependent. It may vary with temperature, state of stress, and pore fluid composition.

G.2.3 THERMOMECHANICAL

Coefficient of Thermal Expansion

The coefficient of thermal expansion defines the change in length (linear coefficient), area (surficial coefficient), or volume (volumetric coefficient) of a body per unit length, area or volume, for a given change in temperature ΔT :

$$a = \frac{\Delta L}{L\Delta T}$$
 (Te⁻¹)

where α = coefficient of linear thermal expansion,

and

$$\beta = \frac{\Delta V}{V \Delta T}$$
 (Te⁻¹)

where β = coefficient of volumetric thermal expansion.

In isotropic materials the coefficient of volumetric thermal expansion is approximately three times the coefficient of linear thermal expansion:

B=3a

The coefficient of linear thermal expansion may be anisotropic and scale dependent. It may vary with temperature, stress level, pore fluid composition, and possibly radiation dosage.

G.2.4 HYDROLOGIC

Hydraulic Conductivity

Hydraulic conductivity, K, is defined by Darcy's Law as the proportionality constant relating the Darcy velocity of a fluid (volume flux rate) to the hydraulic gradient:

$$\kappa_i = q_i \left(\frac{\delta h}{\delta x_i}\right)' (LT^{-1})$$

where

i = 1,2,3 refers to the principal coordinate axes $q_i = Darcy velocity$ in the i direction (LT-1) $\underline{\delta n} = hydraulic gradient, or rate of change in hydraulic head$ $<math>\overline{\delta x_i}$ per unit length, in the i direction (dimensionless) $K_i = principal hydraulic conductivity in the i direction$ (LT-1)

The hydraulic head, h, at any point (x, y, z) can be expressed as:

h (x,y,z) = E(z) +
$$\frac{p(x,y,z)}{\rho g}$$
 (L)

where

E = elevation above an arbitrary datum (L)

 $p = pore pressure (ML^{-1} T^{-2})$

g = acceleration of gravity (LT-2)

 ρ = pore fluid density (ML-3).

Hydraulic conductivity may also be defined in terms of the properties of the porous medium and the pore fluid:

$$\kappa = \frac{k \rho_g}{\mu} (LT^{-1})$$

where

k = intrinsic permeability of the porous medium (L²) μ = pore fluid dynamic viscosity (ML-1 T-1)

The pore fluid density and viscosity depend on the pore fluid composition and vary with temperature.

Hydraulic conductivity may be anisotropic and scale dependent. It may vary with temperature, state of stress and pore fluid composition.

Intrinsic Permeability

Intrinsic permeabiliy, k, is a characteristic property of the porous medium. It is related to the hydraulic conductivity by the pore fluid properties:

$$k = \frac{\kappa \mu}{\rho g} (L^2)$$

where

K = hydraulic conductivity (LT-1) μ = pore fluid dynamic viscosity (ML-1 T-1) P = pore fluid density (ML-3) g = acceleration of gravity (LT-2)

Intrinsic permeability may be anisotropic and scale dependent, and may vary with temperature and stress level.

Specific Storage

The specific storage, Ss, is the volume of fluid taken into or released from storage in a unit volume of the porous medium per unit change in hydraulic head under saturated conditions. It depends on the compressibility of the rock matrix and the pore fluid. For porous media with high permeability, the relationship can be expressed as:

 $S_{c} = (nB+a)\rho q \quad (L^{-1})$

where

- n = total porosity, or the ratio of the volume of void space in the rock to the total volume (dimensionless)
- B = compressibility of the pore fluid ($LT^2 M^{-1}$)
- a = compressibility of the rock matrix, or the change in volume of the rock matrix per unit volume of the porous medium (LT² M⁻¹)
- $P = \text{pore fluid density}(ML^{-3})$
- g = acceleration of gravity (LT-2)

The applicability of this equation to media with very low hydraulic conductivity is uncertain.

Specific storage may be scale dependent and may vary with temperature, stress level, and pore fluid composition.

Effective Porosity

Effective porosity, ne, is defined as the ratio of void space through which fluid moves (i.e., interconnected voids) to the total volume. It is thus approximately equal to the ratio of the Darcy velocity to the true average pore fluid velocity:



(dimensionless)

where

q = Darcy velocity in the x direction (LT-1)

V = true average pore fluid velocity (LT-1)

K = hydraulic conductivity in the x direction (LT-1)

δh = hydraulic gradient in the x direction (dimensionless)

8x

The effective porosity is less than or equal to the total porosity. In fractured media, effective porosity may be much less than the total porosity because the majority of flow may occur through a relatively small number of fractures.

G.2.5 GEOCHEMICAL

Dispersivity

Dispersion describes the spreading of a solute when introduced into a flow field and includes both mechanical dispersion and molecular diffusion. Mechanical dispersion results from the movement of fluid along statistically random paths through the porous medium, while molecular diffusion results from physiochemical properties of the fluid and the surrounding rock. Molecular diffusion is normally neglected in natural groundwater systems and has not been measured in in situ tests.

Dispersivity, a (the measure of dispersion), is a length property of the medium and is strongly dependent on the scale of the flow region under investigation. Values can range from 10^{-2} cm in laboratory tests to 10^4 cm for regional systems. At the laboratory scale it is a consequence of the tortuosity of the medium pore space, velocity gradients in the pore spaces, and variations in the pore space dimensions. At the regional scale it is primarily due to the divergence of flow paths resulting from heterogeneities in aquifer properties, particularly hydraulic conductivity.

Dispersivity is anisotropic and is described by a lateral dispersion coefficient (perpendicular to the flow direction) and a longitudinal dispersion coefficient (in the direction of flow). The longitudinal dispersion coefficient is typically 5 to 20 times as large as the lateral dispersion coefficient.

Equilibrium Distribution Coefficient

For low to moderate solute concentrations, the mass of adsorbed solute per unit dry bulk mass of the medium and the mass of dissolved solute per unit volume of pore fluid are commonly related by the expression:

 $S = K_d C^b$ (dimensionless)

where

- S = mass of adsorbed solute per unit dry bulk mass of the porous medium (dimensionless)
- C = solute concentration, or mass of dissolved solute per unit volume of pore fluid (ML^{-3})

 K_d and b are determined experimentally and depend on the solute species and the geochemical character of the system. This relationship is known as the Freundlich isotherm. If b=1, then

$$K_{d} = S/C (L^{3} M^{-1})$$

is a linear isotherm and $K_{\rm d}$ is known as the equilibrium distribution coefficient.

Equilibrium distribution coefficients are normally measured in the laboratory by batch or column tests. In these tests, a contaminant solution of known concentration is mixed with powdered rock (batch test) or circulated through a column of crushed rock (column test). When chemical equilibrium is achieved, the change in dissolved solute concentration is used to calculate the amount of solute adsorbed by the rock. In many tests the reaction rates are so slow that equilibrium conditions cannot be verified at the end of the test. In these cases, the calculated ratio S/C is called the sorption ratio, R . Measured sorption ratios are thus less than or equal to the equilibrium distribution coefficient.

Sorption Ratio

The sorption ratio, R_d , is the mass of adsorbed solute per unit dry bulk mass of the porous medium divided by the solute concentration of the pore fluid:

$$R_{d} = S/C (L^3 M^{-1})$$

where

- S = mass of adsorbed solute per unit dry bulk mass of the porous medium (dimensionless)
- C = solute concentration, or mass of dissolved solute per unit volume of pore fluid (ML^{-3})

For solutes which have linear adsorption isotherms, the sorption ratio under equilibrium conditions is equivalent to the equilibrium distribution coefficient.

Retardation Factor

The retardation of the front of a reactive solute relative to the bulk mass of groundwater is described by the retardation factor R:

$$R = I + \frac{\rho_b}{n} \kappa_d = \frac{\bar{v}}{\hat{v}_c}$$
 (dimensionless)

where

 $\begin{array}{l} {\mathcal{P}}_b \ = \ bulk \ density \ of \ the \ porous \ medium \ (ML^{-3}) \\ n \ = \ total \ porosity \ (dimensionless) \\ {\mathcal{K}}_d \ = \ equilibrium \ distribution \ coefficient \ (L^3 \ M^{-1}) \\ \overline{v} \ = \ average \ linear \ velocity \ of \ the \ groundwater \ (LT^{-1}) \\ \overline{v}_c \ = \ velocity \ of \ the \ (C/Co \ = \ 0.5) \ point \ on \ the \ concentration \ profile \ of \ the \ reactive \ solute \ (LT^{-1}) \end{array}$

This relationship holds for fast reversible adsorption with linear isotherm ($S = K_dC$).

G.3 COMMON INSTRUMENTATION USED FOR IN SITU TESTING

In situ tests generally require the measurement of some of the following responses:

- Strain and displacement
- Stress (absolute and changes)
- Temperature
- Groundwater pressure

These responses are typically measured at discrete points by instrumentation and interpolations or extrapolations made to determine the complete response field. Many of the in situ tests described in the Appendix utilize the same instrumentation. The instrumentation that is common to several tests is described in this section of the Glossary. Two references have been used extensively for this section of the Glossary: Cording et al (1975) and IECO (1979).

G.3.1 STRAIN AND DISPLACEMENT

Strain or displacement in the rock mass can be measured by a variety of methods (see Table G-2), either within boreholes or on the exposed surface. Although the results are usually referenced to a discrete point in the rock mass, measurement of strain and displacement occurs over a finite length. Borehole extensometers measure displacements that are parallel to the axis of the borehole (i.e., longitudinal measurements), whereas deflectometers and inclinometers measure displacements that are transverse to the axis of the borehole. Strain gauges, which measure strain based on changes in electrical resistance, frequency of vibration, or electrical inductance, are often used as components in other instrumentation systems (including extensometers, inclinometers, deflectometers, load cells, stress cells, and piezometers).

Borehole Extensometers

Borehole extensometers consist of one or more anchors at various depths in the borehole and a reference head at the borehole collar (see Figure G-3). Units with a single anchor are called Single Position Borehole Extensometers or SPBX's; those with two anchors are called Double Position Extensometers, or DPBXs; those with more than two anchors are termed Multiple Position Extensometers, or MPBX's. The anchors in the borehole are connected to the reference head by steel rods or tubes in the rod type (see Figure G-3a), and by tensioned wires in the wire type (see Figure G-3b). A deformation sensor, such as a micrometer or strain gauge, is inserted between the ends of the rods or wires and the reference head so that relative movements between the anchors and reference head can be measured. Typical accuracies for rod units are of the order of +0.002 in. over lengths of up to 100 ft. Relative displacements between anchors can be obtained since all the displacements are related to a common reference. If the bottom anchor is positioned outside the zone of movement, then that anchor can be considered fixed and absolute displacements of each anchor can be assessed.

						EST (1982)	U.S. DOL	LARS)
INSTRUMENT	RANGE	ACCURACY	ADVANTAGES	LIMITATIONS AND PRECAUTIONS	REL TABILITY	ACQUI- SITION	IN- STALL- ATION	CALT- BRATION
LONGITUDINAL MOVEME	INT IN BOREH	OLES						
(BOREHOLE EXTENSOME	ETERS)							
Rod type ex- tensometer		* 0.001 to * 0.005 inches	Simple and precise. Multiple anchors possible. Can also accept remote read- out transducers.		Good	\$10,000*	\$560	\$140
Variable ten- sion wire type extenso- meter	0.6 in. up to 3.5 in. Some can be reset	+ 0.002 to + 0.005 Inches	Multiple anchors - up to 6 or 8. Some models designed for remote readings with transducers using bonded resistance or vibrating-wire strain gauges.	Variable tension requires varying calibration factors. Wire friction and hysteresis can seriously affect accu- racy. Risk of electrical failure.	Fair - Short term	\$6,000*	\$560	\$140
Constant tension wire type exten- someter	2 in, can be reset	+ 0.002 to + 0.005 Taches	Multiple anchors. Designed for re- mote readings using potentio- meters. Constant calibration factor.	Wire friction and hysteresis can seriously affect accuracy. Risk of electri- cal failure. Complex mechanically.	Fair - Short term	\$6,000*	\$ 560	\$ 140
TRANVERSE MOVEMENT	IN BOREHOLE	s						
(INCLINOMETERS)								
Fixed multi- point bore- hole inclino- meters		+ 0.001 Tn. per 10 feet	Precise, re- movable for repairs or re- use. Uses standard in- clinometer casing.	Complex, does not measure continuous profile. Accuracy lost when removed and replaced.		\$13,000	\$840	\$210
Portable bore-	+ 12 ⁰	+ 1 in.	Long experience	Lengthy calculations. Reads one axis at a time.	Good	\$2,000	\$ 840	\$ 280

Rev. Dwg. No. 113-1638 40018 Date 182 Eng. wh

						ESTIMATED COST (1982 U.S. DOLLARS)	
INSTRUMENT	RANGE	ACCURACY	ADVANTAGES	LIMITATIONS AND PRECAUTIONS	RELIABILITY	ACQUI- INSTALL- CALI- SITION* ALLUN BRATION	
(DEFLECTOMETERS)						
Fixed multi- point borehole deflectometers		* 20 arc sec.(To- tal dis- placement accuracy depends on pivot spacing).	Portable version available and double pivot ver- sion available to measure movement along two axes.	Complex, does not measure continuous profile.	Good	\$13,000 \$840 \$210	
TRANSVERSE AND	LONGITUDINAL MC	DVEMENT IN BO	DREHOLES				
Iseth Extnso - Deflectometer	13 mm/m 17 mm/m	± 0.01 mm/ m-longi- tudinal ± 0.03 mm/ m-trans- verse	Rapid use. Precise.	Limited experience. Special casing needed. May not have enough range in areas of high strain. No remote readout possible.		\$30,000- \$1,400 \$700 \$40,000	
TUNNEL CONVERGEN	NCE						
(PORTABLE EXTENS	OMETERS)						
Portable rod extensometer	3 to 25 feet	+ 0.001 to + 0.01 inches	Simple, precise, portable.	Limited span, accuracy limited by sag. Invar tubes can be used to minimize temperature corrections.	Excellent	\$2,000 \$35** \$10	

	STRAIN SENSITIVITY,	GAGE LENGTH.	TYPICAL RANGE,		LIMITATIONS		(198 ACOUL	TIMATE 32 U.S.	D COSTS DOLLARS)
INSTRUMENT	MICROSTRAINS	INCHES	MICROSTRAINS	ADVANTAGE S	PRECAUTIONS	RELIABILITY	SITIO	N ATION	BRATION
SMALL SCALE STRAIN/DIS	PLACEMENT								
(RESISTANCE STRAIN GAUG	ES)								
Bonded electrical re- sistance gauge	2-4	.008-6	20,000 - 50,000	Small size, low cost. Tem- perature compensation available.	Errors due to lead wire and circuit resistance changes unless compensated. Long term stability may be poor due to cement creep. Meti- culous installation proce- dure. Difficult to water- proof.	Poor - Fair	\$5	\$70	\$35
Encapsulated, unbonded, electrical resistance (Carlson gauge)	•	8 - 20	700 ten- sion, 1400 com-	Factory waterproofing. easy to install. Long experience record.	Errors due to lead wire and circuit resistance changes unless compensated. Small range Temperature correc-	Good			
/VIDDATING_UIDE CTDAIN	CAUCE								
Vibrating-wire Sikain	1.2	4 . 14	600	Not affected by land	(mall manness Temperature	Cond	435	\$20	410
Toracting are gauge	1-6		7,000	wire resistance changes. Easy to install, factory waterproofing. Long ex- perience record. Robust, reusable. Good long- term stability.	correction required.	6000	133	\$10	10
(LINEAR DISPLACEMENT TR	ANSDUCERS)								
Linear Potentiometers	0.25-0.025		1 - 60 cm	Easy to install. Direct remote readout.	Difficult to seal against moisture. Low sensitivity.		\$50	\$70	\$35
DCDT/LVDT	10		1 ~ 60 cm	Little friction between components. Easy to in- stall. Direct remote readout.			\$50	\$70	\$35
(MECHANICAL STRAIN GAUGE	5)								
Mechanical Gauge	5-10	2-80	10,000 - 50,000	Simple, low cost, water- proofing not required. Removeable.	Requires skill in reading. Can not be read remotely.	Excellent	\$450	\$35	135

Rev.

134

Dwg. No. 3/1-/1035 14021 Date / 82 Eng 24



Rod or wire extensometers are available that can use up to 8 anchor positions in a 76 mm diameter or smaller borehole. The deformation sensors can be mechanical (micrometer or dial gauge) or electrical (DCDT, linear potentiometer, vibrating-wire strain gauge, sonic pulse, or resistance strain gauge). The mechanical readouts are usually removed from the reference head between readings.

The type of anchor used depends on the characteristics of the material being monitored. Grouted anchors, using expansive grout or polyester resin, are the most reliable and are particularly useful in weak rock or rock that tends to creep (see Figure G-4). Mechanical expansion shell anchors or rock bolt anchors can be used in most sound rock. Spring-loaded wedge types and a hydraulically activated prong-type are also available. If any doubt exists on the ability of a particular anchor type to function adequately in a given material, lab or field tests should be conducted to test for slippage and adequate anchor strength.

In wire-type units, the wires are usually encased in a flexible, oilfilled, PVC tube for protection. At the level of each anchor, a special joint is placed and one wire is tied off to a washer at that point. The other wires pass through separate guide holes in the washer and extend to the deeper anchor points. The area between the tubing and borehole wall is fully grouted, thus anchoring the unit. The tubing must be strong enough to withstand the grout pressures, otherwise it may collapse and jam the extensometer rods or wires. The grout is relatively weak and moves with the rock, thereby displacing the anchor assembly. For the rod-type units, the rods are usually protected by individual PVC tubes down to the anchor position where a portion of one of the rods (or an extension consisting of a rebar or grooved anchor to improve bonding to the grout) is exposed. The borehole is then fully grouted.

The wire extensometers may be used in boreholes that are less than 6 to 9 m deep. The friction between wires may result in large error in boreholes deeper than about 9 m. Rod extensometers are superior for deep boreholes because friction introduces much less error and the reliability of the anchor can be more easily tested.

Changes in temperature can produce significant errors in both rod and wire-type extensometers. If significant temperature changes are anticipated during the test, it may be necessary to install temperature sensors along the length of the extensometer so that temperature corrections can be applied. The use of super invar rods will reduce, but not eliminate, error due to temperature changes. Shown below are some coefficients of thermal expansion for some of the materials that are typically used in borehole extensometers:

Material	Coefficient of Thermal Expansi (in microstrains/°C)					
Stainless steel	17.8					
Mild steel	11.8					
Aluminum	23.1					
Super invar	0.36 to 1.0					



A 30 m long rod that is heated by 1°C will increase in length by 0.5 mm if it is stainless steel and by 0.03 mm if it is super invar.

Installation of a borehole extensometer involves drilling a borehole, determining anchor locations, assembling and testing the instrument, inserting it into the borehole, setting the anchors, and possibly grouting.

Borehole extensometers may require holes that are 40 to 90 mm in diameter, but many are designed for a 76 mm diameter hole because this is a common size used in geotechnical engineering. The hole is cored so that rock quality information may be obtained. Deep holes should be surveyed so that the drift of the hole can be included in analysis of results.

The anchor location is based on the anticipated rock movements and the quality of the core recovered from the hole. Some anchors are located in the areas where the greatest movements are anticipated and others are spaced out along the length of the borehole. Anchors should be located so that they will bracket zones of discontinuities and should not be located directly across a discontinuity. Once positioned, the anchors are set. If grouted anchors are used, neat cement grout is pumped into the grout tube so that it will fill the borehole from the bottom. If possible, horizontal holes are generally inclined downward a few degrees to facilitate grouting.

Assembly of the extensometer includes the following:

- Cut wire, rods, conduit, and grout tubes to length
- Straighten the wire or rods that will be used
- Thread wires (if used) into conduit
- Attach anchors to conduit and wires
- Fill conduits with oil.

The extensometer is then inserted into the borehole.

For wire extensometers, it may be necessary to use a mechanical or hydraulic anchor so that the wires can be tensioned before grouting the hole. For rod extensometers, the rods are installed in the conduit and the mechanical readout head is installed after the grout has set.

A mechanical readout rod extensometer requires little calibration. It simply requires an initial reading to be taken using a depth micrometer or dial gauge. The wire-type extensometer requires cycling of the tension with measurements taken after each cycle in order to measure the repeatability of the measurements. Electronic extensometers require a series of readings checked against the values obtained from the mechanical readout to determine the amount of zero drift.

Inclinometers

Displacement transverse to the borehole axis can be determined using a borehole inclinometer (see Figure G-5). This instrument measures the



-

.

inclination of the casing emplaced within a borehole. Lateral displacement is detected by changes in the inclination and can be determined by integrating the changes in inclination over the length of the casing. The bottom of the inclinometer is usually placed deep enough below the zone of movement so that it can be regarded as fixed. Where large displacement occurs along discontinuities, the special casing may shear off. Hence, borehole inclinometers are generally applicable to soil or soft rock conditions.

Borehole inclinometers may be either permanently mounted or portable. Both types use the same casing. Fixed inclinometers have an accuracy of about 0.1 mm in 10 m, but this accuracy is lost if the instrument is removed from the hole for repair. Portable inclinometers generally have an accuracy of about 3 mm over a distance of 10 m, which may not be accurate enough for measuring the typically small displacements that occur in most rock types.

The inclinometer casing is installed in a borehole and grouted in place. Casing is available with outside diameters of 48, 70, and 85 mm. The borehole is sized for the couplings, which are 54, 78, and 94 mm in diameter. The borehole is cored (usually NX size, 76 mm diameter) and then reamed out to the size required for the couplings. The casing is installed with a grout pipe along the side and then grouted into place using neat cement grout. When the grout has set, the casing is surveyed to determine the amount of spiraling.

There is little calibration required except for taking of initial readings. The inclinometer probe is generally calibrated in a casing that is set in an area that will experience no movement. For the portable inclinometer, a probe is lowered down the hole and readings are taken at 30 to 60 cm intervals along the casing. A check on the accuracy is provided by turning the probe 180 degrees and repeating the process. The readings are later reduced manually or by computer to determine the slope and displacement along the casing.

The form of readout used depends on the type of sensing mechanism in the probe. Various probes are available which use Wheatstone bridge or vibrating wire readouts.

Deflectometer

A deflectometer consists of steel tubes that are connected by pivots installed in a borehole (see Figure G-6). Deflections of the pivots are detected by strain gauges that are mounted on cantilevers. The transverse movement along the borehole is calculated by summing the deflections along the length of the borehole. Up to 8 pivots may be used per borehole, at a maximum spacing of 6 m between pivots. The pivots are usually single axis, but dual axis models are also available. The deflectometer can be installed in 76 mm I.D. pipe that is grouted into the borehole. Portable deflectometers are also available that consist of two tube sections and a pivot that are designed to be inserted and withdrawn in standard inclinometer casings.



Deflectometer pipe is installed in boreholes in the same way as inclinometer casing. The deflectometer requires 76 mm I.D. pipe, so it is necessary to install the pipe in a borehole that is about 95 mm in diameter. The pipe is grouted in place using a grout tube attached to the side of the pipe.

The deflectometer is installed in the pipe and the anchors are locked into place. For rock, the anchor spacing is generally based on the condition of the rock core that is recovered from the borehole and the location of the zone of expected greatest movement. The lowest anchor is usually set in an area where negligible movement is anticipated so that it can be assumed to be fixed.

The accuracy of the deflectometer is dependent on the spacing and number of pivots. The individual pivots are accurate to within about 30 seconds of arc (1 mm in 10 m).

The deflectometer is read using a standard Wheatstone bridge type readout unit.

Resistance Strain Gauges

The electrical resistance of a wire is inversely proportional to its cross-sectional area. Straining of the wire changes its cross-sectional area and, consequently, its resistance. Thus, if the wire is attached to a surface, measurements of the changes in the wire's resistance can be used to determine the strains in the surface in the direction of the wire. This is the operating principle of the resistance strain gauge (see Figure G-7).

The most common form of the resistance strain gauge is the bonded gauge. In the bonded strain gauge, a thin wire filament or metal foil is formed into a pattern and bonded to a backing of paper-thin plastic or epoxy (see Figure G-7a). The backing is in turn cemented to the surface where the measurements are to be made. These gauges are available in a wide variety of sizes (gauge lengths from as small as 0.008 in. to as large as 6 in.) and shapes (including multi-element rosette configurations). Their strain sensitivity is usually around 2 to 4 microstrain with ranges up to 20,000 to 50,000 microstrain (2 to 5 percent strain) for normal gauges. Post-yield or high elongation gauges that work up to strains of 10 to 20 percent are available. Bonded strain gauges are primarily used for surface strain measurements.

Bonded gauges are difficult to use under most field conditions because they require meticulous care and skill in preparing the measurement surface and cementing the gauge in place. They are very sensitive to moisture and good waterproofing is difficult to achieve under field conditions. Bonded resistance strain gauges can be mounted and sealed in the factory into a stainless steel or brass envelope (see Figure G-7b) that is then welded to the measurement surface or embedded into concrete. This gauge is termed an encapsulated gauge. One version of



the weldable encapsulated gauge uses a fine wire supported in an insulating medium and enclosed in a sealed stainless steel tube. To avoid heat damage, a small capacitive discharge welder is used to attach these gauges.

Another form of the resistance strain gauge is the <u>unbonded</u>, <u>encapsula-</u> ted gauge (see Figure G-7c). This gauge uses a fine wire strung under tension over ceramic insulators mounted on a flexible metal frame to form a resistance coil. Usually two coils are used and they are arranged so that one coil contracts while the other expands when the frame is strained. The coils and frame are factory sealed into a tubular metal cover. The gauge is mounted by bolting it to saddle brackets previously attached to the measurement surface. One version of this type of strain gauge, the Carlson strain meter, has a very long and successful experience record in field usage, e.g., Carlson meters embedded in concrete dams have worked successfully for periods of over 20 years. The Carlson system is also used in stress cells and piezometers.

Wheatstone bridge type readouts are used for resistance-type strain gauges. Correction must be made for the temperature of the gauge and the resistance of the lead wires.

Vibrating-Wire Strain Gauges

The vibrating-wire strain gauge consists of a length of steel wire that is stretched between two posts (see Figure G-8). The two posts are connected to the points between which the displacement is to be measured. Displacement between the two posts causes a change in the tension in the steel wire, which is sensed by determining the change in natural frequency of vibration of the wire. In order to allow for remote readout, the vibrating-wire strain gauge has a magnet, an exciter coil, and a sensor coil. The exciter coil is used to "pluck" the wire and set it vibrating. As the wire vibrates in the magnetic field, it induces an electrical current in the sensor coil which has a frequency equal to the frequency of vibration. The current is amplified by the readout unit and the frequency is measured by a frequency counter contained in the readout. Automatic electronic conversion of changes in frequency to strain are available.

The advantage of the vibrating-wire strain gauge is that the frequency that is measured is independent of the resistance of the leads and the currents that may be induced in the lead wires. The disadvantage is that they have a relatively small range, because the wire either may be stretched too tight or may become too slack if strain is excessive. During manufacture or during installation, the initial tension in the wire should be set so that 3/4 of its range is available in the expected direction of movement and 1/4 is available in the opposite direction.

The strain gauge is generally installed by attaching the ends onto the surface that is to be monitored. Embedment-type gauges are cast in concrete or grouted in boreholes. Calibration consists of measuring the initial frequency and temperature after installation is completed.



Subsequent readings include measuring frequency and temperature. A thermistor may be mounted to the gauge for measuring temperature, or an additional circuit may be installed in the readout to measure the resistance of the exciter coil so that temperature may be calculated. A strain gauge which is physically isolated from structural strains can also be used to correct for temperature effects and to monitor temperatures.

The vibrating wire strain gauge is accurate to within about 5 to 10 percent. Although other strain gauges provide greater accuracy, the vibrating wire strain gauge has been the most successful for use in long-term monitoring of underground excavations where remote readout is necessary. This is because the vibrating wire strain gauge is completely waterproof and is not plagued by the electrical problems of the other types of strain gauges. Although zero drift can be a problem with vibrating wire strain gauges due to creep of the wire, this can be compensated by using dummy gauges.

Linear Displacement Transducers

Two types of linear displacement measuring transducers, the linear potentiometer and the linear variable differential transformer or LVDT (see Figure G-9), are used extensively in many geotechnical instruments, such as extensometers. The LVDT is sometimes referred to as a DC to DC linear differential transformer or DCDT. The LVDT is sensitive enough that it can be used as a strain gauge.

The linear potentiometer is a resistance device. It consists of a mandrel wound with fine wire or conductive film. A wiper attached to a shaft rides along the mandrel and divides the mandrel resistance into two parts. The resistance ratio of these parts is measured with a Wheatstone bridge circuit to determine the displacement of the wiper and shaft. Linear potentiometers are available with ranges from 0.5 to 4 in. Average sensitivity is 0.01 in. to 0.001 in. Since linear potentiometers are a combined mechanical and resistance device, moisture can cause severe problems and waterproofing of potentiometers can be difficult. Usually the device is sealed into a case with a mechanical seal, such as an 0-ring, at the shaft.

The LVDT converts a displacement into a voltage change by varying the inductance path between a primary coil and two or more secondary coils when an excitation voltage is applied to the primary coil. Variations in the output signal can be calibrated to displacements of the LVDT core. The calibration is usually linear. The coils are wound on a hollow mandrel. A ferromagnetic core slides within the mandrel to vary the inductive coupling. AC voltage is usually used, but DC models are available. Readout units contain an accurate voltmeter and a carefully regulated power supply. Since the output voltage is proportional to the input voltage, the power supply must be controlled. Since the electrical portions of the LVDT can be isolated and waterproofed by encasing the mandrel in potting compound, the LVDT is much less sensitive to



-

moisture than a linear potentiometer. It is also less affected by temperature and is mechanically simpler. LVDT's are available with displacement ranges from 0.1 in. to several feet. Sensitivities of 10 microstrain are possible.

Mechanical Strain Gauges

An alternative to the remote-reading electrical gauges is a mechanical strain gauge that uses a dial gauge indicator for direct reading. Since dial gauges are easily damaged and cannot be left in place, systems have been designed that allow removal of the gauge between readings and replacement of the gauge in a way that provides sufficient repeatability in the readings. A typical mechanical strain gauge (such as the Whittemore or Huggenberger types) has two pointed arms that fit into conical holes or gauge points drilled and punched into the instrumented surface, or into studs set into or on the surface. The change in distance between the gauge holes is measured by determining the distance between the arms when inserted in the holes. Each arm is attached to the end of a tube. The two tubes are coaxial and slide one inside the other. A dial gauge indicator clamped to the inner tube measures the displacement between the tube ends. An initial measurement of the separation between the gauge holes is made. Changes in this distance recorded in subsequent measurements serve to determine the surficial strains. The gauges usually read to the nearest 0.0001 in., but the repeatability of the readings depends partially on the quality of the seat for the points and partially on the skill of the operator. Gauge lengths of 2 to 80 in. are available. For most engineering measurements, a 10-in. gauge length is used (with a sensitivity of about 10 microstrain).

The holes for one of the typical mechanical gauges, the Whittemore gauge, are 1/8 in. deep and are made with a standard No. 54 drill. The holes are then countersunk with a 90-degree center angle conical punch. This forms a cylindrical edge upon which the measuring arm points, which have 30-degree conical points, can rest (see Figure G-10a). The punching action also serves to strain harden the surface where the measuring arm point and the side of the reference hole touch. An alternative procedure is to use a No. 1 high-speed combination drill and countersink to make a 1/8-in. deep by 1/32-in. diameter conical hole and to use spherical contact points that fit snugly in the holes (see Figure G-10b).

While gauge holes can be drilled directly into a steel or cast iron member, metal inserts are used for rock. These inserts are stainless steel bolts grouted into holes drilled into the rock. The gauge holes are then drilled into the ends of the inserts. For temporary readings, holes can be made on small pieces of steel which are then glued in place.

To ensure repeatability with the mechanical gauges, extreme care and precision must be exercised during their initial installation. The gauge holes must be drilled so that they are exactly perpendicular to



the line of the instrument gauge casing and at the required spacing. The field engineer should acquire some laboratory experience with such drilling prior to site installation.

Measurements are made by placing the gauge contacts into the gauge holes. The gauge is rocked back and forth gently to provide a secure junction of the contacts with the gauge holes. When the dial gauge has stabilized, a measurement is taken. An awkward location for measurements can often induce imprecision; thus, the gauge holes should be arranged so that they can be measured from a reasonably comfortable position. Multiple measurements are typically used to improve the reliability of results.

An important part of the mechanical strain gauge is a calibration bar. This bar is used to determine temperature compensation and also to provide a reference in case it is necessary to repair or replace the gauge. The bar is made of invar steel and mild steel and is designed so that the expansion of the invar will be cancelled by the expansion of the mild steel (see Figure G-10c).

Mechanical strain gauges are simple, reliable, relatively low-cost devices that can be used in a variety of situations. They are not as sensitive as most of the electrical type strain gauges, but their sensitivity and accuracy are adequate for verifying performance of other types of gauges or for use as a backup to other gauges. Since the mechanical gauge is removed between readings, its calibration can be checked at any time (a considerable advantage). The gauge holes are easy to protect from damage and water. The disadvantages of the mechanical gauges are that they cannot be read remotely, and the repeatability of the readings depends on the skill and experience of the operator.

Portable Extensometers

The two common types of tunnel convergence gauges are the rod extensometer (see Figure G-11a) and tape extensometer (see Figure G-11b). The micrometer and dial gauge rod types can provide an accuracy of ±0.08 mm, while the tape extensometers have accuracies within the range of ±0.06 to 2.5 mm, depending on the spring tension force used in the tape. An error of one pound in the spring force can cause an inaccuracy of 0.1 mm (Dunnicliff, 1971). The rod types are bulkier and more easily damaged. Their use is generally limited to spans of 10 ft or less, unless sectioned rods are employed. For larger spans, flexure of the rod may reduce the accuracy to ±0.1 in. or less. The tape units are more versatile and can be easily used for spans up to 30 ft or more.

The rod and tape extensometers both use reference heads that are set in the rock. The reference heads are usually epoxied into small holes (3 to 15 mm diameter) that are drilled into the rock and protective covers are placed over the head to prevent damage between readings. The heads are designed to provide a positive, repeatable seating for the extensometers.



Both types of units are temperature sensitive and standard temperature corrections should be applied. Before taking measurements, the extensometers should be allowed to come to equilibrium with the surrounding air temperature. A calibration bar consisting of a steel bar with hook attachments for the tape can be placed in the tunnel to calibrate the steel tape before each set of measurements, and thus provide for mechanical adjustments of the unit.

G.3.2 STRESS

Strictly speaking, there is no way to directly measure and monitor the absolute stress at a point, as the installation of the monitoring instrument causes a change or redistribution in the stress field. Only changes in stress, once the instrument is installed, can typically be measured. However, procedures have been developed for back-calculating the original absolute stress from those types of measurements.

There are two borehole methods that are used to measure stress changes (see Table G-3): the hard inclusion cell and the soft inclusion cell. The hard inclusion cell is a stiff gauge that is placed in a borehole and is either cemented or grouted in place. If the stiffness of the cell is at least five times the stiffness of the rock, then the cell measures changes in the rock stress directly and is independent of the deformational properties of the rock. The soft inclusion cell is placed in the borehole and measures the deformations of the borehole wall. The stress changes are calculated from these displacements, based on elastic theory and elastic properties of the rock in the borehole wall.

Hard Inclusion Cells

There are three main types of hard inclusion cells, although there are many variations of these types available. The benefit of hard inclusion cells is that they read stress change directly and are not sensitive to the deformation properties of the rock.

The solid inclusion stressmeter is a brass or steel bar that has a strain gauge embedded in it. The bar is wedged or cemented into a borehole and it monitors stress changes that are parallel to the orientation of the strain gauge. Before installation, the cell may be calibrated in the laboratory by mounting it in a borehole drilled in a large block of rock. The block is then subjected to various stresses in a load frame and the response of the strain gauge is monitored. In the field, the stressmeter may be installed in boreholes up to 30 m deep. The vibrating wire stressmeter (see Figure G-12) uses a 37.5 mm diameter borehole and as many as three cells may be stacked in a single borehole.

The "stiff" hydraulic cell can be considered as another type of hard inclusion cell. This is a flatjack that is filled with a stiff fluid, such as mercury, so that the fluid has approximately the same stiffness as the rock. The flatjack may be installed in a slot cut in the wall of Rev._____ Dwg. No. 8/2-8638 Abesa Date /-82 Eng. w.w.

NSTRUMENT RANGE ACCURACY ADVANTAGES LIMITATIONS AND PRECAUTIONS ARD INCLUSION CELLS Calculation of stress is not dependent on elastic proper- ties of the rock. Ibrating wire 69 MPa ± 0.1 tressmeter 69 MPa ± 0.1 MPa Reusable Need to correct for creep of wire for long-term measure-	ESTI (1982 1 ACQUI- IN SITTON 1 \$500	MATED C U.S. DON VSTALL- ATION	OST LARS) CALI- BRATION
NSTRUMENT RANGE ACCURACY ADVANTAGES LIMITATIONS AND PRECAUTIONS ARD INCLUSION CELLS. Calculation of stress is not dependent on elastic proper- ties of the rock. Need to correct for creep of wire for long-term measure- measure-	ACQUI- IN SITION S	VSTALL- ATION	CALI- BRATION
ARD INCLUSION CELLS Calculation of Stress is not direction. elastic proper- ties of the rock. Need to correct for creep of wire for long-term measure- measure- measure only one stress direction.	\$500		
tressmeter MPa + 0.1 Reusable Need to correct for creep of wire for long-term measure-	\$500		
is required.		\$280	\$280
tiff Hydraulic ell 20 MPa ± 1% Simple, rugged Stiff fluid must be used to ensure that the cell is stif- fer than the rock.	\$500	\$280	\$140
lotz]Hy- 30 MPa ±1 MPa Simple, rugged. Requires constant volume pump. raulic Cell Cannot measure stress decreases.	\$750	\$280	\$140
DFT INCLUSION CELLS Measure stress change tensor. Stress is calculated based on measured deformation of bore- hole walls and the elastic rock properties. Cannot be used in fractured rock or rock that is creeping.			
SBM Bore- Long experience. Can only calculate stresses ole Deformation Reusable, rugged. perpend/cular to borehole. auge	\$2,500	\$140	\$140
oorstopper Can be used in rock subject to core discing.		\$560	\$560
SIR Cell All three prin- Difficult to seal strain cipal stresses gauges against moisture. monitored in one cell.		\$560	\$560
SIRO Tri- 100 MPa + 0.2 All three principal xial Cell MPa stresses measured	\$600	\$280	\$560


a tunnel or in a borehole. In either case, it is pressurized approximately to the rock stress acting normal to the flatjack. Changes in stress are measured by monitoring changes in the pressure of the fluid in the cell. If the initial pressure in the cell is too low, then changes in stress will tend to arch over the flatjack and the measured pressure change will be too low. The pressure in the cell can be read using either a mechanical pressure gauge or a diaphragm connected to a strain gauge.

The <u>Glotzl cell</u> is similar to the "stiff" hydraulic cell, except that it is connected to a constant volume pump. It is assumed that the pressure required to maintain a constant volume in the flatjack is equal to the change in the rock stress normal to the flatjack. The Glotzl cell is cemented into a borehole and pressurized to the approximate in situ stress normal to the cell. Changes in pressure necessary to maintian constant volume are monitored.

The accuracy of the <u>hard inclusion cells</u> is limited because the inhomogeneities in a rock mass result in a nonuniform stress distribution at the measurement scale, i.e., the hard inclusion cells measure the stress changes in a small area which may not be representative of the rock mass. Also, the hard inclusion cells measure stress changes only in one direction (longitudinal to the strain gauge or normal to the flatjack faces). Thus, it is necessary to use six cells oriented in different directions to determine the changes in the entire stress tensor. In addition, there is always some disturbance of the rock around the cell and this may affect the stress changes measured by the cell. Under laboratory conditions, the hard inclusion cells have an accuracy of 5 to 10 percent, but under field conditions, the accuracy has been estimated at about 25 percent.

Soft Inclusion Cells

Soft inclusion cells monitor the deformation of the rock around a borehole, so that the stress changes can be calculated. There is one type of mechanical cell and three types of electrical cells that are commonly used. All four cells were originally developed to determine the in situ stress tensor using overcoring techniques (see Appendix, Section A.8), but they can also be used to monitor borehole deformation and stress changes by installing them in a borehole and not overcoring.

The USBM borehole deformation gauge (see Figure G-13) is a mechanical gauge that is installed in a 37.5 mm diameter borehole. The gauge measures three diameters of the borehole using resistance strain gauges mounted on cantilevers. The changes in the diameters are related to the stress changes in the rock around the borehole (see Appendix, Section A.8 for equations). The cell may be used in boreholes up to 30 m deep, and only one cell may be used per hole. The cell has two advantages compared to the other soft inclusion cells:

 The cell may be used in wet holes because it does not need to be bonded to the rock



The cell may be easily removed for repair or calibration.

The cell can measure changes in diameter of as little as 0.08 mm, which would correspond to a stress change of about 0.1 MPa. Three boreholes with different orientation are required to determine the complete stress tensor. Calibration consists of checking changes in resistance of the strain gauges against known displacement.

The <u>CSIR</u> "Doorstopper" (see Figure G-14a) consists of a strain gauge rosette that is epoxied to the bottom of a borehole. The rosette is encased in a potting compound in order to protect it from water. Installation is as follows:

- A 37.5 mm diameter hole is drilled and the bottom is ground smooth using a special bit
- The hole is dried using compressed air
- The bottom of the doorstopper is coated with epoxy, placed in the hole, and pressed against the bottom of the hole using a special tool (see Figure G-14b) until the epoxy dries.

After installation, initial strain gauge readings are taken using a Wheatstone bridge type read out.

The <u>CSIR triaxial strain cell</u> (see Figure G-15) is an apparatus which glues three electrical strain gauge rosettes directly to the wall of a 37.5 mm diameter borehole.

The cell measures 12 separate strains in 3 different planes, which by appropriate analysis can be related to the change in stress at that point within the rock. Only six independent measurements are required to determine the six components of the stress change tensor. However, as all the strain measurements contain some errors, the additional readings (six in the case of the four element rosette strain cell) can be included to improve the accuracy and obtain an indication of the precision of the data.

The <u>CSIRO hollow inclusion triaxial strain cell</u> (see Figure G-16) is identical in principle to the <u>CSIR</u> triaxial cell. In the <u>CSIRO</u> cell, however, the three strain guage rosettes are embedded in a hollow cylindrically shaped epoxy probe. When the cell is glued to the wall of a 37.5 mm diameter borehole, the strain gauges are separated from the wall by a thin layer (1 to 2 mm) of epoxy. This partially alleviates the difficult practical problem of bonding the gauges in the borehole.

The boreholes for the soft inclusion cells are cored and the cells are installed in a location that is free of discontinuities. When the cell no longer needs to be used for stress change monitoring, it is overcored. Both the overcore and the core samples must be tested to determine the deformation characteristics of the rock around the cell.







B...

The main limitation of the soft inclusion cells is that the deformation properties of the rock must be relatively well known in order to calculate the stress changes in the rock. However, most analyses assume that the rock behaves as a linear elastic material. The cells cannot be used reliably in rock that is subject to creep or is fractured. The Doorstopper, CSIR and CSIRO cells are subject to some potential error because the epoxy may creep with time during stress change monitoring. The cells also are affected by inhomogeneities in the rock which result in a nonuniform distribution of stress in the rock mass at the measurement scale.

G.3.3 TEMPERATURE

The two most commonly used types of temperature sensors are thermocouples and thermistors (see Table G-4). However, vibrating wire gauges can also be used. The temperature sensors are mounted in probes and installed in boreholes to monitor the temperature of the rock. Also, the sensors may be mounted on other instruments, such as extensometer rods or strain gauges to facilitate making temperature corrections.

The same calibration method is used for both thermocouples and thermistors. The sensors are placed in baths of known temperature and the potential difference or resistance measured. Both types of probes are generally calibrated by the manufacturer, so little calibration is required in the field.

The probes may be installed practically anywhere that a temperature measurement is needed. For measuring the temperature of a rock mass, the sensor is placed in a borehole which is then backfilled with sand. A series of measurements is taken until the probe reaches equilibrium. This is necessary because the temperature of the rock around a borehole may be influenced by the temperature of the drill fluid used, the friction from the drill tools, and the temperature of the sand backfi'l. After equilibrium has been attained (this could take from a few hours to a few weeks), both types of temperature sensors have a very fast response to changes in temperature in the rock and both are very accurate.

Thermocouples

Thermocouples consist of two wires made of dissimilar metals, such as chromel and alumel. The wires are jointed together at the point where the temperature measurement is to be made. The free ends of the wires are maintained at a constant reference temperature. The difference in temperature results in a potential difference between the two free ends which can be measured with a potentiometer. The potential difference can be related to the temperature at the junction of the two wires. An advantage of the thermocouple is that the resistance of the leads is not critical. Hence, when long lengths of lead wire are required, the thermocouple is usually selected over the thermistor so that readings do not have to be corrected for the resistance of the wire.

Dwg. No. 8/3-1618 4+031 Dete 1-82 Eng 1/ 4

						ES	TIMATED COS	515
	INSTRUMENT	RANGE	ACCURACY	ADVANTAGES	LIMITATIONS AND PRECAUTIONS	ACQUI- SITION	INSTALL- ATION	CALI- BRATION
	Thermocouple		<u>+</u> 1°C	Reliable, rugged.	Reference temperature necessary	\$150	\$70	\$70
162	Thermistor	0 to 100°C	± 0.2℃	Long experience.	Correction for lead wire resistance is necessary	\$450	\$70	\$70

Table G-4

Thermistors

Thermistors measure temperature based on the change in electrical resistance with temperature. This change in resistance can be measured very accurately with a Wheatstone bridge. The resistance of the leads to the probe must be accurately known and readings must be corrected for lead wire resistance. It is thus necessary to measure the resistance of the leads at various temperatures so that corrections can be applied.

G.3.4 GROUNDWATER PRESSURES

Instruments used to monitor groundwater pressures are called piezometers. The data (pressure head) is used to determine the piezometric head (elevation head plus pressure head) in a flow field at the point of measurement and, if the flow is transient, the changes in head with respect to time. The piezometers are sealed within individual sections of a borehole and isolated from contact with other zones so that the location of the reading is accurately known.

Careful installation is required for accurate measurements (see Figure G-17). The piezometer tip must be sealed into the zone where pressure measurements are desired and the seal must not allow any leakage which could "short circuit" the pressure reading. Usually the piezometer tip is placed in the borehole and sand is tamped around the tip. Bentonite is then tamped on top of the sand. The bentonite has low enough permeability to prevent short circuiting. Pneumatic, mechanical, or hydraulic packers may also be used to isolate a portion of the borehole. Packers must be used in holes that are inclined upward. The piezometers may be stacked in a borehole to measure the distribution of water pressure along the length of the hole. The piezometer tip and the zone of measurement must be de-aired or erroneous measurements will result.

Piezometers all operate on the same basic principle. This involves the measurement of an internal pressure head which exactly balances the pressure (piezometric) head external to the instrument. The internal and external heads are usually balanced through a flexible membrane, which is located in a porous tip or filter installed in the ground at the point to be monitored. The tip of the piezometer is usually installed in a filter pack, such as Ottawa sand, to increase its collection area and sensitivity.

Piezometers are classified on the basis of the technique used for measurement. The three basic types of piezometers are open system, closed system and diaphragm piezometers.

Open System Piezometers

In an open system piezometer, the elevation of a free water surface that fluctuates with changes in external head is measured in a standpipe. The free water surface is located by dropping a probe down the standpipe until it contacts the water surface. The depth is measured on the



marked probe cable. The open system piezometers are the most reliable and least expensive of the various piezometers. Their principal disadvantages are that they are very slow to respond to changes in piezometric head (because a considerable volume of water must flow into the standpipe to register a change in head), they cannot be used in upward inclined holes, and the location and source of flow into or from the hole is often not easily determined.

Closed System Piezometers

Closed system piezometers use a fluid filled hose to transmit the external head to a pressure measuring device, usually a Bourdon gauge or manometer. An open piezometer can usually be converted to a closed system by adding a pressure gauge, if the piezometric head rises above the top of the open system standpipe. Closed systems usually use two tubes to allow the flushing out of trapped air or gasses that would otherwise lead to erroneous readings. Closed systems are more expensive and difficult to maintain than open systems, but have advantages over open systems in that they can be connected to a central monitoring station, can measure negative pore pressures, and are more sensitive than open systems.

Diaphragm Piezometers

Diaphragm piezometers use a sealed, flexible or movable diaphragm to separate the interal and external heads. Diaphragm piezometers can be divided into fluid-actuated sensor and electrical sensor types (see Table G-5). Although these piezometers are more expensive, sophisticated, and prone to failure than other types, they do have high sensitivity and rapid response time. The diaphragm responds to very small head changes quickly because very little water flow is required to displace the diaphragm. These piezometers can be read remotely from a central station and can measure negative pore pressures. They are delicate and a 10 percent failure rate is common. However, they are often the only units which are sensitive and respond rapidly enough for low permeability materials.

In the <u>fluid-actuated sensors</u>, the fluid pressure is adjusted (either manually or automatically) to balance the external pressure on the diaphragm. When the differential pressure across the diaphragm is zero, the fluid pressure is measured at the surface. Various systems of valves and flow indicators are used to detect the balance condition. The sensitivity of the fluid-actuated sensors is a function of the deformation induced in the diaphragm per unit of differential pressure across the diaphragm. The measurements are usually repeated several times to improve the accuracy of the reading. The fluid can be hydraulic (e.g., oil, see Figure G-18) or pneumatic (e.g., air or nitrogen gas, see Figure G-19).

The electrical sensor types have a strain gauge or displacement transducer attached to the diaphragm. The deflection of the diaphragm under the external head is measured to determine the pressure head. Either

	APPLICABLE					(1982 ACQU1-	INSTALL-	CALI-
INSTRUMENT	PERMEABILITY RANGE	ACCURACY	ADVANTAGES	LIMITATIONS AND PRECAUTIONS	RELIABILITY	STTION	AT LON*	BRATION
DIAPHRAGM PIEZO- METERS	10 ⁻⁹ to 10 ⁻¹¹ cm/sec	Depends on pres- sure gauge used Typically * 1% or less of full scale	Very small time lag, usable in very low permeability media. Central ob- servation system can be used. Can measure negative pore pres- sures.	Most costly and difficult to install and operate.	Good to Fair			
(FLUID ACTUATED	SENSOR)							
Hydraulir Types			Simple. Easy to seal against leakage.	Can not measure negative pore pressures. Requires constant volume pumps.		\$250	\$280	\$35
Pneumatic Types			Not subject to freezing. Smaller, less expensive tubing used and less sensitive to tem- perature, relative to hydraulic types.	Can not measure negative pore pressures. Leakage and mois- ture in lines are problems.		\$200	\$280	\$35
(ELECTRICAL SENS	OR)							
Electrical Re- sistance Strain Gauge Type		Depends on trans- ducer. Typically * 1% or Tess of range.	Can often be local- ly fabricated from commercially avail- able parts. Adapt- able to automatic data recording. Can measure negative pore pressures.	Service life of resistance type strain gauge is often limited. Susceptible to wiring damage. Correction for temperature needed.		\$500	\$280	\$35
Vibrating-Wire Strain Gauge Type			More reliable than resistance strain gauge type. Adapt- able to automatic data recording.	Correction for temperature needed.		\$200	\$280	\$35



.

ł



resistance or vibrating-wire strain gauges can be used as sensors (see Figure G-20a and b, respectively).

G.3.5 DATA ACQUISITION SYSTEMS

The advantage of electronic instrumentation is that it is compatible with the use of computerized data acquisition systems. Computerized data acquisition systems can increase the reliability of data measurement, and decrease costs by automating readings, and preprocessing data.

The functions that can be performed by data acquisition systems include:

- Automatic timed readings of instrumentation
- Recording of instrumentation data (analog)
- Conversion of analog signals to numerical (digital)
- Recording of digital information
- Preprocessing of data for analysis
- Graphical and tabular presentation of data
- Data analysis
- Graphical and tabular presentation of analytical results
- Storage of data and results
- Transfer of data and results to computing facilities.

Data acquisition systems can perform any number of these functions, and the cost of the system is directly proportional to its capabilities. In situ testing requires a sufficiently large number of readings so that the use of a computerized data acquisition system is generally justified.

Data acquisition systems can be divided into two categories:

- Test specific devices which read and record data from only one test
- Centralized systems which read and record data from a number of tests simultaneously.

Test specific devices have the advantage that they are portable, and can be located at the test site. They can also be designed to provide exactly the needs of that specific test. Disadvantages include increases in expense due to duplication of reading and recording capabilities among data acquisition systems for different tests. In



addition, individual reading and recording devices lack the advantages of centralized facilities, and data must be periodically transferred to a computer facility for reduction, analysis, and presentation.

Centralized facilities offer greater capabilities and flexibility, but suffer from problems of connection to instrumentation, data loss over greater distances, and risks of major disruption due to system malfunctions. Centralized facilities can offer on-line data analysis for continuous monitoring.

The choice between test specific and centralized facilities is one which must be made during the design of the overall test program. The magnitude of the data acquisition system required is related to the number of instruments to be read, the frequency of reading and the need for concurrent analysis of data.

A data acquisition system is made up of a combination of the following components:

- Central Processing Unit (CPU) controls the logic of data acquisition including control of measurement and recording devices, and reduction and analyses of data
- <u>Real Time Clock</u> provides the CPU with time signals required for scheduling of reading
- Analog-Digital (A-D) Converter converts analog signals from measuring devices (generally voltages) to digital signals understood by the CPU
- Multiplexer couples numerous input and output channels to the CPU
- <u>Data Storage Devices</u> include high speed (disk) and low speed (tape) storage systems. Storage devices provide both permanent and temporary storage of data from initial measurement through archiving.
- Input/Output (I-O) Devices include on-line terminals such as CRT's, teletypes, and decwriters, highspeed printers, and plotting devices. I-O devices provide the interface between the computer and engineering personnel.
- Software programs allow the engineer to control the data acquisition system through the CPU. Software includes the operating system for control of data storage and input/output devices, high-level programming languages such as Basic and Fortran, and applications programs written to perform the specific functions required from the data acquisition system.

The combination of components, and the size and speed of CPU, data storage, I-O devices, and software determines the capability of the data acquisition system. Typical unit costs for system components have been estimated (see Table G-6).

\$

SYSTEM COMPONE	Table G-6			
COMPONENT	SPECIFICATION*	MODEL (1	ESTIMATED COST 982 U.S. Dollar	rs
Central Processing Unit	Nova 4/C 64 KB, 16 Slot	8391-H	6,092	
Real Time Clock	Data Acquisition Control Subsystem	4300	2,670	
Analog-Digital Converter	DG/DAC 50 KHZ A/D Converter (+/- 10V)	4280	1,820	
Multiplexer	DG/DAC 16-Line Differential A/D MUX	4281	670	
Data Storage Device (High Speed)	Winchester 12.5M Byte Moving Head Desk Including Controllers	6099	6,050	
Data Storage Device (Slow Speed)	1600 BPI Streaming Tape Drive Subsystem	6125	6,800	
I/O Devices (Video Terminal)	Dasher D200 Alphanumeric CRT Display	6109	2,350	
I/O Devices (Printer)	Dasher TPI Printer	6043	2,200	
Software (Operating System)	DOS-Initial License/STR/ CSS/ 1 YR SSS/ Install	3574-01F	2,000	
Software (Programming Language)	DOS Fortran IV Initial License	3597-01F	1,100	
Miscellaneous	19" Cabinet with Blower, Circuit Breaker, Auxiliary Blower	1144-A 4269	2,020	

*Based on provisional quotation from Data General Corporation. The use of Data General equipment in this example does not signify endorsement of their products or an offering by Data General. Number of units, models and actual costs must be determined when the entire data acquisition system is designed.

G.4 COMMON SITE PREPARATION AND CHARACTERIZATION ACTIVITIES FOR IN SITU TESTING

G.4.1 EXCAVATION

There are two basic methods which are used to excavate underground openings: the drill and blast and mechanical excavation methods. Both of these methods may be used for the excavation of an in situ test facility and for the excavation of an HLW repository.

Drill and Blast

Drill and blast excavation uses high-energy explosives placed in holes that are drilled in the excavation in order to reduce the rock to blocks and fragments that may be easily removed from the tunnel. The method consists of three main operations:

- Dril. blast holes
- Blast
- Muck removal.

Percussion drills are typically used for drilling the blast holes. These drills are often mounted on a drill jumbo (see Figure G-21) or, more typically for small excavations, single air leg drills are used. The holes are drilled in a pattern that is dependent on the geological conditions, the size of the tunnel, and the limitations of the mucking equipment. Various patterns may be used to minimize the disturbance to the rock around the tunnel while maximizing blasting efficiency.

The blast holes are charged with explosives such as dynamite or ammonium nitrate/fuel oil. The amount of explosive used in each of the blast holes depends primarily on the requirement to produce adequate fragmentation of the rock without causing excessive damage to rock around the tunnel. The explosives are ignited using various millisecond delays which allow the firing to occur in a sequence. The firing sequence is designed to maximize the number of charges that will blast towards a free face. Blasting towards a free face maximizes the efficiency of the blast and minimizes the damage to the surrounding rock.

The blasted rock is removed during the muck cycle. Tunnel supports may be installed during the muck cycle and prior to or during the following drill cycle.

The disturbance to the rock surrounding the tunnel is a function of the following:

- Blast hole size, length, and orientation
- Charge per blast hole
- Blast hole pattern
- Firing sequence



Rock strength, including the effects of discontinuities

In situ stresses.

Various techniques can be used to minimize the amount of disturbance. One common technique is smoothwall blasting. This involves drilling small diameter "trim" holes at close spacing around the perimeter of the tunnel. The "trim" holes are loaded with a light charge. They are delayed so that they all fire simultaneously and are the last holes to fire. This ensures that they all fire towards a free face.

The advantages of the drill and blast method are:

- It can be used in any rock type
- It is versatile and adaptable to changing geological conditions along the alignment of the tunnel
- Any size and shape of opening can be excavated
- Mobilization costs are low.

Costs for drill and blast tunnel excavation and support of two sizes of tunnel in salt, we'ded tuff, basalt, and granite have been estimated (see Table G-7). In all cases, it has been assumed that the contractor and necessary equipment have already been mobilized at the site.

Mechanical Excavation

There are two mechanical excavation methods: full face and part face.

Tunnel boring machines (TBM's) excavate the full cross-sectional area of the tunnel in one pass, similar to a very large diameter drill. A TBM has a rotating shield or head that is about the same diameter as the tunnel that is to be excavated. Small cutting wheels are mounted on the rotating shield. The cutting wheels chip, grind, and fracture the rock as the shield is rotated and thrust forward. All but a few of the TBM's currently available excavate a circular cross section. TBM's have been used to excavate tunnels from 2 to 12 m in diameter. TBM's are best suited for use in tunnels with uniform, well-known geologic conditions where the rock is stable and is not very abrasive. TBM's are not very versatile and unexpected geologic conditions can result in long delays in excavation.

TBM's have the following advantages:

- Cause very little disturbance to the rock around the tunnel, so that less support is required
- May be more economical than drill and blast in long tunnels in suitable geologic conditions.

TUNNELING COST ESTIMATES

Table G-7

	ESTIMATED COST (1982 U.S. DOLLARS/LINEAR METER)					
Media	3 X 3 m Drill & Blast	3 m Ø TBM	6 X 6 m Drill & Blast	6 m Ø TBM		
Salt	800	700	700*	500*		
Welded Tuff	900	800	1,500	1,300		
Basalt	1,200	NA	2,100	NA		
Granite	1,000	NA	1,900	NA		

*Unit cost for 6 X 6 m opening in salt is lower than 3 X 3 m opening because excavation and mucking equipment are well developed for the larger size, resulting in more efficient excavation.

Note: These estimated costs include excavation and support, but not mobilization.

w w Eng

 May have higher rates of advance than drill and blast method in suitable geologic conditions.

TBM's are usually not used in short tunnels because of the high costs for acquisition and the long time required for assembly and mobilization.

There are several types of part face tunneling machines available. One of the most commonly used types is the road header (see Figure G-22). The road header has a small cutting wheel mounted on an articulated boom. The operator moves the cutting wheel across the face in order to cut the desired cross section. The road header is suitable for tunnels less than about 6 m in diameter in weak to moderately strong rock. The road header has advantages over both drill and blast and TBM's:

- The amount of disturbance to the rock is comparable to that caused by a TBM
- The road header is very mobile and can easily be adapted to various geologic conditions. If rock is encountered that is too hard, it can easily be moved back from the face to allow intermittent drill and blast tunneling through the hard rock.
- Any cross-sectional shape can be cut.

Mechanical excavation is probably suitable for use in salt and tuff and probably not suitable in granite and basalt. A road header would probably be more feasible for the in situ test facilities because of the various shapes, sizes, and short lengths that are envisioned. Even though a full face TBM might be used in the final HLW repositories, a road header would be suitable for simulating the disturbance caused by a TBM.

Cost for mechanical tunnel excavation and support for two sizes of tunne! in salt and welded tuff have been estimated (see Table G-7). In all cases, it has been assumed that the contractor and the necessary equipment have already been mobilized at the site.

G.4.2 DRILLING, SAMPLING, AND LOGGING

The in situ test program will require drilling boreholes in various orientations from underground openings for installation of the instrumentation, as well as for assessment of characteristics (see Figure G-23). Boreholes may be bored using either coring or noncoring methods, depending on whether core samples are required. Borehole logging techniques may be used to provide additional information. Drilling and some direct logging costs have been estimated (see Table G-8); these estimated costs do not include mobilization or analysis/ interpretation of data.





Figure G-23



after Goodman, 1976

Eng. 4.1

Dwg. No. 2/14/18 44-021 _ Date 1-12

179

t	DRILLING AND DIRECT	т	Table G-8			
A)	CORING Size Designation	Hole Diameter (mm)	Core Diameter (mm)	ESTIMATED COST (1982 U.S. DOLLARS)		
	FX	37.5	22.2	66		
	NX	75.8	47.6	100		
	PX	122.6	85.0	130		
	NX-Oriented	75.8	45+	130		
	Integral Sampling	122.6	85.0	250		
B)	NON-CORING	50		25		

C) DIRECT LOGGING COSTS

Method	
Borehole Television	
Seisviewer	
Caliper Logging	<pre>(typically included in standard suite of geophysical logs)</pre>

6

Note: These estimated costs do not include mobilization or subsequent analysis/interpretation.

Core Drilling

The two general methods of core drilling in common use differ in the way core is retrieved from the hole. In the conventional method all of the drill rods and the entire core barrel are removed from the hole at the end of each run. In the wireline method, the inner tube of the core barrel is removed through the center of the drill rods by a cable. Wireline is not generally applicable for use in short boreholes drilled underground.

All drilling methods require a drill fluid to be pumped through the drill rods to the bottom of the hole to cool the bit and remove cuttings from the hole. The most frequently used fluid is water, sometimes with bentonite or soluble oil added. Compressed air may also be used. The disadvantage of using water for drilling in salt is obvious. The problems associated with using air is that it is difficult to blow cuttings from deep holes, and any moisture in the hole will result in the formation of a mud cake which can prevent the discharge of cuttings.

Core recovery is influenced by the type of bit used and the type of core barrel. The selection of the proper bit is greatly dependent upon the characteristics of the rock being cored, such as its solubility, hardness, abrasiveness, and degree of cementation. Diamond bits are usually used for coring, although tungsten carbide bits may be used in a soft rock. The type of barrel most often used is a double tube swivel-type design in which the inner barrel does not rotate with the outer barrel and cutting bit. In highly fractured or loosely cemented rock, a triple-tube core barrel in which the inner tube is split lengthwise is highly recommended. With this barrel, the inner tube is removed and opened lengthwise to expose the core, thereby assuring a relatively undisturbed sample for study and photography.

Two methods used to obtain additional information from cores are:

- Integral sampling This specialized overcoring technique is used to recover badly fractured rock with a minimum of disturbance, i.e., complete recovery and with fracture aperture unchanged. A small-diameter hole is drilled and a steel rod is grouted into the hole. The rod is then overcored with a larger diameter bit, and the rod and surrounding cylinder of rock are removed as a unit (Rocha, 1971). This method is slow and difficult, and not feasible for deep holes.
- Oriented coring Various instruments and techniques have been devised in an attempt to determine the original in situ orientation of drill core (Goodman, 1976). The objective is to allow determination of the absolute dip and strike of bedding planes and fractures. Mechanical, electrical, and magnetic methods have been attempted but no completely satisfactory system has yet been developed. Any attempt to obtain and utilize oriented cores, particularly from deep holes, should include a borehole survey.

Noncore Drilling

In some cases. it may not be necessary to obtain core samples from the boreholes, so that it is not necessary to use a coring method. There are three main types of noncore drilling methods:

- Blind bits are similar to coring bits, but they cut the full cross section of the rock. Blind bits may use either diamonds or tungsten carbide for cutting the rock.
- Rotary bits use three rollers with hardened steel teeth or tungsten carbide buttons to grind the rock.
- Percussion drills advance the hole by repetitious rapid blows, thereby crushing and chipping the rock ahead of the bit. Downhole hammers, which have a hydraulic or compressed air motor just behind the bit, are suitable for large diameter deep holes.

Noncoring methods are suitable for advancing a borehole down to the point where coring is required or for drilling holes that will be logged using geophysical or other methods. Noncoring methods generally are not suitable for drilling holes for installing instrumentation because of the need to know the geologic details that may affect the instrument installation and data interpretation.

Borehole Direct Logging

Valuable information may be obtained from observation and measurement of the borehole walls. Boreholes may be logged using direct observation and measurements or by geophysical methods which log some property of the rock in the walls of the borehole. The advantages and limitations of some of the common types of borehole logging methods are as follows:

- Borehole survey Boreholes seldom follow their initial bearing and inclination. Even "vertical" boreholes drift and are seldom if ever plumb. A borehole survey measures the true inclination and direction of the borehole at frequent intervals in order to determine the true location of any point within that borehole.
- Borehole photography and TV Photographic and TV cameras have been developed that can be lowered into boreholes three inches in diameter or larger. One of the cameras photographs the image on a conical mirror, which gives a donut-shaped photograph of the circumference of the hole; calculation of true strike and dip is possible. On one model of TV camera, the lens is directed downward onto an angled mirror that can be rotated from a control panel. Another model can be made to view axially, radially, or with a 180 degree lens opening and can be fitted with a zoom lens and a powerful floodlight. Dips and strikes can be estimated and the TV images can be recorded on tape for more detailed study. In addition to allowing determination of the spacing and orientation of fractures, joints and bedding, borehole cameras allow inspection of

zones of little or no core recovery. Their major disadvantage is that in holes filled with muddy or oily water they do not work well and are subject to frequent malfunction.

Under ideal conditions, camera surveys can give an understanding of subsurface conditions that is unobtainable by any other means except by direct inspection in large diameter shafts and adits.

A related instrument is the borehole televiewer which presents a continuous acoustic picture of the borehole produced by a rotating ultrasonic scanner. The reflected ultrasonic waves are converted into a visual image of the borehole. Fracture systems and the dip of fracture planes can be determined by this method (Zamanek, 1970). In order to obtain a good borehole picture, a slow and constant logging speed should be used and the suspended solids content of the borehole fluid should be low.

- Caliper logging Caliper logging provides a continuous recording of borehole diameter versus depth. The caliper has three arms that can be controlled. The logs are used to determine borehole diameter, to locate caved zones, to identify fractures, and to correlate geologic boundaries. Geophysical logging methods can be correlated with the caliper logs to enhance the interpretation of the data and to obtain direct correlation with the actual rock condition. Thus, caliper logs are often obtained in conjunction with geophysical logs to aid the interpretation of the data.
- Impression packer logging Impression packers are soft plastic packers that are inflated against the wall of the borehole in order to obtain an impression of the features that are present on the wall. The impression packer is oriented so it is possible to determine the strike and dip of fractures and other features on the borehole walls. It is also possible to obtain a rough estimate of joint aperture using the impression packer.

Borehole Geophysical Logging

Geophysical well-logging techniques test the earth with instruments inserted into boreholes; data are transmitted to recording devices. The logs give a detailed and continuous record of the borings and allow detection of subtle layer boundaries that can often be correlated between borings.

Qualitative interpretation of the logs enables estimating of porosity, permeability, lithology, pore-water chemical quality, geologic structure, fracturing, and fluid movement and distribution. Quantitative interpretation of porosity, permeability, and water salinity may also be performed. By correlating the logs obtained from boreholes, a subsurface geologic map can be drawn showing faults, structures, and changes in lithology and sedimentation. Geophysical methods that have been applied in oil field well logging include electrical (self potential and resistivity), acoustic and nuclear logs. Caliper and temperature logs are also made to aid data interpretation. Generally, a standard ensemble of logs is run; the appropriate logging tools are selected according to the in situ rock conditions and the specific information required for the study. Costs for the standard set of logs is about 66 (1982 U.S.) dollars per meter, including interpretation.

The geophysical logs must be interpreted in conjunction with each other and with available core samples. Geophysical logs are made (by a field geologist, engineer, or technician) after drilling and core sampling have been completed and an initial evaluation of conditions has been performed. Depending on the method used, the borehole may be cased or uncased. For unstable rock in uncased holes, the boreholes should be kept open using a material such as drilling mud during the logging operation. Drilling mud also provides an electrolytic medium between the downhole probe and the wall of the borehole if electric logging is performed.

- Electrical logging The physical properties measured in electrical well logging are electrical resistivity and self (or spontaneous) potential. Self potential and resistivity were the earliest and are still the most frequently applied methods of electrical logging. An electric log consist of simultaneously run measurements of electrical resistivity and self potential. Such logs are valuable for correlation between boreholes and for subsurface mapping.
 - <u>self-potential (spontaneous potential) logging</u> Potentials in boreholes may be caused by a number of effects, e.g., streaming (electrokinetic) potential, shale (diffusion) potential, liquid junction, and mineralization. The principal effect encountered, however, is probably caused by electrochemical reactions occurring between the drilling fluid and the formation interstitial water (Wyllie, 1949).

Measurements are usually made by recording electric potential differences between an electrode in a borehole and another electrode at the surface of the underground opening. In some instances, the potential gradient is measured between two downhole electrodes positioned at small spacings.

The density and resistivity of the mud can seriously affect the measurement of potential. When the drilling mud is very salty, the potential curve is flat and may be useless. This condition may occur in salt.

- resistivity logging - The electrical resistivity of a fluidsaturated rock depends mainly on the pore fluid conductivity, porosity, interconnection of voids, and bed thickness. In general, resistivity is inversely proportional to porosity and salinity. Usually, direct current or alternating current of low frequency is applied to current electrodes and the potential is measured between two or more potential electrodes. The result is a plot of apparent resistivity versus depth.

Acoustic logging - Acoustic logs measure the elastic or seismic properties of the rock. In the acoustic logging method, sonic energy is generated by a transmitter in the borehole, and transmitted to one or more receivers. When the acoustic energy is recorded, identification can be made of four wave types. The first wave arrival is a compressional (P-) wave that travels in the rock surrounding the borehole. The second wave arrival is a shear (S-) wave that takes the same path. The third arrival is a fluid wave that travels through the fluid column in the borehole. The last (fourth) arrival is a Stoneley wave that travels along the area of contact between the borehole wall and the borehole liquid.

The principal applications of sonic logs have been in detecting fractures (King and McConnell, 1973) and in determining porosity (Berry, 1959). Porosity can be determined using a standard formula that relates porosity to the wave velocity in the rock (formation) and the wave velocity in the fluid that fills the pore space (Wyllie et al, 1958).

Nuclear logs - Some atomic nuclei emit natural radiation and others can be induced to do so. Although several types of rays are emitted (alpha, beta, and gamma rays, and neutrons), only gamma rays and neutron emissions have enough penetration to be of practical use in logging.

Wel' logging instruments are basically of three types: (1) those that detect gamma radiation from the natural radioactive decay of uranium (U), thorium (Th), and potassium (K) in rocks; (2) those that use artificial gamma rays; and (3) those that induce nuclear processes using neutron sources.

All natural rocks contain some radioactivity due to the presence of U, Th, and K40. Radioactivity is generally lowest in basic igneous rocks, intermediate in metamorphic rocks, and highest in some sediments and granitic rocks. Shales, clays, and marls are generally several times more radioactive than sandstones, limestones, and dolomites. In a given area, the radioactivity of shales does not vary much, so that a gamma-ray log is an approximate measure of the amount of shale in the formation.

G.4.3 GEOPHYSICAL SEISMIC SURVEYS

Seismic surveys are useful for correlating rock units between boreholes, estimating rock quality, and estimating the elastic constants of the rock mass. There are three basic types of seismic survey: refraction, crosshole, and reflection. All three methods involve measuring the time required for the shear or compressional waves to travel between the source and the sensor. Seismic refraction and reflection are performed from inside of the tunnel while crosshole seismic is performed between boreholes (see Figure G-24).

Seismic Refraction

Seismic refraction measurements consist of creating a seismic impulse by explosive or mechanical means and recording arrival times of refracted seismic waves along a line of measurement (see Figure G-24a). Straight lines fitted to distance (from shot point) vs. arrival-time plots permit calculation of seismic velocities for the different layers encountered. Locations of velocity breaks in the travel-time curve are used to calculate distances to interfaces. In this method, the distance between shotpoints and detectors (geophones) is large compared to the distance to the seismic layers investigated; the separation needed is roughly three to five times the penetration.

Seismic body waves are either of the compressional (P-) or shear (S-) type. P-waves are always faster than S-waves in a given medium. For this reason, P-waves are used almost exclusively in seismic refraction, although experiments using S-waves have been performed and newer developments have made the use of S-waves in lithologic determinations potentially more interesting.

Seismic refraction is used primarily to determine seismic velocities and distance to adjacent layers. Such determinations are used not only for engineering purposes but also for corrections applied to seismic reflection surveys. In fact, seismic refraction invesigations of adjacent layers can be made part of a seismic reflection survey.

The usefulness of seismic refraction measurements is limited to investigation of the rock near the test opening. As the distance from the opening increases, the method becomes more and more approximate and difficult to interpret. The refraction method depends on the assumption that seismic velocities increase with distance from the source. If this assumption is not fulfilled, the method gives erroneous results.

Seismic velocities are directly related to the dynamic elastic constants of a medium, and velocities can be used to characterize the rock and soil types encountered.

With eversed refraction profiles, location and dip of subsurface layers can be calculated. Other methods of interpretation exist for determining variations in subsurface depth for each geophone location.

Crosshole Seismic

In the crosshole shear-wave method, a seismic impulse is created in one borehole and wave arrivals are detected in an adjacent borehole (see Figure G-24b). It has been found that this method gives more accurate results and more detailed information concerning layers or fractured zones than other acoustic measurements, either in single boreholes or in uphole and downhole configurations (Mirafuente et al, 1974).

SEISMIC SURVEY TECHNIQUES





a) Seismic refraction

Eng. WW

Dwg. No. 8/1/12/1 At 024 Date 142



b)Schematic of a cross-hole seismic survey technique after Stokoe and Woods, 1972

Processing of crosshole shear-wave data is straightforward, and little or no interpretation is needed. The distance between borenolas divided by the travel time of shear waves gives the shear-wave velocity directly. The velocity can then be converted to dynamic shear modulus, if an estimate of density is available. Simultaneous measurement of compressional wave velocity can be used to estimate the dynamic elastic constants, e.g., Young's modulus or Poisson's ratio. Amplitudes of recorded shear and compressional waves can be used to determine attenuation characteristics of materials. These characteristics give additional information on rock quality and fracturing.

Seismic Reflection

Seismic reflection measurements are performed by measuring travel times of seismic waves artifically generated at or near a surface that return to that surface after being reflected. Compared with seismic refraction, shot-to-detector distances are small compared to distances investigated, and a smaller amount of energy is needed per shot for a given penetration. Normally, only P-waves are used.

Seismic reflection is by far the most expensive exploration method. However, it is also the most effective and most highly developed method which gives detailed and accurate information on the location of subsurface layers, faults, and cavities, as well as velocities. These are derived from measurements of reflection time vs. separation distance. In addition, amplitude measurements made on digital recordings define the reflection c efficient (velocity times density) and attenuation characteristics.

With accurate velocity information, distances to reflective layers can be calculated, usually to within a few percent. Correlation with nearby borings increases the accuracy. Resolution is a function of penetration distance and contrast in properties between layers. With increasing penetration distance, higher frequencies are attenuated, resulting in decreased resolution.

Seismic cross sections provide a wealth of data on the subsurface configuration, including faults and voids in, for example, salt. These structures are derived by interpreting the seismic cross section, and skill of the interpreter is a key factor in the quality of the interpretation. Voids may be indicated only by the presence of a diffraction pattern in the seismic section; however, if large enough, they may be completely resolved on a seismic section.

In spite of sophisticated methods used to attenuate multiple reflections, interpretation of the exact nature of a seismic horizon is still subject to critical evaluation by the interpreter.

REFERENCES

- Berry, J.E., "Acoustic Logging in Porous Media," American Institute of Mining Engineer, Transactions 216, 262-270 (1959). Available in public technical libraries.
- Cording, E.J. et al, "Methods for Geotechnical Observations and Instrumentation in Tunneling," prepared for NSF, Grant No. GI-33644X, December 1975. Available in public technical libraries.
- Council on Scientific and Industrial Research (CSIR), Instruction Manual for the Use of the CSIR Triaxial Rock Stress Measuring Equipment, 1973. Available from CSIR, Pretoria, South Africa.
- Commonwealth Scientific and Industrial Research Organization (CSIRO), Field Manual for CSIRO Hollow Inclusions Gauge, 1979. Available from CSIRO, Melbourne, Australia.
- Dunnicliff, C.J., "Equipment for Field Deformation Measurements," pp. 319-332 in Proceedings of the Fourth Pan-American Conference on Soil Mechanics and Foundation Engineering, San Juan, Vol. 2, 1971. Available in public technical libraries.
- Golder Associates, "Geologic, Stratigraphic, Thermal and Mechanical Factors which Influence Repository Design in the Bedded Salt Environment," for Lawrence Livermore Laboratory, December 1979a. Available for purchase from NTIS, UCRL-15203.
- Golder Associates, "Derivation of Parameters Necessary for the Evaluation of Performance of Sites for Deep Geological Repositories," for Lawrence Livermore Laboratory, 1979b. Available for purchase from NTIS, UCRL-15166 (V. 2).
- Golder Associates, "Geologic, Structural, Thermal and Mechanical Factors which Influence Repository Design in Granitic and Basaltic Rocks," for Lawrence Livermore Laboratory, 1979c. Available for purchase from NTIS, ERA5-17446.
- Golder Associates, "Data for the Basalt Post-Closure Reference Repository, Hanford Reservation, Washington; Vol. 2 - Prior to Construction," for NRC under Contract No. NMS-02-81-027, December 1981. Available in NRC PDR for inspection and copying for a fee.
- Golder Associates, "Identification of Characteristics which Influence Repository Design - Domal Salt," Report for Nuclear Regulatory Commission, NUREG/CR-2613, 1982a. Available in NRC PDR for inspection and copying for a fee.

- Golder Associates, "Identification of Characteristics which Influence Repository Design - Tuff," Report for Nuclear Regulatory Commission, NUREG/CR-2614, 1982b. AVailable in NRC PDR for inspection and copying for a fee.
- Golder Associates, "Evaluation of Alternative Shaft Sinking Techniques for High Level Nuclear Waste (HLW) Deep Geologic Repositories," NUREG/CR-2854, for Nuclear Regulatory Commission, 1982c. (March 1983)
- Golder Associates, "Relationship of an In Situ Test Facility to a Ceep Geologic Repository for High Level Nuclear Waste," Task 4 Report for Nuclear Regulatory Commission, NUREG/CR-2959, 1982d. (March .983)
- Golder Associates and J.F. MacLaren Limited, <u>Tunneling Technology</u>, an <u>Appraisal of the State-of-the-art for Application to Transit</u> <u>Systems</u>, Ontario Ministry of Transportation and Communications, <u>1976</u>. Available from Ontario Ministry of Transportation and Communications.
- Goodman, R.E., Methods of Geological Engineering in Discontinuous Rocks, p. 472, West Publishing Co., 1976. Available in public technical libraries.
- International Engineering Company (IECO), "Review of Geotechnical Measurement Techniques for a Nuclear Waste Repository in Bedded Salt," prepared for University of California, Lawrence Livermore Laboratory, December 1979. Available for purchase from NTIS UCRL-15141.
- King, M.S. and B.V. McConnell, "Fracture Evaluation with Acoustical Log in Dry Boreholes," <u>Rock Mechanics</u>, 273-292 (1973). Available in public technical libraries.
- Lawrence Berkeley Laboratory, "Geotechnical Assessment and Instrumentation - Needs for Nuclear Waste Isolation in Crystalline and Argillaceous Rocks - Symposium Proceedings, July 16-20, 1978, prepared for ONWI and DOE, Report No. LBL-7096, January 1979. Available for purchase from NTIS.
- Mirafuente, N.T., E.G. Zurflueh, and T. Stratton, "Improved Shear Wave Measurement Technique of Better Earthquake Resistant Design," <u>Woodward-Clyde Consultants Geotechnical and Environmental Bulletin</u> 7(2), 25-29 (1974). Available in public technical libraries.
- National Research Council, Rock-Mechanics Research Requirements for Resource Recovery, Construction, and Earthquake-Hazard Reduction, Panel on Rock Mechanics Research Requirements, U.S. National Committee for Rock Mechanics, Assembly of Mathematical and Physical Sciences, National Research Council, Report No. NRC/AMPS/RM-81-1 National Academy Press, Washington, D.C., 1981. Available for purchase from NTIS.
- Roberds, W.J., "Risk Assessment Methodology for Geologic Repository of High Level Nuclear Waste," in <u>Proceedings of Symposium on</u> <u>Uncertainties Associated with the Regulation of the Geologic</u> <u>Disposal of High-Level Radioactive Waste, NUREG/CP-0022, Oak Ridge</u> <u>National Laboratory, Tennessee, March 1982</u>. Available for purchase from NTIS.
- Rocha, M., "A New Method of Integral Sampling of Rock Masses," Rock Mechanics 3, 1-12 (1971). Available in public technical libraries.
- Stagg, K.G. and O.C. Zienkiewicz, Rock Mechanics in Engineering Practice, J. Wiley and Sons, London, 1968. Available in public technical libraries.
- Stokoe, K.H. and R.D. Woods, "In Situ Shear Wave Velocity of Cross-Hole Method," J. Soil Mech. Founda. Div., ASCE, 98 (SM5), 443-460 (May 1972). Available in public technical libraries.
- U.S. Dept. of Energy, "Geologic Disposal of Radioactive Waste: Program Plan for Field Testing," DOE/NWTS-80(1), Battelle Memorial Institute, 1981. Available for purchase from NTIS.
- U.S. Environmental Protection Agency, <u>Code of Federal Regulations 40</u>, Part 191, "Environmental Standards and Federal Radiation Protection Guidance for Management and Disposal of Spent Nuclear Fuel, High-Level and Transuranic Radioactive Wastes," U.S. Gov't Printing Office, 1982. Available in public technical libraries.
- U.S. Nuclear Regulatory Commission, Code of Federal Regulations 10, Part 60, "Disposal of High Level Radioactive Wastes in Geologic Repositories," U.S. Gov't Printing Office, 1982. Available in public technical libraries.
- U.S. Nuclear Regulatory Commission, "Standard Format and Content of Site Characterization Reports for High-Level-Waste Geologic Repositories," Draft Regulatory Guide and Value/Impact Statement, Task G5 027-4, April, 1981. Available from U.S. NRC.
- Wyllie, M.J.R., "Statistical Study of Accuracy of Some Connate-Water Resistivity Determinations Made from Self-Potential Log Data," <u>American Association of Petroleum Geologists Bulletin</u> 33, 1892-1900 (1949). Availale in public technical libraries.
- Wyllie, M.J.R., A.R. Gregory, and G.H.F. Gardener, "An Experimental Investigation of Factors Affecting Elastic Wave Velocities in Porous Media," <u>Geophysics</u> 23, 459-493 (1958). Available in public technical libraries.

Zamanek, J.J., "Formation Evaluation by Inspection with the Borehole Televiewer," <u>Geophysics</u> 35, 254-269 (1970). Available in public technical libraries.

IRC FORM 335 U.S. NUCLEAR REGULATORY COMMISSION BIBLIOGRAPHIC DATA SHEET		1. REPORT NUMBER (Assigned by DDC) NUREG/CR-3065, Vol. 1 813-1163	
4 TITLE AND SUBTITLE (Add Volume No., if appropriate) In Situ Test Programs Related to Design and Construction of High Level Nuclear Waste (HLW) Deep Geologic Reposi-		2. (Leave blank) 3. RECIPIENT'S ACCESSION NO.	
tories - Main Report			
W. Dershowitz, F. Marinelli, K. Jones, C. Weldon, B. Nebbi M. Stewart, R. Gates, D. Pentz, J. Byrne		5. DATE I PORT COMPLETED tt MONTH YEAR November 1982	
9. PERFORMING ORGANIZATION NAME AND MAILING ADDRESS (Inclus	de Zip Codel	DATE REPORT IS	SUED
Golder Associates 2950 Northup Way		Month March	1983
Bellevue, WA 98004-1486	Bellevue, WA 98004-1486		
		8. (Leave blank)	
12. SPONSORING ORGANIZATION NAME AND MAILING ADDRESS (Include Zip Code) Division of Waste Management Office of Nuclear Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20555		10 PROJECT/TASK/WORK UNIT NO.	
		11. FIN NO.	
		B-6983	
13. TYPE OF REPORT	PERIOD COVER	RED (Inclusive dates)	
Technical Assistance (formal)	June, 198	1 - November, 1982	
15. SUPPLEMENTARY NOTES		14 (Leave blank)	
This report includes the general recommendation sidered in designing media/site specific in si in an exploratory shaft and an underground tes horizon. Testing plans will be presented by DO ted in the LA. A licensing perspective is outl and utilized for the test selection process. T bilities of tests to the perceived information function of significant engineered components, and the acceptable level of confidence in sati	n of availab tu test prog t facility a DE in an SCR ined and a d his rational needs for L site charac	le tests which rams. Tests w t the prospect and the comple efensible rati e consists of A. The inform teristics, ava	should be con- ill be conducted ive repository te results presen- onale developed matching the capa- ation needs are a
Tests which are available and respond to the p needs, either by simulation or assessment of t capabilities assessed. Specific in situ tests	sfactory per perceived med the character are investi	formance for e ia/site specif istics,are ide gated and desc	ilable information ach licensingstep ic information ntified and their ribed in detail.
Tests which are available and respond to the p needs, either by simulation or assessment of t capabilities assessed. Specific in situ tests	sfactory per perceived med the character are investi	formance for e ia/site specif istics,are ide gated and desc	ilable information ach licensingstep ic information ntified and their ribed in detail.
Tests which are available and respond to the p needs, either by simulation or assessment of t capabilities assessed. Specific in situ tests 17. KEY WORDS AND DOCUMENT ANALYSIS 17. IDENTIFIERS OPEN-ENDED TERMS	sfactory per perceived med the character are investi	formance for e ia/site specif istics, are ide gated and desc	ilable informatior ach licensingstep ic information ntified and their ribed in detail.
Tests which are available and respond to the p needs, either by simulation or assessment of t capabilities assessed. Specific in situ tests 17. KEY WORDS AND DOCUMENT ANALYSIS 17. IDENTIFIERS OPEN-ENDED TERMS	sfactory per perceived med the character are investi	formance for e ia/site specif istics,are ide gated and desc	ilable informatior ach licensingstep ic information ntified and their ribed in detail.
Tests which are available and respond to the p needs, either by simulation or assessment of t capabilities assessed. Specific in situ tests 17. KEY WORDS AND DOCUMENT ANALYSIS 17. IDENTIFIERS OPEN-ENDED TERMS 18. AVAILABILITY STATEMENT	sfactory per perceived med the character are investi 17a DESCRIPTON 17a DESCRIPTON 19 SECURI Uncli	formance for e ia/site specif istics, are ide gated and desc ssis	ilable information ach licensing step ic information ntified and their ribed in detail.